# IMPLEMENTATION OF PHYSICAL ACTIVITY RECOMMENDATIONS IN PEOPLE WITH AXIAL SPONDYLOARTHRITIS

Anne-Kathrin Rausch Osthoff

# Implementation of physical activity recommendations in people with axial spondyloarthritis

Anne-Kathrin Rausch Osthoff

ISBN:	978-94-6419-604-7
Cover image:	Jonas Wüllner
Cover:	llse Modder – www.ilsemodder.nl
Lay-out:	llse Modder – www.ilsemodder.nl
Printing:	Gildeprint – www.gildeprint.nl

The studies in this thesis were financially supported by:

European Alliance of Associations for Rheumatology (chapters 2 and 3), Dutch Arthritis Foundation (chapter 4), Swiss Physical Therapy Association physioswiss (chapters 5 and 6), Kurt-and-Senta-Herrmann Foundation (chapter 7), and Rheumaliga Zurich (chapter 7).

Copyright © 2022 Anne-Kathrin Rausch Osthoff

# Implementation of physical activity recommendations in people with axial spondyloarthritis

Proefschrift

ter verkrijging van de graad van doctor aan de Universiteit Leiden, op gezag van rector magnificus prof. dr. ir. H. Bijl, volgens besluit van het college voor promoties te verdedigen op woensdag 30 november 2022 klokke 11.15 uur

door

Anne-Kathrin Rausch Osthoff geboren te Offenbach a.M., Duitsland in 1984

#### Promotor

Prof. Dr. T.P.M. Vliet Vlieland

#### **Co-Promotor**

Prof. Dr. K. Niedermann, School of Health Sciences, Institute of Physiotherapy, Zurich University of Applied Sciences, Winterthur, Switzerland

#### Leden promotiecommissie

Prof. Dr. J.C. Kiefte-de JongProf. Dr. A.E.R.C.H. Boonen, Maastricht University Medical CentreProf. Dr. S. Rubinelli, Department of Health Sciences and Medicine, University of Lucerne, SwitzerlandProf. Dr. R. Hilfiker, School of Health Sciences, University of Applied Sciences HES-SO

Valais-Wallis, Switzerland

# Contents

Chapter 1	General Introduction	9
Chapter 2	2018 EULAR recommendations for physical activity in people with inflammatory arthritis and osteoarthritis <i>Annals of Rheumatic Diseases 2018; 77 (9):1251-1260</i>	23
Chapter 3	Effects of exercise and physical activity promotion: meta- analysis informing the 2018 EULAR recommendations for physical activity in people with rheumatoid arthritis, spondyloarthritis and hip/knee osteoarthritis <i>RMD Open 2018; 4 (2): e000713</i>	47
Chapter 4	The perspective of people with axial spondyloarthritis regarding physiotherapy: room for the implementation of a more active approach	71
Chapter 5	Understanding beliefs related to physical activity in people living with axial spondyloarthritis - a theory-informed qualitative study BMC Rheumatology 2022; 6 (1): 40	91
Chapter 6	Reliability of an adapted core strength endurance test battery in individuals with axial spondyloarthritis <i>Clinical Rheumatology 2021; 40 (4): 1352-1360</i>	111
Chapter 7	Lessons learned from a pilot implementation of physical activity recommendations in axial spondyloarthritis exercise group therapy <i>BMC Rheumatology 2022; 6 (1): 12</i>	129
Chapter 8	General Discussion	159
Chapter 9	Summary	177
Chapter 10	Nederlandse samenvatting	187
Appendix	List of publications Curriculum vitae Acknowledgements	200 204 205

1

General introduction

Rheumatic and Musculoskeletal Diseases (RMDs) encompass over 200 conditions of the musculoskeletal system, including osteoarthritis and inflammatory arthritis. RMDs are characterized by pain, stiffness, loss of locomotor function and, sometimes, premature death (1). About one quarter of the European population of all ages is affected by RMDs. This results in direct and indirect healthcare costs of around EUR 240 billion per year, or some 2% of the gross domestic product of European countries (2). Physical Activity (PA) is one of the most important lifestyle factors influencing human health, along with balanced diet, cessation of smoking and/or abusive substance use, and reduction of stress. PA is considered a modifiable behaviour that can reduce the risk of developing an RMD, positively influence its course and prevent some of its consequences (3-6). This thesis will in part focus on axial spondyloarthritis, a form of inflammatory arthritis.

# Definition, consequences, and epidemiology of axial spondyloarthritis

Axial spondyloarthritis (axSpA), which includes both radiographic axSpA (r-axSpA) and non-radiographic axSpA (nr-axSpA), is a chronic, inflammatory RMD that predominately affects the sacroiliac joints and spine. Approximately 30% of people with axSpA have both axial and peripheral joint involvement (7). AxSpA can also present with associated musculoskeletal (e.g., arthritis, enthesitis, dactylitis) or extra-articular manifestations (e.g., uveitis, inflammatory bowel disease, cardiovascular diseases) (8, 9). In people with axSpA, the presence of disease characteristics based on the Assessment of Spondyloarthritis international Society (ASAS) classification varies significantly. These variations include, for example, radiographic or non-radiographic changes (10), or the presence of biomarkers such as the C-reactive protein, the human leukocyte antigen (HLA) B27 status, or the erythrocyte sedimentation rate (11). The chronic inflammation of the axial skeleton leads to back pain and progressive stiffness. The symptoms can be improved with exercise, but not through inactivity. The chronic inflammation may also result in functional and structural impairments (8), e.g., reduced flexibility (12), balance (13), muscle strength (14), or cardiorespiratory capacity (15). Comorbidities are common in people with axSpA. According to data from an international ASAS comorbidity study including 3'370 patients (66% male, mean ±SD age 43±14 years), 51% of people with axSpA suffered from at least one comorbidity and 9% from three or more comorbidities (16). This study also showed that the higher the number of comorbidities (reflected by a higher score on the Rheumatic Disease Comorbidity Index (17)), the greater the impact on physical function, work ability, and quality of life (16). In Spanish, German and Dutch cohorts, the comorbidities most commonly found were hypertension, depression and obesity (18-20). It has also been demonstrated that people with axSpA have an increased risk of cardiovascular diseases (21).

The prevalence of axSpA in the general population is about 0.1%-0.6%, according to European disease prevalence data (22, 23). The male-female prevalence ratio is equal (24).

However, there are gender differences regarding radiologic prognosis (with males more affected) and burden of disease (with females more affected) (25). Characteristically, disease onset is in early adulthood and results in considerable personal burden of disease and negative economic consequences for the individual and society (26-29). Despite efforts in clinical practice to increase awareness and improve diagnostic procedures, the diagnostic delay is still 2-11 years on average, with women waiting longer than men (30-32).

# The management of axSpA

The recommendations of the European Alliance of Associations for Rheumatology (EULAR) for the management of axSpA define the aim of primary treatment as: 'To maximise longterm health-related quality of life through control of symptoms and inflammation, prevention of progressive structural damage, preservation/normalisation of function and social participation' (8). This treat-to-target strategy (goal of remission or low disease activity) implies that the management of axSpA should be personalised, and a combination of pharmacological and non-pharmacological treatment modalities applied (8, 9, 33). The recommended first-line pharmacological treatment for people suffering from pain and stiffness is the prescription of nonsteroidal anti-inflammatory drugs (NSAIDs). For people with consistently high disease activity, additional biological therapies are available (8). Two groups of biological disease-modifying antirheumatic drugs (bDMARDs) and their biosimilars (drugs that are virtually identical to the original) have been approved as effective and safe: tumour necrosis factor inhibitors (TNFi) and interleukin17A inhibitors (IL17Ai) (34). The bDMARDs are used to achieve a threshold of remission or low disease activity and focus on physical function (35). The maximal reduction of symptoms by pharmacological treatment enhances the success of non-pharmacological treatment (36). The ASAS recommendations for non-pharmacological treatment include physiotherapy, education, smoking cessation, and the promotion of PA (8).

Regarding the reduction of disease activity in the management of axSpA, there is evidence to show that combining pharmacological and non-pharmacological treatment modalities through the combination of bDMARDs and physiotherapy is superior to medication alone (36-38). The effect on disease activity was established using the Bath Ankylosing Spondylitis Mobility Index (BASMI) and Bath Ankylosing Spondylitis Disease Activity Index (BASDAI) scores (37, 38).

In research specifically on the effectiveness of exercise programmes, routine pharmacological therapy was usually continued in both the intervention and the control groups. In the most recent Cochrane review from 2019, 14 randomized controlled trials investigating the effect of exercise programmes were examined. The exercise programmes had a mean duration of 12 weeks at a median frequency of three sessions per week and consisted mainly of land-based flexibility, breathing and strength exercises, in combination with standard-NSAIDs or

1

DMARDs (39). The conclusions drawn from the pooled data were that these programmes slightly improve function, potentially reduce pain and, probably, slightly reduce disease activity (moderate-to-low quality evidence) in comparison with no intervention (but standard-NSAIDs) (39). The authors were uncertain about potential adverse events due to lack of evidence and insufficient reporting in the studies (39). The effectiveness of physiotherapy interventions is difficult to demonstrate because of methodological weaknesses, such as small sample sizes, heterogeneous study populations, insufficient description of the interventions and/or flaws in the assessments (39).

The studies investigated in the Cochrane review (39) represent the traditional exercise concept for people with axSpA that is commonly employed in current clinical practice. However, it has been suggested that this traditional exercise concept is no longer state-of-the-art. Recent studies, for example from Sveaas and colleagues (40), have shown that high-intensity aerobic exercise interventions have promising effects on disease activity, inflammatory markers, and symptoms such as pain, fatigue, and stiffness. A greater focus on aerobic exercises also seems to be advisable because of the growing body of knowledge on the increased cardiovascular risk in this patient group (41, 42). Patients also appear to support such a shift in the focus of axSpA exercises and concur with the inclusion of fitness-related exercises alongside to conventional disease-specific exercises (43).

# Definition of physical activity and recommendations

*Physical Activity* (PA) is defined as 'any bodily movement produced by skeletal muscles that results in energy expenditure above resting (basal) levels' (44, 45). PA encompasses any exercise, sport or physical activities done as part of daily living, occupation, leisure and active transportation (46). *Exercise* is a subcategory of PA that is 'planned, structured, and repetitive and [that] has as a final or intermediate objective to improve or maintain one or more dimensions of physical fitness' (44). In contrast, *therapeutic exercises and exercise therapy* comprise specific exercises designed to address distinct functional health problems and are supervised by a qualified health professional, e.g., a physiotherapist.

The World Health Organisation (WHO) provides current, evidence-based public health PA recommendations on the amount (frequency, intensity, and duration) and type of regular PA necessary to offer significant health benefits and mitigate health risks. The *current PA recommendations* for adults (including elderly or people with chronic conditions) encompass four exercise dimensions: cardiorespiratory/aerobic; muscle strength; flexibility; and neuromotor exercises (3, 47). A person is advised to perform at least 150-300 minutes of moderate-intensity PA, or 75-150 minutes of vigorous-intensity aerobic PA (or combination of these) per week, plus additional muscle strengthening activities involving the major muscle groups on at least two days of the week (47). Flexibility and neuromotor exercises are additionally recommended on three or more days during the week (3, 47). Correspondingly,

sitting time should be reduced to a minimum (47).

The intensity of PA is defined through the estimation of energy demands of the metabolic system, i.e., metabolic equivalent of task (MET) (48): for example, walking at 5km/h is described by 3.2METs and classified as a moderate-intensity aerobic activity; and jogging at 9km/h is described by 8.8METs and classified as vigorous aerobic activity. The use of absolute measures may be misleading, however, because personal factors, such as weight, sex, age, and fitness level are not considered (46). For this reason, a relative *measure of intensity* (e.g., %MET<sub>max</sub>, %HR<sub>max</sub> or perceived exertion), which measures the energy costs relative to the individual's maximal capacity, is more appropriate. Moderate exercise is defined as 64% - 76% of the maximum heart rate (%HR<sub>max</sub>) and vigorous exercise as 77% - 95%HR<sub>max</sub>. Another commonly used assessment to evaluate the intensity of exercise is the Borg Rating of Perceived Exertion Scale, which measures the individual's effort and exertion, breathlessness, and fatigue during PA (49). On a numeric visual scale of 6-20, the numbers 6- 12 indicate a moderate-intensity exercise and the numbers 13-20 a vigorous-intensity exercise (49).

The dose-response relationship between daily PA and health benefits has been confirmed (50). Exceeding the minimum recommended amount of PA improves fitness, reduces the risk factors of chronic conditions, and prevents unhealthy weight gain (46). Furthermore, there is evidence that shorter vigorous-intensity exercise (e.g., 30 minutes of activity  $\geq$  6METs) is better than longer moderate-intensity exercise (e.g., 60 minutes of activity at 3-6METs) and is associated with a greater reduction of cardiovascular risk (51) and lower mortality rate in the general population (52).

People not meeting the minimum recommendations are considered physically inactive. The WHO stresses that, when possible, people with chronic conditions should try to meet the public health PA recommendations. It also states that exercising below the recommended levels may still be beneficial to people not able or willing to fulfil the recommended amount of PA (47). The American College of Sports Medicine (ACSM) supports this statement, providing evidence on the importance of PA in the prevention and treatment of chronic conditions. It provides examples of appropriate PA programmes and guidance for the performance of exercise (3).

# Physical activity and axSpA

Meeting the PA recommendations described above and incorporating the four exercise dimensions of aerobic, muscle strength, flexibility and neuromotor exercising is especially difficult for people with axSpA. They face barriers of pain, fatigue, or stiffness that healthy people do not have (53). Other aspects, such as the fear of flare-ups or the presence of comorbidities, may also hamper participation in certain types of exercise. Some physical

consequences of the disease, such as osteoporosis (risk of fracture), severe deformities of the spine (risk of neurological complication), or extra-articular manifestations (e.g., tendons, bowels, kidney, eyes), can make an active lifestyle even more difficult to realise (54). It is therefore not surprising that people living with axSpA have been found to be generally less physically active than healthy controls (55-57). Expert support from physicians and health professionals in rheumatology (HPRs), e.g., physiotherapists, may thus be required to identify and address the individual challenges with respect to PA behaviour.

# Exercise is medicine for people with axSpA: clear recommendations are needed

Despite the fact that exercise (and group exercise, in particular) is mentioned in the recommendations and guidelines for the management of axSpA (8, 58), it has been uncertain for some time whether the public health PA recommendations are applicable to people with axSpA. The public health PA recommendations focus on aerobic exercises at an appropriate dosage (moderate/vigorous intensity, and most days per week frequency) which contrasts with the content of most exercise programmes for people with axSpA found in the literature (39, 47). This is an important distinction.

In 2016, this highly relevant issue led to the initiation of an EULAR task force on the development of PA recommendations for people with RMDs, including axSpA (among others).

The first part of this thesis concerns the development process of the EULAR PA recommendations for people with RMDs, while the second part describes the lessons learned from their pilot implementation specifically in people with axSpA.

# Implementation of PA recommendations in people with axSpA

After the development of a clinical innovation, it must subsequently be translated and implemented into clinical routine. Implementation is the uptake of research findings by routine health care in the clinical, organisational and policy contexts using strategies tailored to the target population, setting, and goals for improvement (59). However, the implementation of innovations into clinical routine is a great challenge (60, 61). It needs a planned, systematic approach with clear strategies. Successful implementation must be based on an analysis of the contextual factors and the knowledge-practice gap, inclusion of stakeholders, and theory-informed systematic approach (59, 62). The Grol&Wensings' model of change (63) is a theoretical approach that is commonly applied to guide and describe an implementation process. After a comprehensive analysis of current practice

and targets of change, implementation strategies and interventions can be developed, evaluated, and continuously and cyclically adapted (63).

It is known that awareness of the public health PA recommendations and their resulting benefits does not necessarily lead to behavioural changes (64-66). PA is behaviourally complex and increasing PA as a lifestyle change remains a challenge (67, 68). A successful health behaviour change can be defined as 'the shift from risky behaviours to the initiation and maintenance of healthy behaviours and functional activities, and the self-management of chronic health conditions' (69). Several theories have been proposed to explain PA behaviour, which is determined by individual, social, environmental, and societal factors (68). Common to many theories is the central distinction of motivation, which is conceptualised (for example) in intention, stage of change, and autonomous motivation (70). Motivation determines whether an individual will change PA behaviour over the long term. Consequently, an understanding of how PA promotion interventions can influence motivation is critical. The use of behaviour change techniques (71), such as goal-setting and self-monitoring, has been shown to be associated with beneficial changes in PA behaviour (70, 72). Fenton and colleagues (73) summarised the evidence on theory-informed interventions for people with axSpA. They emphasised that motivation and theories should be used more intensively to develop interventions and learn "how things work". This would help to develop new strategies, optimise current interventions, and support the translation of the theory into practice process.

# Physical activity promotion in people with axSpA in Switzerland

In Switzerland, 76% of the adult population is considered physically active (74). The Swiss health care system is fee-for-service and regulated by law and organised at cantonal level (75, 76). The cost of only a few PA promotion intervention programmes for people with chronic conditions is partly refunded by health insurance (77). A Swiss national agency for the development of guidelines does not exist and evidence-based interventions are often not systematically applied in clinical care (78). The Swiss Association for Rheumatology did not publish its own guidelines for the management of axSpA, but referenced those published by EULAR and the American College of Rheumatology.

The Ankylosing Spondylitis Association of Switzerland (Schweizerische Vereinigung Morbus Bechterew, SVMB, www.bechterew.ch) represents approximately 4'300 members with axSpA. The SVMB is connected to the national association Swiss League Against Rheumatism and is a member of the Ankylosing Spondylitis International Federation (ASIF) and the European Alliance of Associations for Rheumatology Patient Representatives (EULAR PARE). Since the early 1980s, the SVMB offers - among other services - physiotherapist-supervised, weekly exercise group therapy in approximately 65 locations

across Switzerland. The concept of this exercise group therapy was, however, outdated since it was mainly based on flexibility and low-intensity strength exercising. Recent evidence indicates that greater attention should be paid to cardiovascular exercise (79-82), individual dosage of exercise intensity based on regular assessments (72, 83-85), encouragement to use exercise for self-management of disease and regular performance of individual exercise (86). Consequently, in line with the ambition of the EULAR task force, the SVMB aimed to translate the EULAR PA recommendations into a new exercise group concept that would meet both disease-specific exercising and general PA recommendations, and to subsequently implement it into routine clinical care.

# Outline of this thesis

Given the lack of specific PA recommendations for people with RMDs, particularly axSpA, and their deficient implementation into clinical care, the aims of this thesis were to:

- 1) Describe the development, definition, and evidence base of the 2018 EULAR recommendations for PA in people with inflammatory arthritis and osteoarthritis.
- 2) Evaluate the pilot implementation of exercising according to the EULAR PA recommendations in exercise groups for people with axSpA in Switzerland, including the analysis of the situation, determinants of PA behaviour, and the exploration of a muscle strength assessment.

The first aim is addressed in the following chapters:

- Chapter 2 describes the development and definition of the 2018 EULAR recommendations for PA in people with inflammatory arthritis and osteoarthritis.
- Chapter 3 summarises the evidence on the effects of exercise and PA promotion, according to public health recommendations, in people with rheumatoid arthritis, spondyloarthritis, and hip/knee osteoarthritis. This meta-analysis informed the EULAR task force, defining the EULAR recommendations for PA.

The second aim is addressed in the following chapters:

- Chapter 4 evaluates patients' use of physiotherapy and their perspective on it. A quantitative analysis of the current situation in the Netherlands and Switzerland.
- Chapter 5 explores the patients' beliefs related to PA. A qualitative evaluation of the current situation in Switzerland.
- Chapter 6 describes the reliability of an adapted core strength endurance test battery. This assessment is needed to tailor individual exercises. It is a relevant component of the implementation strategy described below.
- Chapter 7 elucidates the implementation strategy and lessons learned from the pilot implementation of the PA recommendations in axSpA exercise group therapy in Switzerland.
- Chapter 8 discusses the most important findings of the studies in this thesis

and their implications for clinical care. Suggestions for further research are also presented.

• Chapter 9 summarises the thesis.

# References

- EULAR. Horizon 2020 Framwork Programme: EULAR's position and recommendations Brussels: European League against Rheumatism; 2011 [cited December 2021]. Available from: https://www.eular.org/myUploadData/files/EU\_ Horizon\_2020\_EULAR\_position\_paper.pdf.
- 2. EULAR. EULAR contribution to the EU reflection on chronic diseases. Postion stand. Kilchberg, Switzerland: European League Against Rheumatism; 2012.
- American College of Sports Medicine. ACSM's Guidelines for exercise testing and prescription. tenth edition ed. Philadelphia: Wolters Kluwer; 2018.
- Europe WHO. Physical activity strategy for the WHO European Region 2016-2025 Copenhagen, Denmark; 2016 [cited December 2021] Available from: https://www.euro.who.int/en/health-topics/disease-prevention/physical-activity/ publications.
- 5. Azeez M, Clancy C, O'Dwyer T, Lahiff C, Wilson F, Cunnane G. Benefits of exercise in patients with rheumatoid arthritis: a randomized controlled trial of a patient-specific exercise programme. Clinical rheumatology. 2020;39(6):1783-92.
- Kraus VB, Sprow K, Powell KE, Buchner D, Bloodgood B, Piercy K, et al. Effects of Physical Activity in Knee and Hip Osteoarthritis: A Systematic Umbrella Review. Medicine and science in sports and exercise. 2019;51(6):1324-39.
- de Winter JJ, van Mens LJ, van der Heijde D, Landewe R, Baeten DL. Prevalence of peripheral and extra-articular disease in ankylosing spondylitis versus non-radiographic axial spondyloarthritis: a meta-analysis. Arthritis research & therapy. 2016;18:196.
- van der Heijde D, Ramiro S, Landewe R, Baraliakos X, Van den Bosch F, Sepriano A, et al. 2016 update of the ASAS-EULAR management recommendations for axial spondyloarthritis. Annals of the rheumatic diseases. 2017;76(6):978-91.
- Magrey MN, Danve AS, Ermann J, Walsh JA. Recognizing Axial Spondyloarthritis: A Guide for Primary Care. Mayo Clin Proc. 2020;95(11):2499-508.
- Sieper J, Rudwaleit M, Baraliakos X, Brandt J, Braun J, Burgos-Vargas R, et al. The Assessment of SpondyloArthritis international Society (ASAS) handbook: a guide to assess spondyloarthritis. Annals of the rheumatic diseases. 2009;68:1-44.
- 11. Brown MA, Li Z, Cao KL. Biomarker development for axial spondyloarthritis. Nature reviews Rheumatology. 2020;16(8):448-63.
- 12. Dagfinrud H, Kvien TK, Hagen KB. Physiotherapy interventions for ankylosing spondylitis. The Cochrane database of systematic reviews. 2008(1):CD002822.
- Dursun N, Sarkaya S, Ozdolap S, Dursun E, Zateri C, Altan L, et al. Risk of falls in patients with ankylosing spondylitis. J Clin Rheumatol. 2015;21(2):76-80.
- 14. Sahin N, Ozcan E, Baskent A, Karan A, Kasikcioglu E. Muscular kinetics and fatigue evaluation of knee using by isokinetic dynamometer in patients with ankylosing spondylitis. Acta reumatologica portuguesa. 2011;36(3):252-9.
- Peters MJ, Visman I, Nielen MM, Van Dillen N, Verheij RA, van der Horst-Bruinsma IE, et al. Ankylosing spondylitis: a risk factor for myocardial infarction? Annals of the rheumatic diseases. 2010;69(3):579-81.
- 16. Nikiphorou E, Ramiro S, van der Heijde D, Norton S, Molto A, Dougados M, et al. Association of Comorbidities in Spondyloarthritis With Poor Function, Work Disability, and Quality of Life: Results From the Assessment of SpondyloArthritis International Society Comorbidities in Spondyloarthritis Study. Arthritis care & research. 2018;70(8):1257-62.
- England BR, Sayles H, Mikuls TR, Johnson DS, Michaud K. Validation of the rheumatic disease comorbidity index. Arthritis care & research. 2015;67(6):865-72.
- Redeker I, Callhoff J, Hoffmann F, Marschall U, Haibel H, Sieper J, et al. The prevalence and impact of comorbidities on patients with axial spondyloarthritis: results from a nationwide population-based study. Arthritis Res Ther. 2020;22(1).
- Maas F, Arends S, van der Veer E, Wink F, Efde M, Bootsma H, et al. Obesity Is Common in Axial Spondyloarthritis and Is Associated with Poor Clinical Outcome. Journal of Rheumatology. 2016;43(2):383-7.
- Puche-Larrubia MA, Ladehesa-Pineda L, Font-Ugalde P, Escudero-Contreras A, Molto A, Lopez-Medina C, et al. Distribution of comorbidities in spondyloarthritis with regard to the phenotype and psoriasis: data from the ASAS-COMOSPA study. Therapeutic Advances in Musculoskeletal Disease. 2021;13.
- 21. Mathieu S, Gossec L, Dougados M, Soubrier M. Cardiovascular profile in ankylosing spondylitis: a systematic review and meta-analysis. Arthritis care & research. 2011;63(4):557-63.
- 22. Bohn R, Cooney M, Deodhar A, Curtis JR, Golembesky A. Incidence and prevalence of axial spondyloarthritis: methodologic challenges and gaps in the literature. Clin Exp Rheumatol. 2018;36(2):263-74.
- 23. Hamilton L, Macgregor A, Toms A, Warmington V, Pinch E, Gaffney K. The prevalence of axial spondyloarthritis in the UK: a cross-sectional cohort study. BMC musculoskeletal disorders. 2015;16:392.
- Baumberger H, Khan MA. Gradual Progressive Change to Equal Prevalence of Ankylosing Spondylitis among Males and Females in Switzerland: Data from the Swiss Ankylosing Spondylitis Society (svmb). Annals of the rheumatic diseases. 2017;76:929.
- Rusman T, van Vollenhoven RF, van der Horst-Bruinsma IE. Gender Differences in Axial Spondyloarthritis: Women Are Not So Lucky. Curr Rheumatol Rep. 2018;20(6):35.
- 26. Ramonda R, Marchesoni A, Carletto A, Bianchi G, Cutolo M, Ferraccioli G, et al. Patient-reported impact of

spondyloarthritis on work disability and working life: the ATLANTIS survey. Arthritis Res Ther. 2016;18:78.

- 27. Strand V, Singh JA. Patient Burden of Axial Spondyloarthritis. J Clin Rheumatol. 2017;23(7):383-91.
- White D, Lao C, Williams M, Lawrenson R, Dalbeth N. The costs associated with ankylosing spondylitis/axial spondyloarthritis in Aotearoa/New Zealand. N Z Med J. 2019;132(1505):38-47.
- Redeker I, Callhoff J, Hoffmann F, Saam J, Haibel H, Sieper J, et al. Krankheitskosten bei axialer Spondyloarthritis für Patienten mit und ohne Tumor-Nekrose-Faktor-Inhibitor-Behandlung: Ergebnisse einer Routinedatenanalyse. Zeitschrift für Rheumatologie. 2019:1-9.
- Korotaeva T, Dina O, Holdsworth E, Fallon L, Milligan G, Meakin S, et al. Investigating diagnosis, treatment, and burden of disease in patients with ankylosing spondylitis in Central Eastern Europe and the United States: a real-world study. Clinical Rheumatology. 2021.
- Redeker I, Callhoff J, Hoffmann F, Haibel H, Sieper J, Zink A, et al. Determinants of diagnostic delay in axial spondyloarthritis: an analysis based on linked claims and patient-reported survey data. Rheumatology (Oxford, England). 2019;58(9):1634-8.
- Jovani V, Blasco-Blasco M, Ruiz-Cantero MT, Pascual E. Understanding How the Diagnostic Delay of Spondyloarthritis Differs Between Women and Men: A Systematic Review and Metaanalysis. Journal of Rheumatology. 2017;44(2):174-83.
- Smolen JS, Schols M, Braun J, Dougados M, FitzGerald O, Gladman DD, et al. Treating axial spondyloarthritis and peripheral spondyloarthritis, especially psoriatic arthritis, to target: 2017 update of recommendations by an international task force. Annals of the rheumatic diseases. 2018;77(1):3-17.
- Mease P. Emerging Immunomodulatory Therapies and New Treatment Paradigms for Axial Spondyloarthritis. Curr Rheumatol Rep. 2019;21(7):35.
- Braun J, Baraliakos X, Kiltz U. Treat-to-target in axial spondyloarthritis what about physical function and activity? Nature reviews Rheumatology. 2021;17(9):565-76.
- Noureldin B, Barkham N. The current standard of care and the unmet needs for axial spondyloarthritis. Rheumatology (Oxford, England). 2018;57(suppl\_6):vi10-vi7.
- Liang HL, W. R.; Zhang, H.; Tian, X.; Wei, W.; Wang, C. M. Concurrent Intervention With Exercises and Stabilized Tumor Necrosis Factor Inhibitor Therapy Reduced the Disease Activity in Patients With Ankylosing Spondylitis: A Meta-Analysis. Medicine. 2015;94(50):e2254.
- Lubrano ES, A.; Amato, G.; Benucci, M.; Cavazzana, I.; Chimenti, M. S.; Ciancio, G.; D'Alessandro, G.; Angelis, R. D.; Lupoli, S.; Lurati, A. M.; Naclerio, C.; Russo, R.; Semeraro, A.; Tomietto, P.; Zuccaro, C.; De Marco, G. Tumour necrosis factor alpha inhibitor therapy and rehabilitation for the treatment of ankylosing spondylitis: a systematic review. Seminars in arthritis and rheumatism. 2015;44(5):542-50.
- Regnaux JP, Davergne T, Palazzo C, Roren A, Rannou F, Boutron I, et al. Exercise programmes for ankylosing spondylitis. The Cochrane database of systematic reviews. 2019;10:CD011321.
- Sveaas SH, Bilberg A, Berg IJ, Provan SA, Rollefstad S, Semb AG, et al. High intensity exercise for 3 months reduces disease activity in axial spondyloarthritis (axSpA): a multicentre randomised trial of 100 patients. British journal of sports medicine. 2020;54(5):292-+.
- Ferraz-Amaro I, Rueda-Gotor J, Genre F, Corrales A, Blanco R, Portilla V, et al. Potential relation of cardiovascular risk factors to disease activity in patients with axial spondyloarthritis. Therapeutic advances in musculoskeletal disease. 2021;13:1759720X211033755.
- 42. Kim JH, Choi IA. Cardiovascular morbidity and mortality in patients with spondyloarthritis: A meta-analysis. Int J Rheum Dis. 2021;24(4):477-86.
- Bilberg A, Sveaas SH, Dagfinrud H, Mannerkorpi K. How Do Patients With Axial Spondyloarthritis Experience High-Intensity Exercise? Acr Open Rheumatol. 2020;2(4):207-13.
- 44. Caspersen CJ, Powell KE, Christenson GM. Physical activity, exercise, and physical fitness: definitions and distinctions for health-related research. Public Health Rep. 1985;100(2):126-31.
- Keogh JWL, Grigg J, Vertullo CJ. Is Home-Based, High-Intensity Interval Training Cycling Feasible and Safe for Patients With Knee Osteoarthritis?: Study Protocol for a Randomized Pilot Study. Orthopaedic Journal of Sports Medicine. 2017;5(3).
- 46. Garber CE, Blissmer B, Deschenes MR, Franklin BA, Lamonte MJ, Lee IM, et al. American College of Sports Medicine position stand. Quantity and quality of exercise for developing and maintaining cardiorespiratory, musculoskeletal, and neuromotor fitness in apparently healthy adults: guidance for prescribing exercise. Medicine and science in sports and exercise. 2011;43(7):1334-59.
- 47. World Health Organization. WHO Guidelines on physical activity and sedentary behaviour. Geneva: World Health Organization; 2020.
- Jette M, Sidney K, Blumchen G. Metabolic equivalents (METS) in exercise testing, exercise prescription, and evaluation of functional capacity. Clin Cardiol. 1990;13(8):555-65.
- 49. Williams N. The Borg Rating of Perceived Exertion (RPE) scale. Occup Med-Oxford. 2017;67(5):404-5.
- World Health Organization. WHO Guidelines on Physical activity and sedentary behaviour Web Annex Evidence Profiles [cited December 2021] Available from: https://apps.who.int/iris/bitstream/handle/10665/336656/9789240015128-eng. pdf2021
- 51. Swain DP, Franklin BA. Comparison of cardioprotective benefits of vigorous versus moderate intensity aerobic exercise.

Am J Cardiol. 2006;97(1):141-7.

- Lee IM, Paffenbarger RS, Jr. Associations of light, moderate, and vigorous intensity physical activity with longevity. The Harvard Alumni Health Study. Am J Epidemiol. 2000;151(3):293-9.
- Liu SH, Morais SA, Lapane KL, Kay J. Physical activity and attitudes and perceptions towards physical activity in patients with spondyloarthritis: A systematic review. Seminars in arthritis and rheumatism. 2020;50(2):289-302.
- 54. Iversen MD, Scanlon L, Frits M, Shadick NA, Sharby N. Perceptions of physical activity engagement among adults with rheumatoid arthritis and rheumatologists. Int J Clin Rheumtol. 2015;10(2):67-77.
- 55. Swinnen TW, Scheers T, Lefevre J, Dankaerts W, Westhovens R, de Vlam K. Physical activity assessment in patients with axial spondyloarthritis compared to healthy controls: a technology-based approach. PloS one. 2014;9(2):e85309.
- O'Dwyer T, O'Shea F, Wilson F. Decreased physical activity and cardiorespiratory fitness in adults with ankylosing spondylitis: a cross-sectional controlled study. Rheumatology international. 2015;35(11):1863-72.
- 57. Passalent LA, Soever LJ, O'Shea FD, Inman RD. Exercise in Ankylosing Spondylitis: Discrepancies Between Recommendations and Reality. Journal of Rheumatology. 2010;37(4):835-41.
- Ward MM, Deodhar A, Gensler LS, Dubreuil M, Yu D, Khan MA, et al. 2019 Update of the American College of Rheumatology/Spondylitis Association of America/Spondyloarthritis Research and Treatment Network Recommendations for the Treatment of Ankylosing Spondylitis and Nonradiographic Axial Spondyloarthritis. Arthritis & rheumatology (Hoboken, NJ). 2019;71(10):1599-613.
- 59. Wensing M. Implementation science in healthcare: Introduction and perspective. Z Evid Fortbild Qual Gesundhwes. 2015;109(2):97-102.
- Ferreira de Meneses S, Rannou F, Hunter DJ. Osteoarthritis guidelines: Barriers to implementation and solutions. Annals of physical and rehabilitation medicine. 2016;59(3):170-3.
- Swinkels RAHM, van Peppen RPS, Wittink H, Custers JWH, Beurskens AJHM. Current use and barriers and facilitators for implementation of standardised measures in physical therapy in the Netherlands. BMC Musculoskeletal Disorders. 2011;12.
- 62. Coles E, Anderson J, Maxwell M, Harris FM, Gray NM, Milner G, et al. The influence of contextual factors on healthcare quality improvement initiatives: a realist review. Systematic reviews. 2020;9(1):94.
- 63. Grol R, Wensing M. Improving patient care. The implementation of change in health care. 2nd ed. Oxford: Wiley Blackwell; 2013.
- 64. McGuire WJ. Attitudes and attitude change. In: Lindzey G, Aronson E, editors. Handbook of social psychology:Vol 2 Special fields and applications (3rd ed, pp233-346). New York, NY: Random House1985.
- Nicolson PJA, Hinman RS, French SD, Lonsdale C, Bennell KL. Improving Adherence to Exercise: Do People With Knee Osteoarthritis and Physical Therapists Agree on the Behavioral Approaches Likely to Succeed? Arthritis Care Res (Hoboken). 2018;70(3):388-97.
- 66. Roter DL, Hall JA, Merisca R, Nordstrom B, Cretin D, Svarstad B. Effectiveness of interventions to improve patient compliance: a meta-analysis. Med Care. 1998;36(8):1138-61.
- 67. Gabriel KKP, Morrow JR, Woolsey ALT. Framework for Physical Activity as a Complex and Multidimensional Behavior. Journal of physical activity & health. 2012;9:S11-S8.
- 68. Biddle SJH, Mutrie N. Psychology of Physical Activity. London: Routledge; 2001.
- 69. Nieuwenhuijsen ER, Zemper E, Miner KR, Epstein M. Health behavior change models and theories: contributions to rehabilitation. Disability and rehabilitation. 2006;28(5):245-56.
- Knittle K, Nurmi J, Crutzen R, Hankonen N, Beattie M, Dombrowski SU. How can interventions increase motivation for physical activity? A systematic review and meta-analysis. Health Psychol Rev. 2018;12(3):211-30.
- Michie S, Richardson M, Johnston M, Abraham C, Francis J, Hardeman W, et al. The behavior change technique taxonomy (v1) of 93 hierarchically clustered techniques: building an international consensus for the reporting of behavior change interventions. Ann Behav Med. 2013;46(1):81-95.
- 72. O'Dwyer T, Monaghan A, Moran J, O'Shea F, Wilson F. Behaviour change intervention increases physical activity, spinal mobility and quality of life in adults with ankylosing spondylitis: a randomised trial. Journal of physiotherapy. 2017;63(1):30-9.
- Fenton SAM, Duda JL, Veldhuijzen van Zanten J, Metsios GS, Kitas GD. Theory-informed interventions to promote physical activity and reduce sedentary behaviour in rheumatoid arthritis: a critical review of the literature. Mediterr J Rheumatol. 2020;31(1):19-41.
- 74. Bundesamt für Statistik. Schweizerische Gesundheitsbefragung 2019: Körperliche Aktivität und Gesundheit. Neuchatel: Bundesamt für Statstik; 2019. [cited December 2021] Available from: https://www.swissstats.bfs.admin.ch/collection/ ch.admin.bfs.swissstat.de.issue19142131708/article/issue19142131708-01
- 75. Bundesamt für Gesundheit. Gesetzgebung 2021 [cited December 2021] Available from: https://www.bag.admin.ch/bag/ de/home/gesetze-und-bewilligungen/gesetzgebung.html.
- Monitor THSaP. Health Systems in Transition (HiT) profile of Switzerland 2021 [cited December 2021] Available from: https://www.hspm.org/countries/switzerland25062016/livinghit.aspx?Section=3.7%20Payment%20 mechanisms&Type=Section.
- 77. Nast I, Niedermann K, Carlander M, Mattli R, Rausch Osthoff AK, Sommer B, et al. Bewegung als Therapie. Schlussbericht im Auftrag des Bundesamt für Gesundheit Winterthur: Zürcher Hochschule für Angewandte Wissenschaften; 2021.
- 78. Birrenbach T, Kraehenmann S, Perrig M, Berendonk C, Huwendiek S. Physicians' attitudes toward, use of, and perceived

barriers to clinical guidelines: a survey among Swiss physicians. Adv Med Educ Pract. 2016;7:673-80.

- Sveaas SH, Smedslund G, Hagen KB, Dagfinrud H. Effect of cardiorespiratory and strength exercises on disease activity in patients with inflammatory rheumatic diseases: a systematic review and meta-analysis. British journal of sports medicine. 2017;51(14):1065-72.
- 80. Bilberg A, Dagfinrud H, Sveaas SH. Supervised intensive Exercise strengthen Exercise Health Beliefs in Patients with Axial Spondyloarthritis: A Multicentre Randomized Controlled Trial. Arthritis care & research. 2021.
- Niedermann K, Muggli C, Dagfinrud, H, Hermann M, Tamborrini, G, Ciurea A, Bischoff-Ferrari H. Effect of cardiovascular training on fitness and perceived disease activity in people with ankylosing spondylitis. Arthritis care & research. 2013;65(11):1844-52.
- Hilberdink B, Vlieland TV, van der Giesen F, van Gaalen F, Goekoop R, Peeters A, et al. Adequately dosed aerobic physical activity in people with axial spondyloarthritis: associations with physical therapy. Rheumatology international. 2020;40(9):1519-28.
- Millner JRB, Beinke KM, Butterworth RH, Chasle BE, Dutton LJ, Lewington MA, Lim EG, Morley TB, O'Reilly JE, Pickering KA, Winzenberg T, Zochling J. Exercise for ankylosing spondylitis: An evidence-based consensus statement. Seminars in arthritis and rheumatism. 2016;45(4):411-27.
- 84. Hilberdink B, van der Giesen F, Vliet Vlieland T, Nijkamp M, van Weely S. How to optimize exercise behavior in axial spondyloarthritis? Results of an intervention mapping study. Patient education and counseling. 2020;103(5):952-9.
- Fongen C, Sveaas SH, Dagfinrud H. Barriers and Facilitators for Being Physically Active in Patients with Ankylosing Spondylitis: A Cross-sectional Comparative Study. Musculoskeletal care. 2015;13(2):76-83.
- 86. Zangi HA, Ndosi M, Adams J, Andersen L, Bode C, Bostrom C, et al. EULAR recommendations for patient education for people with inflammatory arthritis. Annals of the rheumatic diseases. 2015;74(6):954-62.

# 2

# 2018 EULAR recommendations for physical activity in people with inflammatory arthritis and osteoarthritis

Anne-Kathrin Rausch Osthoff\*, Karin Niedermann\*, Jürgen Braun, Jo Adams, Nina Brodin, Hanne Dagfinrud, Tuncay Duruöz, Bente Appel Esbensen, Klaus-Peter Günther, Emalie Hurkmans, Carsten Juhl, Norelee Kennedy, Uta Kiltz, Keegan Knittle, Michael Nurmohamed, Sandra Pais, Guy Severijns, Thijs W Swinnen, Irene Pitsillidou, Louise Warburton, Zhivko Yankov, Theodora P M Vliet Vlieland

\*AR and KN contributed equally to the manuscript.

Annals of Rheumatic Diseases 2018; 77 (9):1251-1260 doi: 10.1136/annrheumdis-2018-213585

Oral presentations at EULAR Congress in Amsterdam (June 2018), World Physiotherapy Congress in Geneva (May 2019), Deutscher Rheumatologiekongress in Munich (September 2020) Poster Presentation at KNGF Dag van de Fysiotherapeut Hertogenbosch (2018)

## Abstract

Regular physical activity (PA) is increasingly promoted for people with rheumatic and musculoskeletal diseases as well as the general population. We evaluated if the public health recommendations for PA are applicable for people with inflammatory arthritis (iA; Rheumatoid Arthritis and Spondyloarthritis) and osteoarthritis (hip/knee OA) in order to develop evidence-based recommendations for advice and guidance on PA in clinical practice.

The EULAR standardised operating procedures for the development of recommendations were followed. A task force (TF) (including rheumatologists, other medical specialists and physicians, health professionals, patient-representatives, methodologists) from 16 countries met twice. In the first TF meeting, 13 research questions to support a systematic literature review (SLR) were identified and defined. In the second meeting, the SLR evidence was presented and discussed before the recommendations, research agenda and education agenda were formulated.

The TF developed and agreed on four overarching principles and 10 recommendations for PA in people with iA and OA. The mean level of agreement between the TF members ranged between 9.8 to 8.8. Given the evidence for its effectiveness, feasibility and safety, PA is advocated as integral part of standard care throughout the course of these diseases. Finally, the TF agreed on related research and education agendas.

Evidence and expert opinion inform these recommendations to provide guidance in the development, conduct and evaluation of PA-interventions and promotion in people with iA and OA. It is advised that these recommendations should be implemented considering individual needs and national health systems.

# Introduction

Physical activity (PA) is defined as 'any bodily movement produced by skeletal muscles that results in energy expenditure above resting (basal) levels. PA broadly encompasses exercise, sports and physical activities done as part of daily living, occupation, leisure and active transportation' (1, 2). Exercise is a subcategory of PA 'that is planned, structured and repetitive and [that] has, as a final or intermediate objective, the improvement or maintenance of one or more dimensions of physical fitness' (1, 2). PA-interventions can be provided or performed individually or in groups, supervised or non-supervised, in acute or chronic health states, but should always include behavioural change techniques (BCT) to promote long-term adherence (3, 4).

To promote the health benefits of PA in the general population, the WHO (5) and American College of Sports Medicine (ACSM) (2) have provided internationally accepted recommendations for PA (table 1). In this manuscript, the term PA always includes both physical activity and exercise according to the definitions above.

Inflammatory arthritis (iA, in this manuscript encompassing rheumatoid arthritis (RA) and spondyloarthritis (SpA)) and osteoarthritis (OA) (in this manuscript encompassing hip/knee OA (HOA/KOA)) are major causes of pain and disability worldwide (6). There is strong evidence for the benefits of PA on improvements on disease activity (7), activities and participation, however, people with rheumatic and musculoskeletal diseases (RMDs) are in general less active compared with healthy controls (8-10). Possible underlying reasons could be that health care providers (HCP, including rheumatology health professionals (eg, physiotherapist (PT), occupational therapist (OT), nurse, podiatrist, psychologist), physical education professions and medical doctors (rheumatologists and other specialists)) and people with iA and OA may be reluctant towards engaging in PA, fearing flare-up or joint damage by exercising (11). Furthermore, current clinical management recommendations such as the European League Against Rheumatism (EULAR) recommendations on the management of RA (12), SpA (13) or HOA/KOA (14) and the ACSM guidelines for exercise testing and prescription (15) recommend exercise and/or PA, but none of these is specific regarding the required type and dosage. Therefore, it is not clear how these recommendations should be used in routine clinical care. In particular, the evidence on the effectiveness and safety of exercise and PA to a level that meets public health (PH) recommendations has not yet been clearly examined and defined in people with RMDs. A EULAR task force (TF) was therefore set up (1) to evaluate if the PH recommendations for PA are applicable for people with iA and OA; (2) to develop evidencebased recommendations on PA-promotion and -delivery in the management of people with iA and OA and (3) formulate an educational and research agenda.

These EULAR recommendations for PA in people with iA and OA are for HCPs, patient organisations and policy makers.

#### Table 1 Public Health recommendations for PA

#### The ACSM-AHA primary physical activity recommendations\*

- All healthy adults aged 18-65 years should participate in moderate intensity aerobic PA for a minimum of 30 min on 5 days/week or vigorous intensity aerobic activity for a minimum of 20 min on 3 days/week.
- Combinations of moderate and vigorous intensity exercise can be performed to meet this recommendation.
- Moderate intensity aerobic activity can be accumulated to total the 30min minimum by performing bouts each lasting ≥10min.
- Every adult should perform activities that maintain or increase muscular strength and endurance for a minimum of 2 days/week.
- Because of the dose-response relationship between PA and health, individuals who wish further improve their fitness, reduce their risk of chronic diseases and disabilities, and/or prevent unhealthy weight gain my benefit by exceeding the minimum recommended amounts of PA.

Cardiorespiratory ("	(aerobic	exercise**
----------------------	----------	------------

Frequency	$\geq$ 5 days/week of moderate exercise, or $\geq$ 3 days/week of vigorous exercise, or a combination of moderate and vigorous exercise on $>3-5$ days/week is recommended.
Intensity	Moderate and/or vigorous intensity is recommended for most adults. Light to moderate intensity exercise may be beneficial in deconditioned persons.
Time	30–60 min/day (150 min/week) of purposeful moderate exercise, or 20–60 min/day (75 min/week) of vigorous exercise, or a combination of moderate and vigorous exercise per day is recommended for most adults. ≥20 min/day (150 min/week) of exercise can be beneficial, especially in previously sedentary persons.
Туре	Regular, purposeful exercise that involves major muscle groups and is continuous and rhythmic in nature is recommended.
Volume	A target volume of ≥500–1000 MET min/week is recommended. Increasing pedometer step counts by ≥2000 steps per day to reach a daily step count ≥7000 steps per day is beneficial. Exercising below these volumes may still be beneficial for persons unable or unwilling to reach this amount of exercise.
Pattern	Exercise may be performed in one (continuous) session per day or in multiple sessions of $\geq$ 10 min to accumulate the desired duration and volume of exercise per day. Exercise bouts of $\geq$ 10 min may yield favourable adaptations in very deconditioned individuals. Interval training can be effective in adults.
Progression	A gradual progression of exercise volume by adjusting exercise duration, frequency, and/or intensity is reasonable until the desired exercise goal (maintenance) is attained. This approach may enhance adherence and reduce risks of musculoskeletal injury and adverse CHD events.
resistance exe	rcise**
Frequency Intensity	Each major muscle group should be trained on 2–3 days/week 60%–70% of the 1RM (moderate to hard intensity) for novice to intermediate exercisers to improve strength. ≥80% of the 1RM (hard to very hard intensity) for older persons beginning exercise to improve strength. 40%–50% of the 1RM (very light to light intensity) for older persons beginning exercise to improve strength. 40%–50% of the 1RM (very light to light intensity) may be beneficial for improving strength in sedentary persons beginning a resistance training program. ≤50% of the 1RM (light to moderate intensity) to improve muscular endurance. 20%–50% of the 1RM in older adults to improve power.
Time Type	No specific duration of training has been identified for effectiveness. Resistance exercises involving each major muscle group are recommended. A variety of exercise equipment and/or body weight can be used to perform these exercises.
Repetitions	8–12 repetitions are recommended to improve strength and power in most adults. 10–15 repetitions are effective in improving strength in middle aged and older persons starting exercise 15–20 repetitions are recommended to improve muscular endurance.
Sets	Two to four sets are the recommended for most adults to improve strength and power. A single set of resistance exercise can be effective especially among older and novice exercisers. $\leq 2$ sets are effective in improving muscular endurance.
Pattern	Rest intervals of 2–3 min between each set of repetitions are effective. A rest of ≥48hours between sessions for any single muscle group is recommended.
Progression	A gradual progression of greater resistance, and/or more repetitions per set, and/or increasing frequency is recommended.

Flexibility exercise**				
Frequency	≥ 2-3 days/week is effective in improving joint range of motion, with the greatest gains occurring with daily exercise.			
Intensity	Stretch to the point of feeling tightness or slight discomfort.			
Time	Holding a static stretch for 10–30 s is recommended for most adults. In older persons, holding a stretch for 30–60 s may confer greater benefit. For PNF stretching, a 3- 6s contraction at 20%–75% maximum voluntary contraction followed by a 10-30s assisted stretch is desirable.			
Туре	A series of flexibility exercises for each of the major muscle-tendon units is recommended. Static flexibility (active or passive), dynamic flexibility, ballistic flexibility, and PNF are each effective.			
Volume	A reasonable target is to perform 60s of total stretching time for each flexibility exercise.			
Pattern	Repetition of each flexibility exercise two to four times is recommended. Flexibility exercise is most effective when the muscle is warmed through light to moderate aerobic activity or passively through external methods such as moist heat packs or hot baths.			
Progression	Methods for optimal progression are unknown.			
Neuromotor exercise training**				
Frequency	$\geq$ 2–3 days/week is recommended.			
Intensity	An effective intensity of neuromotor exercise has not been determined.			
Time	≥20-30 min/day may be needed.			
Туре	Exercises involving motor skills (e.g., balance, agility, coordination, and gait), proprioceptive exercise training,			
	and multifaceted activities (e.g., tai ji and yoga) are recommended for older persons to improve and maintain			
	physical function and reduce falls in those at risk for falling. The effectiveness of neuromuscular exercise			
	training in younger and middle-aged persons has not been established, but there is probable benefit.			
Volume	The optimal volume (e.g., number of repetitions, intensity) is not known.			
Pattern	The optimal pattern of performing neuromotor exercise is not known.			
Progression	Methods for optimal progression are not known.			

\*ACSM : American College of Sports Medicine, AHA : American Heart Association, extracted from the ACSM Guidelines for Exercising Testing and Prescription, capter 1, p4 (15) ; \*\*Extracted from ACSM position stand (2), Table 2, p1336; PA, physical activity, MET, metabolic equivalent of task; CHD, coronary heart disease; 1 RM, one-repetition maximum; PNF, proprioceptive neuromuscular facilitation.

## Methods

The EULAR standardised operating procedures for the development of recommendations were followed (16). The AGREE II-instrument (17) was used to structure this manuscript.

The multidisciplinary TF consisted of a selection of 22 European PA-experts (six medical doctors, including three rheumatologists, one of them specialised in cardiovascular diseases, one GP, one orthopaedic surgeon; nine PTs, a psychologist, an OT, a nurse and a human movement scientist) and three patient representatives. A steering group managed the process (convenor KN, methodologist TVV, expert JB, fellow AR).

During the first TF meeting, definitions of exercise and PA were clarified and the TF agreed to follow the ACSM position stand (2). The TF agreed that RA and SpA as predominant iA conditions, and HOA/KOA as most relevant for PA recommendations would represent the field of iA and OA, respectively. Clinically relevant questions on the provision of advice and guidance regarding exercise and PA, from which 13 research questions were defined by consensus to guide the subsequent detailed systematic literature review (SLR) (online supplementary table S1).

Two SLRs were performed by AR with the support of two librarians and under the supervision of the convenor and methodologists. The questions were written according to the Population, Intervention, Comparison, Outcome (PICO) format (18), resulting in two PICOs: (1) on effectiveness, safety and feasibility of PA and (2) on facilitators and barriers towards PA (online supplementary table S2). For the first PICO, the fellow searched for key meta-analyses (MAs) or systematic reviews (SRs) including randomised controlled trials (RCTs) that investigated the effectiveness of PA-interventions in adults with RA/SpA/HOA/KOA. The SLR was performed in PubMed/Medline, Cochrane Library, Embase, Web of Science, Emcare and PsycInfo, using both MeSH terms and freetext, covering the time frame until 4/2017. For the second PICO, a SLR, covering the time frame until 7/2017, was performed in PubMed/Medline and Cochrane Library including qualitative studies if they described facilitators and barriers regarding PA (including exercise) in people with RA/SpA/HOA/KOA. Experts in the field of RA (EH), SpA (HD), OA (CJ) and behaviour change (KK), respectively, checked if all relevant titles and abstracts were included.

Based on the PICOs, the same author (AR) screened the titles and abstracts according to inclusion and exclusion criteria. Potentially relevant articles were identified and full text versions evaluated. Studies including adults (>18 years) with RA/SpA/HOA/KOA that included PA interventions that met the PH recommendations according to the ACSM principles (2) regarding frequency, intensity and duration for effective interventions were eligible for inclusion. All data extractions were checked by experts from the TF.

Studies measuring the effectiveness of PA-interventions were meta-analysed. These results and detailed descriptions of the methods are reported elsewhere (19). Studies were used for answering more than one research question if appropriate. For clinical studies evaluating the effectiveness of PA, the Cochrane Risk of Bias Assessment Tool was used to assess selection bias, performance bias, detection bias, attrition bias and reporting bias (20) by two independent assessors (AR, CH). An additional person (KN) helped to resolve any differences in rating between the assessors. The research evidence was categorised according to the Oxford levels of evidence (21).

During the second TF meeting, the results from the SLR were presented, and the experts developed 'overarching principles' (background statements to preface recommendations) and drafted 10 recommendations through an iterative process of discussion and consensus. After the meeting, the recommendations were collated and sent to the TF members by e-mail, to rate the level of agreement (LoA) independently and anonymously on a 0-10 point scale (0 =totally disagree, 10 =totally agree). Mean LoA >8 would be considered a 'high' LoA. Furthermore, the TF formulated a research agenda and education agenda based on identified gaps in the evidence.

## Results

The search yielded 3471 references, 96 of which were included in the SLR: Four MA/SR (7, 22-24) and 66 RCTs (25-93) investigated the effects of exercise interventions, 11 RCTs (94-106) investigated the effects of a PA-promotion-intervention, 11 qualitative studies and literature reviews (3, 11, 107-115) described barriers and facilitators regarding PA (figure 1A, B). The included RCTs were published between 1985 and 2017. Most information is from studies with low (48%) or unclear (39%) risk of bias (online supplementary figure S1).

The TF agreed on four overarching principles and 10 recommendations for PA in people with RA/SpA/HOA/KOA based on SLR and expert opinion. High IoA was achieved for 9 out of 10 recommendations and 2 recommendations were graded as strength level A. Table 2 summarises the overarching principles and recommendations with their associated level of evidence, strength of recommendation and LoA.

#### Table 2 Recommendations for PA and exercise in people with inflammatory arthritis and OA

#### **Overarching Principles**

1. PA is part of a general concept to optimize health related quality of life.

2. PA has health benefits for people with RA/SpA/HOA/KOA.

 General PA recommendations, including the four domains (cardiorespiratory fitness, muscle strength, flexibility and neuromotor performance) are applicable (feasible and safe) to people with RA/SpA/HOA/KOA.

 The planning of PA requires a shared decision between health care providers and people with RA/SpA/HOA/KOA, which takes people's preferences, capabilities, and resources into account.

Recommendations		Category	Strength of	Level of Agreement
		evidence	dation	Median (Range)
1.	Promoting PA consistent with general PA recommendations should be an integtral part of standard care throughout the course of disease in people with RA/SpA/HOA/KOA.	1B	A	9.81 (0.39) 10 (9-10)
2.	All health care providers involved in the management of people with RA/SpA/ HOA/KOA should take responsibility for promoting PA, and should cooperate, including making necessary referrals, to ensure that people with RA/SpA/HOA/ KOA receive appropriate PA-interventions.	4	D	9.14 (0.98) 9 (7-10)
3.	PA interventions should be delivered by health care providers competent in their delivery to people with RA/SpA/HOA/KOA.	4	D	8.86 (1.48) 10 (5-10)
4.	Health care providers should evaluate the type, intensity, frequency, and duration of the people's actual physical activity by means of standardized methods to identify which of the four domains of general PA recommendations can be targeted for improvement.	3	С	9.05 (1.04) 9 (6-10)
5.	General and disease-specific contra-indications for PA should be identified and taken into account in the promotion of PA.	4	D	9.10 (1.41) 10 (5-10)
6.	PA interventions should have clear personalized aims, which should be evaluated over time, preferably by use of a combination of subjective and objective measures (including self-monitoring when appropriate).	4	D	9.05 (1.25) 9 (5-10)
7.	General and disease-specific barriers and facilitators related to performing PA, including knowledge, social support, symptom control, and self-regulation should be identified and addressed.	3	С	9.19 (1.13) 10 (6-10)
8.	Where individual adaptations to general PA recommendations are needed, these should be based on a comprehensive assessment of physical, social and psychological factors including fatigue, pain, depression, and disease activity.	4	D	9.24 (0.86) 9 (7-10)
9.	Health care providers should plan and deliver PA interventions that include the behavioural change techniques self-monitoring, goal setting, action planning, feedback and problem solving.	1A	A	9.48 (0.79) 10 (7-10)
10.	Health care providers should consider different modes of delivery of PA (eg, supervised/not-supervised, individual/group, face-to-face/ online, booster strategie) in line with people's preferences	4	D	9.00 (1.30) 9 (5-10)

RA, rheumatoid arthritis; SpA, Spondyloarthritis; HOA, hip osteoarthritis; KOA, knee osteoarthritis; OA, osteoarthritis; PA, physical activity



в.

Figure 1 Flowcharts of the literature search related to PICO\_1 (A) and PICO\_2 (B). ACSM, American College of Sports Medicine; OA, osteoarthritis; PA, physical activity; PICO, Population, Intervention, Comparison, Outcome; RA, rheumatoid arthritis; RCT, randomised controlled trial; SpA, spondyloarthritis; SR, systematic review.

#### Recommendation 1: PA as integral part of standard care

Given the evidence for effectiveness, feasibility and safety, the PH recommendations for PA are applicable, and thus, PA should be an integral part of standard care for people with RA/ SpA/HOA/KOA. PA according to PH recommendations (2) is effective on PA level, physical fitness as well as disease-specific and general outcomes in people with RA/SpA/HOA/ KOA (category I evidence (16)). Our MA including eleven RCTs (26, 35, 36, 42, 43, 50, 54, 56, 57, 61, 70) showed that cardiovascular exercises have a moderate beneficial effect on cardiovascular fitness (evaluated in VO2 max) in all three conditions. Our MA including 25 RCTs (28, 31, 38, 39, 47, 49, 50, 62-66, 72, 75-78, 81-83, 85, 86, 88, 90, 91) showed that muscle strength exercises have a moderate beneficial effect for muscle strength in people with RA and HOA/KOA. Our MA including seven RCTs (52, 55, 58, 78, 88, 90, 116) showed that combined exercises (aerobic or strength exercises plus flexibility exercises) had no effect on flexibility in people with SpA or HOA/KOA. However, exercise conditions, assessments and outcome measures varied greatly. There is no study comparing the effect of flexibility exercises alone versus no exercises. In one RCT (48) the effect of a neuromotor-exercise programme on neuromotor performance was investigated in people with RA showing a positive effect. Eleven RCTs described the promotion of daily PA. Our MA including six RCTs (95, 98, 101, 102, 104, 117) applying BCTs for the counselling intervention showed a small beneficial effect.

Feasibility of interventions can be captured by adherence to the intervention or the study protocol (118). Adherence to interventions (number of sessions attended/total number of sessions) has been reported in 26 RCTs (35%) and the mean adherence was 69% in people with SpA, 71% in people with RA and 79% in people with HOA/KOA. However, the (self-) reported adherence to intervention might be over-estimated due to recall bias or social desirability. In 68 RCTs (94%) protocol violations were reported, with approximately 10% of these being disease-related or intervention-related.

PH recommendations for PA can be considered safe. No detrimental effects were reported, rather beneficial effects on disease activity and symptoms in iA (7). Forty-four percent of all included RCTs reported on adverse events (AE), of those 62% described no AE and 38% describe minor AE such as transitional exercise related joint or muscle pain.

#### Recommendation 2: Responsibility for PA promotion

All HCPs should have a responsibility for PA promotion and collaborative working that facilitate a close cooperation between different professions to support appropriate disease management. This statement was based on the finding that 66% of the included studies reported the profession of the HCP providing the intervention, of which 75% were PTs (25, 31, 36, 40, 44, 45, 48, 50, 53, 55, 58, 61, 64-66, 70, 73-79, 81, 84, 87, 88, 91, 94, 96, 101-105, 119, 120). However, the functions and responsibilities of HCPs vary across Europe (121, 122). Therefore, the TF agreed that PA advice should be provided by all HCPs.

#### **Recommendation 3: Delivery of PA**

The delivery of interventions should be performed by HCPs competent in the field of PA principles and rheumatic conditions. The reporting of training on PA guidelines was rare. One study (59) described a '4-hours education session on cardiovascular training', others described the instructing person as 'trained' (25, 50, 69, 70, 84, 88, 123) or 'experienced' (31, 49, 76, 77, 88). Some studies with focus on the promotion of daily PA described training sessions on behaviour change skills like Motivational Interviewing (94, 96, 104).

#### **Recommendation 4: Evaluation of PA**

The PA level (active or non-active) and the exercise domains (cardiorespiratory, muscle strength, flexibility and neuromotor) should be routinely assessed. Of 11 trials investigating the effect of PA promotion interventions, three RCTs (94, 96, 105) described baseline screening to distinguish between active and non-active persons before starting the tailored PA-intervention. Specific tools are needed to assess each domain (15).<sup>p. 68</sup>.

#### Recommendation 5: General and disease-specific contra-indications

Tools for specific contra-indications (CIs) were found (94, 15, 124), however, available general or national guidelines defining absolute or relative CIs should be followed as a priority.

#### Recommendation 6: Personalised aims and evaluation

The PA-interventions should be based on individual aims, which should be regularly evaluated. This can be done by PA assessments and any other assessments related to the individual aims. As PA assessments, performance–based tests, patient reported outcomemeasures (eg, SQUASH (104), PASE (94)) and self-monitoring tools (eg, wearables such as Fitbit (100), pedometer (99) or accelerometer (101)) were identified. However, we did not evaluate the validity and reliability of the assessments applied.

#### Recommendation 7: General and disease-specific barriers and facilitators

General and disease-specific barriers (that are not CI per se) and facilitators should be addressed as described in 11 studies (11, 107-115, 125). Disease-specific barriers included lack of knowledge about the disease, lack of knowledge about safe exercising (both in people with iA/OA and HCPs,) and symptoms like pain, fatigue, stiffness, reduced mobility, fear of flare-ups or causing damage. Disease-specific facilitators included positive impact of exercise in symptoms or disease control, information about disease and correct exercising, the use medication for pain prior to exercising, using self-regulation techniques, supportive, but not controlling encouragement from HCPs and a supportive social background.

#### Recommendation 8: Individual adaptations to PA following individualised assessment

Adaptations to PA should be made on a comprehensive individual assessment. However, no evidence on the necessity of general adaptations in people with RA/SpA/HOA/KOA was found. In some RA studies the '24 hour-rule' was applied, that is, the exercise intensity was reduced when the increased pain persisted for more than 24 hours (23, 40, 50). ACSM

provides adaptations to exercise testing in people with arthritis, (eg, no high-intensity testing if acute inflammation) and training such as exercising when pain is typically least severe or to train carefully in order to reduce risk of associated injuries, although no clear evidence that high-impact activities cannot be engaged during active inflammation (15)- pp 298-301. Individual disease-related barriers (eg, symptoms) may determine these adaptations.

#### **Recommendation 9: Behaviour change techniques**

BCTs should be an integral component of PA-interventions. Several behaviour change theories were used in PA promotion interventions in the field of RA and HOA/KOA (4, 126), but the reporting was poor. Future research based on theories in design, evaluation and interpretation of findings is needed.

A meta-analysis of six RCTs (94, 95, 98, 101, 102, 104) investigating the effects of a PA promotion intervention according to general PA recommendations (2) and based on counselling interventions that apply BCTs showed a small beneficial effect on PA level (19). Counselling interventions show a small beneficial effect if BCTs are applied (19).

#### Recommendation 10: Modes of delivery

HCPs should consider the whole range of modes to deliver interventions. No evidence on the superiority of specific delivery modes was found. The delivery modes of PA-interventions vary considerably and are mostly described as 'land-based and/ or water-based' and 'supervised and individualised', the latter usually applied to group settings. As booster strategies phone calls (36, 96, 98, 105), devices (eg, pedometer (98, 99), wearable (100, 101)), home visits (63, 70), log book (36, 51, 76, 98), web-based instructions (127), written material (51, 54, 103), visual instructions (eg, video (103)) were reported.

#### Research and education agendas

Based on the gaps identified in the literature, the TF discussed and proposed a research agenda (box 1) with the prioritised research topics and an education agenda (box 2) with topics for education and training in PA promotion for HCPs. Evidence on impact of (reducing) sedentary behaviour emerged as an important future research topic.

#### Box 1 Research agenda for physical activity (PA) in people with inflammatory arthritis and osteoarthritis

- 1. To evaluate the long-term effectiveness of PA at different intensities and types and monitoring of Adverse Events (AE).
- 2. To evaluate links between PA behaviour and disease-specific outcomes.
- 3. To evaluate the long-term effectiveness of sedentary behavior reduction, including the monitoring of AE.
- 4. To evaluate links between sedentary behavior and disease-specific outcomes.
- 5. To identify which PA-intervention strategies work best to increase PA level and adherence in various subgroups.
- 6. To identify markers of response and non-response to PA treatment.
- 7. To identify disease-specific contra-indications on different exercise domains (cardiovascular, strength, flexibility, neuromotor).
- 8. To further develop and evaluate strategies to reduce and monitor a change in sedentary behavior.
- 9. To develop PA-interventions targeting all exercise dimensions simultaneously with special focus on feasibility.
- 10. To evaluate and recommend valid PA assessments feasible for the use in clinical practice.
- 11. To study how to facilitate PA behavior change immediately from screening onwards and how to address facilitators and barriers.
- 12. To identify facilitators and barriers of health care providers towards applying the PA recommendations.
- 13. To perform long-term effectiveness trials on combined interventions including other health behaviours.

#### Box 2 Education agenda for physical activity (PA) in people with inflammatory arthritis and osteoarthritis

- 1. Increase knowledge about PA among health professionals (HPs), physicians and people with inflammatory arthritis and osteoarthritis.
- 2. Increase HPs' and physicians' skills in communicating the role of PA in managing general health and disease-specific issues.
- 3. Include knowledge and skills on PA promotion in all HPs' and physicians' undergraduate training curricula.
- 4. Develop a EULAR training module on PA for HPs and rheumatologists.
- 5. Propose a session on PA at every EULAR congress.
- 6. Develop education materials for people with inflammatory arthritis and osteoarthritis.

## Discussion

The TF agreed on 4 overarching principles and 10 recommendations for PA in people with RA/SpA/HOA/KOA, which integrated the perspectives of the TF members from different professional, cultural and personal backgrounds. This led to a broad consensus on the principles and recommendations within the group and ought to foster its feasibility and practicability in the diverging health systems across Europe.

The LoA on the recommendations among the TF members was very high. The only exception was about the competency of HCP, which may be due to country specific differences in the availability of HCP competent in PA promotion.

Although the PH recommendations for PA are well established, the feasibility and applicability of these for people with iA and OA has not been assessed so far. Accordingly, the development of the recommendations was needed. Expectedly, they emphasise the importance of PA and will guide future PA-interventions in people with chronic rheumatic conditions.

PA promotion is a behavioural intervention and therefore BCT are central components in PAinterventions. Identifying effective and cost effective BCT within PA promotion intervention in people with chronic conditions is currently a hot topic in research and for example a research priority of the National Institute for Health and Care Excellence, UK (128).

We decided a priori to include only studies fulfilling the PH recommendations for PA according to ACSM principles (2). This was a far-reaching decision, which allowed drawing stronger conclusions on the effectiveness and especially the safety of correctly dosed PA-interventions. We followed a pragmatic search strategy with the plan to answer all RQs related to PICO 1 with findings of available SR/MA. However, there were no SR/MA on all exercise dimensions and all conditions available; this led to extracting single RCTs from high quality SR/ MA. This, however, excluded high-quality reviews (eg, Cochrane reviews) and RCTs that did not fulfil the ACSM principles and affected the potential to report 1A evidence according to Oxford levels of evidence (21). Furthermore, only one reviewer screened the abstracts and decided on unclear abstracts together with a second reviewer, which is not fully in line with standard procedures of a SR (129). However, we applied a double-check by experts to ensure that no relevant studies were missed.

A major problem for data extraction and interpretation was that the reporting of interventions in most studies was incomplete. Manuscripts that applied TIDieR (130) (Template for Intervention Description and Replication) guidelines reported more precisely the PA-interventions and substantially improved the objective evaluation of the PA-interventions. For the research questions related to the effectiveness and safety of PA-interventions and BCT the PICO scheme was applied, resulting in 1A level of evidence. All other research questions we had to answer in a descriptive way limiting the level of evidence to 3 to 4.
However, this limitation is due to the nature of the research questions. Nevertheless, the qualitative studies may provide valuable insight into important PA-related fields, such as assessments, barriers and facilitators, PA promotion strategies.

The recommendations focused on the conditions RA/SpA/HOA/KOA, the most prevalent RMD conditions to increase the generalisability and applicability of the recommendations. However, large heterogeneity between these conditions may limit the precision of the recommendations. Therefore, additional disease-specific recommendations are desirable. In addition, not all sub-conditions were considered and represented (eg, juvenile arthritis).

The research agenda highlights several areas where scientific evidence is lacking. It is a clear ambition to implement these recommendations into daily clinical routine. Due to the different health systems across Europe, development and evaluation of target group and culture-specific implementation strategies are needed and should involve all stakeholders.

#### Funding

The TF would like to thank EULAR for financial support of this work.

#### Acknowledgements

We thank the librarians Mrs. José Plevier and Mr. Jan W. Schoones, Walaeus Library, Leiden University Medical Center, the Netherlands, for supporting our literature search, and Christian Horvath, Zurich University of Applied Sciences, Switzerland, MSc student for his help in the Cochrane risk of bias assessment.

## Contributions

AR and KN contributed equally. AR was the research fellow for the project, undertaking the SLR. The fellow was supervised by the steering group consisting of KN (convenor), TVV (methodologist), JB (expert). KN and TPMVV supervised the process of the SLR. KN organized and chaired the TF meetings. AR and KN drafted the manuscript with advice from TPMVV and JB. All authors have contributed to the recommendations by participating in the TF meetings; during discussion and agreement on the recommendations; revising and approving the manuscript for publication.

#### Competing interests

None

## Additional files:

- Supplement 1: Research questions
- Supplement S2: PICOs
- Supplement S3: risk of bias assessment

# References

- Caspersen CJ, Powell KE, Christenson GM. Physical activity, exercise, and physical fitness: definitions and distinctions for health-related research. Public Health Rep. 1985;100(2):126-31.
- Garber CE, Blissmer B, Deschenes MR, Franklin BA, Lamonte MJ, Lee IM, et al. American College of Sports Medicine position stand. Quantity and quality of exercise for developing and maintaining cardiorespiratory, musculoskeletal, and neuromotor fitness in apparently healthy adults: guidance for prescribing exercise. Medicine and science in sports and exercise. 2011;43(7):1334-59.
- 3. Larkin LGS, Cramp F, Brand C, Fraser A, Kennedy N. Behaviour change interventions to promote physical activity in rheumatoid arthritis: a systematic review. Rheumatology international. 2015;35(10):1631-40.
- Keogh ATMA, Matthews J, Hurley DA. A review of behaviour change theories and techniques used in group based selfmanagement programmes for chronic low back pain and arthritis. Manual Therapy. 2015;20(6):727-35.
- 5. WHO. Global recommendations on physical activity for health. Geneve, 2010.
- Al Maini M, Adelowo F, Al Saleh J, Al Weshahi Y, Burmester GR, Cutolo M, et al. The global challenges and opportunities in the practice of rheumatology: white paper by the World Forum on Rheumatic and Musculoskeletal Diseases. Clinical rheumatology. 2015;34(5):819-29.
- Sveaas SH, Smedslund G, Hagen KB, Dagfinrud H. Effect of cardiorespiratory and strength exercises on disease activity in patients with inflammatory rheumatic diseases: a systematic review and meta-analysis. British journal of sports medicine. 2017;51(14):1065-72.
- Hernandez-Hernandez V, Ferraz-Amaro I, Diaz-Gonzalez F. Influence of disease activity on the physical activity of rheumatoid arthritis patients. Rheumatology (Oxford, England). 2014;53(4):722-31.
- 9. Swinnen TW, Scheers T, Lefevre J, Dankaerts W, Westhovens R, de Vlam K. Physical activity assessment in patients with axial spondyloarthritis compared to healthy controls: a technology-based approach. PloS one. 2014;9(2):e85309.
- de Groot IB, Bussmann JB, Stam HJ, Verhaar JA. Actual everyday physical activity in patients with end-stage hip or knee osteoarthritis compared with healthy controls. Osteoarthritis and cartilage. 2008;16(4):436-42.
- 11. Iversen MD, Scanlon L, Frits M, Shadick NA, Sharby N. Perceptions of physical activity engagement among adults with rheumatoid arthritis and rheumatologists. Int J Clin Rheumtol. 2015;10(2):67-77.
- Agca R, Heslinga SC, Rollefstad S, Heslinga M, McInnes IB, Peters MJ, et al. EULAR recommendations for cardiovascular disease risk management in patients with rheumatoid arthritis and other forms of inflammatory joint disorders: 2015/2016 update. Annals of the rheumatic diseases. 2017;76(1):17-28.
- van der Heijde D, Ramiro S, Landewe R, Baraliakos X, Van den Bosch F, Sepriano A, et al. 2016 update of the ASAS-EULAR management recommendations for axial spondyloarthritis. Annals of the rheumatic diseases. 2017;76(6):978-91.
- Fernandes L, Hagen KB, Bijlsma JW, Andreassen O, Christensen P, Conaghan PG, et al. EULAR recommendations for the non-pharmacological core management of hip and knee osteoarthritis. Annals of the rheumatic diseases. 2013;72(7):1125-35.
- 15. American College of Sports Medicine. ACSM's Guidelines for exercise testing and prescription. tenth edition ed. Philadelphia: Wolters Kluwer; 2018.
- van der Heijde D, Aletaha D, Carmona L, Edwards CJ, Kvien TK, Kouloumas M, et al. 2014 Update of the EULAR standardised operating procedures for EULAR-endorsed recommendations. Annals of the rheumatic diseases. 2015;74(1):8-13.
- 17. Brouwers MC, Kho ME, Browman GP, Burgers JS, Cluzeau F, .., et al. AGREE II: Advanced guideline developement, reporting and evaluation in healthcare. CMAJ. 2010;182:E839-42.
- 18. Health P. PICO Framework, 2017
- Rausch Osthoff AK, Juhl C, Knittle K, Dagfinrud H, Hurkmans E, Braun J, et al. Effects of exercise and physical activity promotion: meta-analysis informing the 2018 EULAR recommendations for physical activity in people with rheumatoid arthritis, spondyloarthritis and hip/knee osteoarthritis. RMD Open Rheumatic &Musculoskeletal Diseases 2018; 4 (2): e000713
- 20. Higgins JGS. Cochrane Handbook for Systematic Reviews of Interventions. Version 5.1.0. Collaboration TC, editor2011.
- 21. Group OLoEW. The Oxford Levels of Evidence 2. Oxford Centre for Evidence-Based Medicine. 2011.
- Bartholdy C, Juhl C, Christensen R, Lund H, Zhang W, Henriksen M. The role of muscle strengthening in exercise therapy for knee osteoarthritis: A systematic review and meta-regression analysis of randomized trials. Seminars in arthritis and rheumatism. 2017; 47 (1):9-21
- Swardh EBN. Effects of aerobic and muscle strengthening exercise in adults with rheumatoid arthritis: A narrative review summarising a chapter in Physical activity in the prevention and treatment of disease (FYSS 2016). British journal of sports medicine. 2016;50(6):362-7.
- 24. Moseng T, Dagfinrud H, Smedslund G, Osteras N. The importance of dose in land-based supervised exercise for people with hip osteoarthritis. A systematic review and meta-analysis. Osteoarthritis and cartilage. 2017; 25(10): 1563-1576
- Baillet A, Payraud E, Niderprim VA, Nissen MJ, Allenet B, Francois P, et al. A dynamic exercise programme to improve patients' disability in rheumatoid arthritis: a prospective randomized controlled trial. Rheumatology. 2009;48(4):410-5.
- Baslund BLK, Andersen V, Halkjaer Kristensen J, Hansen M, Klokker M, Pedersen BK. Effect of 8 wk of bicycle training on the immune system of patients with rheumatoid arthritis. Journal of applied physiology. 1993;75(4):1691-5.

- De Jong ZM M, Zwinderman AH, Kroon HM, Jansen A, Ronday KH, Van Schaardenburg D, Dijkmans BAC, Van den Ende CHM, Breedveld FC, Vliet Vlieland TPM, Hazes JMW. Is a long-term high-intensity exercise program effective and safe in patients with rheumatoid arthritis? Results of a randomized controlled trial. Arthritis and rheumatism. 2003;48(9):2415-24.
- 28. de Jong ZMM, Kroon HM, van Schaardenburg D, Dijkmans BA, Hazes JM, Vliet Vlieland TPM. Long-term follow-up of a high-intensity exercise program in patients with rheumatoid arthritis. Clinical rheumatology. 2009;28(6):663-71.
- 29. Durcan LWF, Cunnane G. The effect of exercise on sleep and fatigue in rheumatoid arthritis: A randomised controlled study. Irish Journal of Medical Science. 2014;1):S110-S1.
- Flint-Wagner HG, Lisse J, Lohman TG, Going SB, Guido T, Cussler E, et al. Assessment of a sixteen-week training program on strength, pain, and function in rheumatoid arthritis patients. Journal of clinical rheumatology : practical reports on rheumatic & musculoskeletal diseases. 2009;15(4):165-71.
- 31. Hakkinen A. Effectiveness and safety of strength training in rheumatoid arthritis. Current Opinion in Rheumatology. 2004;16(2):132-7.
- 32. Hakkinen AHK, Hannonen P. Effects of strength training on neuromuscular function and disease activity in patients with recent-onset inflammatory arthritis. Scandinavian journal of rheumatology. 1994;23(5):237-42.
- Hakkinen AS, T.; Kotaniemi, A.; Hannonen, P. A randomized two-year study of the effects of dynamic strength training on muscle strength, disease activity, functional capacity, and bone mineral density in early rheumatoid arthritis. Arthritis and rheumatism. 2001;44(3):515-22.
- Hansen TM, Hansen G, Langgaard AM, Rasmussen JO. Longterm physical training in rheumatoid arthritis. A randomized trial with different training programs and blinded observers. Scandinavian journal of rheumatology. 1993;22(3):107-12.
- Harkcom TM, Lampman RM, Banwell BF, Castor CW. Therapeutic value of graded aerobic exercise training in rheumatoid arthritis. Arthritis and rheumatism. 1985;28(1):32-9.
- 36. Hsieh LCS, Chuang C, Chai Hm, Chen W, He Y. Supervised aerobic exercise is more effective than home aerobic exercise in female chinese patients with rheumatoid arthritis. Journal of rehabilitation medicine. 2009; 41(5):332-7
- 37. Janse van Rensburg DCKJA, Grant CC, Fletcher L. Effect of exercise on cardiac autonomic function in females with rheumatoid arthritis. Clinical rheumatology. 2012;31(8):1155-62.
- Komatireddy GR, Leitch RW, Cella K, Browning G, Minor M. Efficacy of low load resistive muscle training in patients with rheumatoid arthritis functional class II and III. The Journal of rheumatology. 1997;24(8):1531-9.
- Lemmey ABMSM, Chester K, Wilson S, Casanova F, Maddison PJ. Effects of high-intensity resistance training in patients with rheumatoid arthritis: a randomized controlled trial. Arthritis and rheumatism. 2009;61(12):1726-34.
- Lyngberg KK, Harreby M, Bentzen H, Frost B, Danneskiold-Samsoe B. Elderly rheumatoid arthritis patients on steroid treatment tolerate physical training without an increase in disease activity. Archives of physical medicine and rehabilitation. 1994;75(11):1189-95.
- 41. Melikoglu MA, Karatay S, Senel K, Akcay F. Association between dynamic exercise therapy and IGF-1 and IGFBP-3 concentrations in the patients with rheumatoid arthritis. Rheumatology international. 2006;26(4):309-13.
- 42. Minor MA, Hewett JE, Webel RR, Anderson SK, Kay DR. Efficacy of physical conditioning exercise in patients with rheumatoid arthritis and osteoarthritis. Arthritis and rheumatism. 1989;32(11):1396-405.
- Neuberger GBALS, Gajewski B, Embretson SE, Cagle PE, Loudon JK, Miller PA. Predictors of exercise and effects of exercise on symptoms, function, aerobic fitness, and disease outcomes of rheumatoid arthritis. Arthritis and rheumatism. 2007;57(6):943-52.
- 44. Sanford Smith S, MaxcKay-Lyons M, Nunes-Clement S. Therapeutic benefits of aquaerobics for individuals with rheumatoid arthritis. Physiotherapy Canada. 1998;50:40.6.
- 45. Seneca THEM, Maribo T. Comparable effect of partly supervised and self-administered exercise programme in early rheumatoid arthritis--a randomised, controlled trial. Danish medical journal. 2015;62(8):A5127.
- 46. Shapoorabadi YJVB, Salesi M, Ramezanian H. Effects of aerobic exercise on hematologic indices of women with rheumatoid arthritis: A randomized clinical trial. Journal of research in medical sciences: the official journal of Isfahan University of Medical Sciences. 2016;21:9.
- Siqueira USOVLG, de Mello MT, Szejnfeld VL, Pinheiro MM. Effectiveness of Aquatic Exercises in Women With Rheumatoid Arthritis: A Randomized, Controlled, 16-Week Intervention-The HydRA Trial. American journal of physical medicine & rehabilitation. 2017;96(3):167-75.
- da Silva KN, Teixeira LE, Imoto AM, Atallah AN, Peccin MS, Trevisani VF. Effectiveness of sensorimotor training in patients with rheumatoid arthritis: a randomized controlled trial. Rheumatology international. 2013;33(9):2269-75.
- 49. Strasser BLG, Strehblow C, Schobersberger W, Haber P, Cauza E. The effects of strength and endurance training in patients with rheumatoid arthritis. Clinical rheumatology. 2011;30(5):623-32.
- van den Ende CHHJM, le Cessie S, Mulder WJ, Belfor DG, Breedveld FC, Dijkmans BA. Comparison of high and low intensity training in well controlled rheumatoid arthritis. Results of a randomised clinical trial. Annals of the rheumatic diseases. 1996;55(11):798-805.
- Westby MDWJP, Rangno KK, Berkowitz J. A randomized controlled trial to evaluate the effectiveness of an exercise program in women with rheumatoid arthritis taking low dose prednisone. The Journal of rheumatology. 2000;27(7):1674-80.
- 52. Altan LKN, Dizdar M, Yurtkuran M. Effect of Pilates training on people with ankylosing spondylitis. Rheumatology international. 2012;32(7):2093-9.

- Fang HCW, Pan Y, Wu D, Liang L. Six-month home-based exercise and supervised training in patients with ankylosing spondylitis. International Journal of Clinical and Experimental Medicine. 2016;9(3):6635-41.
- Hsieh LFCCC. Tseng CS, Wei JC, Hsu WC, Lin YJ. Combined home exercise is more effective than range-of-motion home exercise in patients with ankylosing spondylitis: a randomized controlled trial. BioMed research international. 2014;2014;398190.
- Ince GST, Durgun B, Erdogan S. Effects of a multimodal exercise program for people with ankylosing spondylitis. Physical therapy. 2006;86(7):924-35.
- Jennings FOHA, De Souza MC, Cruz VDG, Natour J. Effects of aerobic training in patients with ankylosing spondylitis. Journal of Rheumatology. 2015;42(12):2347-53.
- Karapolat H, Eyigor S, Zoghi M, Akkoc Y, Kirazli Y, Keser G. Are swimming or aerobic exercise better than conventional exercise in ankylosing spondylitis patients? A randomized controlled study. European journal of physical and rehabilitation medicine. 2009;45(4):449-57.
- Kjeken I, Bo I, Ronningen A, Spada C, Mowinckel P, Hagen KB, et al. A three-week multidisciplinary in-patient rehabilitation programme had positive long-term effects in patients with ankylosing spondylitis: randomized controlled trial. Journal of rehabilitation medicine. 2013;45(3):260-7.
- Niedermann K, Muggli C, Dagfinrud H, Hermann M, Tamborrini G, Ciurea A, Bischoff-Ferrari H. Effect of cardiovascular training on fitness and perceived disease activity in people with ankylosing spondylitis. Arthritis care & research. 2013;65(11):1844-52.
- Rosu MOTI, Chirieac R, Ancuta C. Effects of Pilates, McKenzie and Heckscher training on disease activity, spinal motility and pulmonary function in patients with ankylosing spondylitis: a randomized controlled trial. Rheumatology international. 2014;34(3):367-72.
- Sveaas SHBIJ, Provan SA, Semb AG, Hagen KB, Vollestad N, Fongen C, Olsen IC Michelsen A, Ueland T, Aukrust P, Kvien TK, Dagfinrud H. Efficacy of high intensity exercise on disease activity and cardiovascular risk in active axial spondyloarthritis: a randomized controlled pilot study. PloS one. 2014;9(9):e108688.
- 62. Anwer SAA. Effect of isometric quadriceps exercise on muscle strength, pain, and function in patients with knee osteoarthritis: A randomized controlled study. Journal of Physical Therapy Science. 2014;26(5):745-8.
- 63. Baker KRNME, Felson DT, Layne JE, Sarno R, Roubenoff R. The efficacy of home based progressive strength training in older adults with knee osteoarthritis: A randomized controlled trial. Journal of Rheumatology. 2001;28(7):1655-65.
- Bennell KLET, Pua YH, Abbott JH, Sims K, Metcalf B, McManus F, Wrigley TV, Forbes A, Harris A, Buchbinder R. Efficacy of a multimodal physiotherapy treatment program for hip osteoarthritis: A randomised placebo-controlled trial protocol. BMC musculoskeletal disorders. 2010;11: 238.
- Borjesson MRE, Weidenhielm L, Mattsson E, Olsson E. Physiotherapy in knee osteoarthrosis: effect on pain and walking. Physiotherapy research international : the journal for researchers and clinicians in physical therapy. 1996;1(2):89-97.
- Bruce-Brand RAWRJ, Ong JC, Emerson BS, O'Byrne JM, Moyna NM. Effects of home-based resistance training and neuromuscular electrical stimulation in knee osteoarthritis: a randomized controlled trial. BMC musculoskeletal disorders. 2012;13:118.
- Cheung CWJ, Bronas U, McCarthy T, Rudser K, Mathiason MA. Managing knee osteoarthritis with yoga or aerobic/ strengthening exercise programs in older adults: a pilot randomized controlled trial. Rheumatology International. 2017;37(3):389-98.
- Cheung CWF, Resnick B, Savik K. Yoga for managing knee osteoarthritis in older women: a pilot randomized controlled trial. BMC complementary and alternative medicine. 2014;14:160.
- da Silva FS, de Melo FE, do Amaral MM, Caldas VV, Pinheiro IL, Abreu BJ, et al. Efficacy of simple integrated group rehabilitation program for patients with knee osteoarthritis: Single-blind randomized controlled trial. J Rehabil Res Dev. 2015;52(3):309-22.
- 70. Ettinger WHBR, Messier S, Applegate W, Rejeski WJ, Morgan T, Shumaker S, Berry MJ, Otoole M, Monu J, Craven T. A randomized trial comparing aerobic exercise and resistance exercise with a health education program in older adults with knee osteoarthritis - The Fitness Arthritis and Seniors Trial (FAST). JAMA-J Am Med Assoc. 1997;277(1):25-31.
- Fernandes LSK, Sandvik L, Nordsletten L., Risberg MA. Efficacy of patient education and supervised exercise vs patient education alone in patients with hip osteoarthritis: A single blind randomized clinical trial. Osteoarthritis and cartilage. 2010;18(10):1237-43.
- Foroughi NSR, Lange AK, Baker MK, Fiatarone Singh MA, Vanwanseeele B. Lower limb muscle strengthening does not change frontal plane moments in women with knee osteoarthritis: a randomized controllled trial. Clinical biomechanics. 2011;26:167-74.
- Fransen MNL, Winstanley J, Lam P, Edmonds J. Physical activity for ostoarthritis management: a randomized controlled clinical trial evaluating hydrotherapy or tai chi classes. Arthrits & Rheumatism. 2007;57(3):407-14.
- 74. Henriksen M, Klokker L, Graven-Nielsen T, Bartholdy C, Schjodt Jorgensen T, Bandak E, et al. Association of exercise therapy and reduction of pain sensitivity in patients with knee osteoarthritis: a randomized controlled trial. Arthritis care & research. 2014;66(12):1836-43.
- 75. Hermann AH-LA, Zerahn B, Mejdahl S, Overgaard S. Preoperative progressive explosive-type resistance training is feasible and effective in patients with hip osteoarthritis scheduled for total hip arthroplasty--a randomized controlled trial. Osteoarthritis and cartilage. 2016;24(1):91-8.
- 76. Jan MHLJJ, Liau JJ, Lin YF, Lin DH. Investigation of clinical effects of high- and low-resistance training for patients with

knee osteoarthritis: A randomized controlled trial. Physical therapy. 2008;88(4):427-36.

- 77. Jorge RT, Souza MC, Chiari A, Jones A, Fernandes Ada R, Lombardi Junior I, et al. Progressive resistance exercise in women with osteoarthritis of the knee: a randomized controlled trial. Clinical rehabilitation. 2015;29(3):234-43.
- Juhakoski RS, Malmivaara A, Kiviniemi V, Anttonen T, Arokoski JP. A pragmatic randomized controlled study of the effectiveness and cost consequences of exercise therapy in hip osteoarthritis. Clinical rehabilitation. 2011;25(4):370-83.
- Kim HST, Saito K, Kim M, Kojima N, Ishizaki T, Yamashiro Y, Hosoi E, Yoshida H. Effectiveness of exercise with or without thermal therapy for community-dwelling elderly Japanese women with non-specific knee pain: a randomized controlled trial. Archives of gerontology and geriatrics. 2013;57(3):352-9.
- Lee HPH, Chae Y. Tai Chi Qigong for the quality of life of patients with knee osteoarthritis: a pilot, randomized, waiting list controlled trial. Clinical Rehabilitation. 2009;23:504-11.
- Lim JYTE, Jang SN. Effectiveness of aquatic exercise for obese patients with knee osteoarthritis: a randomized controlled trial. PM & R : the journal of injury, function, and rehabilitation. 2010;2(8):723-31; quiz 93.
- Lin DH, Lin CH, Lin YF, Jan MH. Efficacy of 2 non-weight-bearing interventions, proprioception training versus strength training, for patients with knee osteoarthritis: a randomized clinical trial. The Journal of orthopaedic and sports physical therapy. 2009;39(6):450-7.
- 83. Mikesky AEMSA, Brandt KD, Perkins SM, Damush T, Lane KA. Effects of strength training on the incidence and progression of knee osteoarthritis. Arthritis Care and Research. 2006;55(5):690-9.
- 84. Rogind HB-NB, Jensen B, Moller HC, Frimodt-Moller H, Bliddal H. The effects of a physical training program on patients with osteoarthritis of the knees. Archives of physical medicine and rehabilitation. 1998;79(11):1421-7.
- Salli ASN, Baskent A, Ugurlu H. The effect of two exercise programs on various functional outcome measures in patients with osteoarthritis of the knee: A randomized controlled clinical trial. Isokinetics and Exercise Science. 2010;18(4):201-9.
- Sayers SPGK, Cook CR. Effect of high-speed power training on muscle performance, function, and pain in older adults with knee osteoarthritis: A pilot investigation. Arthritis Care and Research. 2012;64(1):46-53.
- Steinhilber BHG, Miller R, Janssen P, Krauss I. Exercise therapy in patients with hip osteoarthritis: Effect on hip muscle strength and safety aspects of exercise-results of a randomized controlled trial. Modern rheumatology. 2016:1-10.
- Svege IFL, Nordsletten L, Holm I, Risberg MA. Long-Term Effect of Exercise Therapy and Patient Education on Impairments and Activity Limitations in People With Hip Osteoarthritis: Secondary Outcome Analysis of a Randomized Clinical Trial. Physical therapy. 2016;96(6):818-27.
- Tsai PFCJY, Beck C, Kuo YF, Keefe FJ. A pilot cluster-randomized trial of a 20-week Tai Chi program in elders with cognitive impairment and osteoarthritic knee: effects on pain and other health outcomes. Journal of pain and symptom management. 2013;45(4):660-9.
- 90. Wang TJBB, Thompson EF, Whitney JD, Bennett K. Effects of aquatic exercise on flexibility, strength and aerobic fitness in adults with osteoarthritis of the hip or knee. Journal of advanced nursing. 2007;57(2):141-52.
- Weng MCLCL, Chen CH, Hsu JJ, Lee WD, Huang MH, Chen TW. Effects of different stretching techniques on the outcomes of isokinetic exercise in patients with knee osteoarthritis. The Kaohsiung journal of medical sciences. 2009;25(6):306-15.
- 92. Munneke MDJZ, Zwinderman AH, Ronday HK, Van Schaardenburg D, Dijkmans BAC, Kroon HM, Vliet Vlieland TPM, Hazes JMW. Effect of a high-intensity weight-bearing exercise program on radiologic damage progression of the large joints in subgroups of patients with rheumatoid arthritis. Arthritis Care and Research. 2005;53(3):410-7.
- Jan MHLCH, Lin YF, Lin JJ, Lin DH. Effects of weight-bearing versus nonweight-bearing exercise on function, walking speed, and position sense in participants with knee osteoarthritis: a randomized controlled trial. Archives of physical medicine and rehabilitation. 2009;90(6):897-904.
- Bennell KLCPK, Egerton T, Metcalf B, Kasza J, Forbes A, Bills C, Gale J, Harris A, Kolt GS, Bunker SJ, Hunter DJ, Brand CA, Hinman RS. Telephone Coaching to Enhance a Home-Based Physical Activity Program for Knee Osteoarthritis: A Randomized Clinical Trial. Arthritis Care and Research. 2017;69(1):84-94.
- Bieler TSV, Magnusson SP, Kjaer M, Beyer N. Even in the long run nordic walking is superior to strength training and home based exercise for improving physical function in older people with hip osteoarthritis-an RCT. Annals of the rheumatic diseases. 2016;75:55.
- Brodin N, Jensen E, Nisell R, Opava C.Coaching patients with early rheumatoid arthritis to healthy physical activity: a multicenter, randomized, controlled study. Arthritis and rheumatism. 2008; 59(3):325-31
- 97. Sjöquist EBN, Lampa J, Jensen I, Opava C. Physical activity coaching of patients with rheumatoid arthritis in everyday practice: a long-term follow-up. Musculoskeletal care. 2011; 9(2):75-85
- 98. Brosseau LWGA, Kenny GP, Reid R, Maetzel A, Tugwell P, Huijbregts M, McCullough C, De Angelis G, Chen L. The implementation of a community-based aerobic walking program for mild to moderate knee osteoarthritis: a knowledge translation randomized controlled trial: part II: clinical outcomes. BMC public health. 2012;12:1073.
- Hiyama Y, Yamada M, Kitagawa A, Tei N, Okada S. A four-week walking exercise programme in patients with knee osteoarthritis improves the ability of dual-task performance: a randomized controlled trial. Clinical rehabilitation. 2012;26(5):403-12.
- Katz P, Margaretten M, Gregorich S, Trupin L. Physical Activity to Reduce Fatigue in Rheumatoid Arthritis: A Randomized, Controlled Trial. Arthritis care & research. 2017.
- Li LC, Sayre EC, Xie H, Clayton C, Feehan LM. A Community-Based Physical Activity Counselling Program for People With Knee Osteoarthritis: Feasibility and Preliminary Efficacy of the Track-OA Study. JMIR Mhealth Uhealth. 2017;5(6):e86.

- 102. O'Dwyer T, Monaghan A, Moran J, O'Shea F, Wilson F. Behaviour change intervention increases physical activity, spinal mobility and quality of life in adults with ankylosing spondylitis: a randomised trial. Journal of physiotherapy. 2017;63(1):30-9.
- Rodriguez-Lozano C, Juanola X, Cruz-Martinez J, Pena-Arrebola A, Mulero J, Gratacos J, et al. Outcome of an education and home-based exercise programme for patients with ankylosing spondylitis: a nationwide randomized study. Clinical and experimental rheumatology. 2013;31(5):739-48.
- Knittle KDGV, Hurkmans E, Peeters A, Ronday K, Maes S, Vlieland Vlieland TPM. Targeting motivation and self-regulation to increase physical activity among patients with rheumatoid arthritis: a randomised controlled trial. Clinical rheumatology. 2015;34(2):231-8.
- 105. van den Berg MH, Ronday HK, Peeters AJ, Voogt-van der Harst EM, Munneke M, Breedveld FC, et al. Engagement and satisfaction with an Internet-based physical activity intervention in patients with rheumatoid arthritis. Rheumatology. 2007;46(3):545-52.
- 106. Hurkmans E, Vliet Vlieland TPM, Schoones J, Van den Ende EC. Dynamic exercise programs (aerobic capacity and/ or muscle strength training) in patients with rheumatoid arthritis. The Cochrane database of systematic reviews. 2009(4):Cd006853.
- 107. Veldhuijzen van Zanten JJCSR, Peter C, Hale ED, Ntoumanis N, Metsios GS, Duda JL, Kitas GD. Perceived barriers, facilitators and benefits for regular physical activity and exercise in patients with rheumatoid arthritis: A review of the literature. Sports Medicine. 2015;45(10):1401-12.
- Baxter SSC, Treharne G, Stebbings S, Hale L. What are the perceived barriers, facilitators and attitudes to exercise for women with rheumatoid arthrtis? A qualitative study. Disability and rehabilitation. 2016;38(8):773-80.
- 109. Halls S, Law RJ, Jones JG, Markland DA, Maddison PJ, Thom JM. Health Professionals' Perceptions of the Effects of Exercise on Joint Health in Rheumatoid Arthritis Patients. Musculoskeletal care. 2017; 15(3):196-209
- Larkin L, Kennedy N, Fraser A, Gallagher S. 'It might hurt, but still it's good': People with rheumatoid arthritis beliefs and expectations about physical activity interventions. J Health Psychol. 2017; 22(13):1678-1690
- O'Dwyer T, McGowan E, O'Shea F, Wilson F. Physical Activity and Exercise: Perspectives of Adults With Ankylosing Spondylitis. Journal of physical activity & health. 2016;13(5):504-13.
- 112. Dobson FBKL, French SD, Nicolson PJ, Klaasman RN, Holden MA, Atkins L, Hinman RS. Barriers and Facilitators to Exercise Participation in People with Hip and/or Knee Osteoarthritis: Synthesis of the Literature Using Behavior Change Theory. American journal of physical medicine & rehabilitation / Association of Academic Physiatrists. 2016;95(5):372-89.
- 113. Carmona-Teres VM-Q, Pujol-Ribera E, Lumillo-Gutierrez I, Mas, X., et al. Understanding knee osteoarthrtis from the patients' perspective: a qualitative study. BMC musculoskeletal disorders. 2017;18.
- 114. Petursdottir U, Arnadottir SA, Halldorsdottir S. Facilitators and barriers to exercising among people with osteoarthritis: a phenomenological study. Physical therapy. 2010;90(7):1014-25.
- 115. Loeppenthin K, Esbensen B, Ostergaard M, Jennum P, Thomsen T, Midtgaard J. Physical activity maintenance in patients with rheumatoid arthritis: a qualitative study. Clinical rehabilitation. 2014;28(3):289-99.
- 116. Fang FFHG, Zhou F, Li D, Chen HD, Gu W. Physical therapy in grassroot military medical units of pla: Current situation. Academic Journal of Second Military Medical University. 2015;36(1):65-8.
- 117. Nicolson PJA, Hinman RS, French SD, Lonsdale C, Bennell KL. Improving adherence to exercise: Do people with knee osteoarthritis and physical therapists agree on the behavioural approaches likely to succeed? Arthritis care & research. 2018; 70(3):388-397
- 118. Keogh JWL, Grigg J, Vertullo CJ. Is Home-Based, High-Intensity Interval Training Cycling Feasible and Safe for Patients With Knee Osteoarthritis?: Study Protocol for a Randomized Pilot Study. Orthopaedic Journal of Sports Medicine. 2017;5(3).
- 119. Karapolat HAY, Sari I, Eyigor S, Akar S, Kirazli Y, Akkoc N. Comparison of group-based exercise versus home-based exercise in patients with ankylosing spondylitis: effects on Bath Ankylosing Spondylitis Indices, quality of life and depression. Clinical rheumatology. 2008;27(6):695-700.
- 120. Bieler TMSP, Kjaer M, Beyer N. Supervised strength training, NORDIC walking or unsupervised home based exercise in older people with hip osteoarthritis? A randomized trial. Annals of the Rheumatic Diseases Conference: Annual European Congress of Rheumatology of the European League Against Rheumatism, EULAR. 2014;73.
- 121. van Eijk-Hustings Y, van Tubergen A, Bostrom C, Braychenko E, Buss B, Felix J, et al. EULAR recommendations for the role of the nurse in the management of chronic inflammatory arthritis. Annals of the rheumatic diseases. 2012;71(1):13-9.
- 122. Stamm T, Hill J. Extended roles of non-physician health professionals and innovative models of care within Europe: results from a web-based survey. Musculoskeletal care. 2011;9(2):93-101.
- 123. Bennell KLHMA, Wrigley TV, Hunter DJ, McManus FJ, Hodges PW, Li L, Hinman RS. Hip strengthening reduces symptoms but not knee load in people with medial knee osteoarthritis and varus malalignment: A randomised controlled trial. Osteoarthritis and cartilage. 2010;18(5):621-8.
- 124. Riebe D, Franklin BA, Thompson PD, Garber CE, Whitfield GP, Magal M, et al. Updating ACSM's Recommendations for Exercise Preparticipation Health Screening. Medicine and science in sports and exercise. 2015;47(11):2473-9.
- Larkin L, Kennedy N, Gallagher S. Promoting physical activity in rheumatoid arthritis: a narrative review of behaviour change theories. Disability and rehabilitation. 2015;37(25):2359-66.
- 126. Demmelmaier I, Iversen MD. How Are Behavioral Theories Used in Interventions to Promote Physical Activity in Rheumatoid Arthritis? A systematic review. Arthritis care & research. 2018; 70(2):185-196.

- 127. Hurkmans EJ, van den Berg MH, Ronday KH, Peeters AJ, le Cessie S, Vlieland TP. Maintenance of physical activity after Internet-based physical activity interventions in patients with rheumatoid arthritis. Rheumatology (Oxford, England). 2010;49(1):167-72.
- 128. NICE. Behaviour change: individual approaches. Available from: https://www.nice.org.uk/guidance/ph49/chapter/5-Recommendations-for-research#behaviour-change-techniques-2 2014
- 129. Higgins JP, Green S. Cochrane Handbook for systematic reviews of interventions. Available from: http://handbook. cochrane.org: The Cochrane Collaboration; 2011
- Hoffmann TC, Glasziou PP, Boutron I, Milne R, Perera R, Moher D, et al. Better reporting of interventions: template for intervention description and replication (TIDieR) checklist and guide. BMJ 2014;348:g1687.

2

# 3

Effects of exercise and physical activity promotion: meta-analysis informing the 2018 EULAR recommendations for physical activity in people with rheumatoid arthritis, spondyloarthritis and hip/knee osteoarthritis

Anne-Kathrin Rausch Osthoff, Carsten Bogh Juhl, Keegan Knittle, Hanne Dagfinrud, Emalie Hurkmans, Jürgen Braun, Jan Schoones, Theodora P M Vliet Vlieland, Karin Niedermann

RMD Open Rheumatic &Musculoskeletal Diseases 2018; 4 (2): e000713 doi: 10.1136/rmdopen-2018-000713

Oral presentations at EULAR Congress in Amsterdam (June 2018) and World Physiotherapy Congress in Geneva (May 2019)

# Abstract

## Objective

To evaluate the effectiveness of exercise and physical activity (PA) promotion on cardiovascular fitness, muscle strength, flexibility, neuromotor performance (eg, balance) and daily PA in people with rheumatoid arthritis (RA), spondyloarthritis (SpA) and hip/knee osteoarthritis (HOA/KOA).

## Methods

Systematic review (SR) and meta-analysis (MA) were performed searching the databases PubMed/Medline, CENTRAL, Embase, Web of Science, Emcare, PsycInfo until April 2017. We included randomized controlled trials (RCTs) in adults (≥ 18 years) with RA, SpA and HOA/KOA, investigating the effects of exercise or PA promotion according to the public health PA recommendations by the American College of Sports Medicine. The time point of interest was the first assessment after the intervention period. If suitable, data were pooled in a MA using a random effects model presented as standardised mean difference (SMD).

## Results

The SR included 63 RCTs, of which 49 (3909 people with RA/SpA/HOA/KOA) were included in the MA. Moderate effects were found of aerobic exercises and resistance training on cardiovascular fitness (SMD 0.56 (95% CI 0.38 to 0.75)) and muscle strength (SMD 0.54 (95% CI 0.35 to 0.72)), respectively, but no effect of combined strength/aerobic/flexibility exercises on flexibility (SMD 0.12 (95% CI -0.16 to 0.41)). PA promotion interventions produced a small increase in PA behaviour (SMD 0.21 (95% CI 0.03 to 0.38)).

# Conclusion

Exercises and PA promotion according to public health recommendations for PA improved cardiovascular fitness, muscle strength, and PA behaviour, with moderate effect sizes in people with SpA, RA and HOA/KOA.

Trial registration number CRD42017082131.

What is already known about this subject?

• There are public health recommendations for physical activity in healthy adults of all ages. Exercise is a cornerstone for the management of rheumatoid arthritis (RA), spondyloarthritis (SpA) and hip/knee osteoarthritis (HOA/KOA), but exercise interventions may not meet these recommendations.

What does this study add?

- Exercise interventions according to the public health recommendations in appropriate dose (frequency, intensity, time, type, volume, progression) is effective on cardiorespiratory fitness and muscle strength in people with RA, SpA and HOA/ KOA.
- Counselling-based physical activity promotion interventions according to the public health recommendations effectively change physical activity behaviour in people with RA, SpA and HOA/KOA.
- There is a knowledge gap concerning the effects of flexibility and neuromotor exercises in people with RA, SpA and HOA/KOA.

How might this impact on clinical practice?

• Exercise and physical activity promotion according to the public health recommendations is effective and should be an integral part of standard care throughout the course of the disease in people with RA, SpA and HOA/KOA.

# Introduction

Physical Activity (PA) is defined as 'any bodily movement produced by skeletal muscles that result in energy expenditure' (1, 2) above resting (basal) levels. PA encompasses exercise, sports and physical activities, done as part of daily living, occupation, leisure, and active transportation (3). Exercise is a subcategory of 'physical activity that is planned, structured, and repetitive and (that) has as a final or intermediate objective to improve or maintain one or more dimensions of physical fitness' (1). In contrast, therapeutic exercises are specific exercises meant to address particular functional health problems.

WHO recommends that adults, including elderly or people with chronic conditions, perform at least 150 min of moderate-intensity or 75 min of vigorous-intensity PA a week (or a combination of it) and additional muscle strengthening activities involving major muscle groups on at least 2 days a week (4). The American College of Sports Medicine (ACSM) recommends additional regular performance of flexibility and neuromotor activities such as balance or agility on at least 2 days a week (3). Accordingly, exercising in the four dimensions cardiovascular fitness, muscle strength, flexibility and neuromotor performance is recommended. The absolute energy demands of PA can be estimated by MET (metabolic equivalent of task) (5): PA like walking 5 km/h is described by 3.2 METs, which is classified as a moderate-intensity activity; PA like jogging 9 km/h is described by 8.8 METs. Following Garber and colleagues (3), the use of absolute measures may however be misleading because personal factors like weight, sex or fitness level are not considered. For this reason, a relative measure of intensity (ie, the energy costs relative to the individual maximal capacity) like %METmax, %HRmax or perceived exertion are more appropriate. Moderate exercise is defined as 64 to 76% of the maximum heart rate (%HR<sub>max</sub>), vigorous exercise is defined as 77% to 95  $\mathrm{\% HR}_{\mathrm{max}}.$  Furthermore, there is evidence that shorter (eg, 30 min) vigorous intensity (> 6METs) of cardiovascular exercise is associated with a greater reduction of cardiovascular risk (6) and lower mortality rate in the general population (7) than longer (eg, 60 min) of moderate intensity (3-6METs) cardiovascular exercise.

The dose-response relationship between daily PA and health benefits was proven; by exceeding the minimum recommended amount of PA, the results regarding improvement of fitness, reduction of risk factors of chronic conditions, and prevention of unhealthy weight gain are enhanced (3).

The evidence for the benefits of regular PA in healthy people is overwhelming (8, 9). The benefits of PA are obvious in people with rheumatoid arthritis (RA) (10), spondyloarthritis (SpA) (11) and hip/knee osteoarthritis (HOA/KOA) (12), and exercise is the cornerstone of the European League against Rheumatism (EULAR) recommendations for the non-pharmacological disease management of RA (13), axial SpA (14) and HOA/KOA (15). No specific recommendations on PA were available yet; therefore, in 2016, the EULAR project 'recommendations for PA in people with inflammatory arthritis and osteoarthritis' (16) was

started. The task force, developing the recommendations, agreed that RA and SpA as predominant inflammatory arthritis (iA) conditions, and HOA/KOA as most relevant for PA recommendations represented the field of iA and OA respectively. The resulting '2018 recommendations for PA' provide four overarching principles and 10 recommendations which suggest PA as an integral part of standard care throughout the course of disease in people with RA, SpA and HOA/KOA, given its effectiveness, safety and feasibility.

The recommendations for PA in people with RA, SpA and HOA/KOA may constitute a bridge between arthritis-specific exercises and public health recommendations for PA. They are also important because clinical practice shows that patients and clinicians may still believe that PA and exercising according to public health recommendations (3, 4) is potentially harmful and accelerate disease progression. Furthermore, the best strategy for the promotion of daily PA is still a matter of debate and a challenge for health professionals. However, there is evidence for the effectiveness of behaviour change techniques (BCTs) in PA promotion (17, 18).

The EULAR Standardized Operating Procedures (SOPs) for the development of recommendations (19) demand a systematic review (SR) of the literature with qualitative and if possible quantitative analysis, that is, a meta-analysis (MA), to inform the development of recommendations where appropriate. The aim of this paper is to report in detail on the MA that was performed to answer the question on the effectiveness of PA interventions in people with RA, SpA, HOA and KOA as basis for the EULAR PA recommendations.

This SR included only studies that met the dose for effective PA according to the public health recommendations for PA as provided by the ACSM for at least one dimension of exercise.

# Methods

# Design

This study was conducted following the Cochrane Guidelines (20) as requested by the EULAR SOP (19), and the Preferred Reporting Items for Systematic Reviews and Meta-Analysis (PRISMA) analysis grid. The study was registered in PROSPERO (CRD42017082131; https://www.crd.york.ac.uk/ PROSPERO/).

# **Selection Criteria**

Included were randomised controlled trials (RCTs) on patients with RA, SpA and HOA or KOA, where the intervention of interest met the dose of the public health recommendations for effective PA as provided by ACSM for at least one dimension of exercise compared with any intervention not meeting the ACSM recommendations, such as low-dosed exercise, home exercises, attention control, usual care and waiting list. Further, for the inclusion in the SR, studies needed to investigate

- the effect of cardiovascular exercises on cardiovascular fitness, or
- the effect of muscle strength exercise on lower limb muscle strength, or
- the effect of flexibility exercises on flexibility, or
- the effect or neuromotor exercise on neuromotor performance, or
- the effect of a physical activity promotion strategy on the amount of daily PA in people with RA, SpA, HOA or KOA.

The MA involved those studies with the most frequently used proxy for cardiovascular fitness (eg, VO2 max, watts both assessed on a treadmill/bicycle ergometer), muscle strength (of the lower limb like M.quadriceps femoris), flexibility (eg, passive/active range of motion (ROM), Bath Ankylosing Spondylitis Metrology Index) and PA (eg, questionnaire, accelerometer). The time point of interest was the first assessment after the intervention period.

# Search strategy

Given the presumably high number of available literature, we chose a pragmatic approach: We first searched for key MA or SR including RCTs and, second updated the yield with recently published RCTs that investigated the effectiveness of PA interventions in adults with RA/SpA/HOA/KOA according to the inclusion criteria described before. The SR was performed in PubMed/Medline, Cochrane Library, Embase, Web of Science, Emcare, and PsycInfo, using both MeSH terms and freetext, covering the time frame until April 2017.

No language restrictions were applied. No additional search for grey literature or ongoing studies or unpublished data was conducted. Reference lists of included studies were scanned, and experts in the field of RA (EH), SpA (HD), HOA/KOA (CBJ), and behaviour change (KK) were consulted identifying relevant missing studies. The search strategy for Pubmed/Medline is provided in online supplementary file 1. A-KRO and KN screened firstly the abstracts of all identified records and secondly assessed full-text papers fulfilling the inclusion criteria.

#### **Risk of bias Assessment**

The risk of bias assessment included the methodological domains: random sequence generation (selection bias), allocation concealment (selection bias), blinding of participants (performance bias), blinding of outcome assessment (detection bias), incomplete data (attrition bias), selective bias (reporting bias,) and other potential bias. These criteria were judged as 'high risk', 'low risk' or 'unclear risk' as described by the Cochrane Handbook for Systematic Reviews of Interventions (20). Due to the vast number of studies included, we used a pragmatic approach. If a previous SR or MA had coded included studies for risk of bias, then these ratings were used here, after being spot-checked by the first author (A-KRO). When no previous review or meta-analyses had assessed risk of bias with the Cochrane risk of bias assessment tool, risk of bias rating was performed by two independent assessors (A-KRO, CH). If agreement was not achieved, a third review author (KN) made a final decision. Publication bias was investigated as small study bias and assessed by visual inspection of the funnel plot asymmetry.

#### Data extraction and management

The data extraction was performed by one reviewer (A-KRO) and verified by double reading (CBJ). In studies with more than one intervention group, the intervention groups were pooled if there were only small differences in the intervention (eg, walking or swimming as aerobic exercise intervention). Some authors were contacted to obtain unreported data.

#### Synthesis of data

Applying the PRISMA procedure, this MA focused on the quantitative analysis. The effects of PA on cardiovascular fitness, muscle strength and flexibility were calculated as the standardised mean difference (SMD), allowing for pooling the results of the various outcomes assessed in individual trials. The SMD was estimated as the difference between the postintervention scores in the intervention and control groups divided by the pooled SD. The SD was extracted or estimated from the SE, the 95% Cl, the p value, IQR or other methods recommended by the Cochrane Collaboration (20). The SMD was clinically interpreted as originally proposed by Cohen (21). An SMD of 0.2 was considered small, an SMD of 0.5 was considered moderate (and would be recognised as clinically important) and an SMD equal to 0.8 was considered large (20). This estimate of the effect size is slightly biased, overestimating the effect size of small studies, and a correction factor was applied to convert the effect size to Hedges's g. Meta-analyses using a random-

effects model were applied. Heterogeneity was tested using the Q-test and was further quantified as the I<sup>2</sup> statistic (22) measuring the proportion of variation (ie, inconsistency) in the combined estimates due to between study variance (23). An I<sup>2</sup> value of <40% means low inconsistency between the results of individual trials, and an I<sup>2</sup> value of 100% means maximal inconsistency. All statistical analyses were performed in <stata V.15 (Stata Corp, College Station, Texas, USA).

# Results

#### Study selection and study characteristics

The literature search revealed 3460 records; 237 of these were screened in full text (figure 1). A total of 174 studies were excluded due to unclear or insufficient dose (n=117), inappropriate study design (n=11), therapeutic exercise intervention (n= 9), no between-group difference reported (n=12), joint replacement during intervention (n=5) and no appropriate outcome (n=20). The SR comprised 63 RCTs (11, 24-87) (two RCTs were reported by two papers each (33, 49, 78, 82)) published between 1985 and 2017, of which 49 RCTs were included in the MA. The sample size of these studies ranged from 16 to 222 participants; overall, a total of 3909 participants (mean age (SD) 56.2 (9.6) years) were included in the MA (1049 people with RA, 508 people with SpA, 1855 people with KOA and 497 people with HOA). The intervention. However, further assessment time points were heterogeneous; six studies (13%) performed one or several follow-ups, up to 29 months (37, 52, 54, 66, 81, 83). No detrimental effects of PA were reported in any study. Study characteristics of all studies included in the SR are provided in online supplementary table 1; interventions with assessments and outcomes are described in online supplementary tables 2-5.



Figure 1. Preferred Reporting Items for Systematic Reviews and Meta-Analysis flow chart describing the detailed selection process.

#### Risk of bias within studies

Online supplementary figures 6-10, describe the Cochrane risk of bias assessment. Ratings of 32 studies were adopted from existing SRs/MAs and spot-checked by the first author (12, 18, 88-90). The remaining articles were coded for risk of bias. Two RCTs were rated as low risk of bias for all items (30, 37). Randomization was performed according Cochrane criteria (20) in 44 (67%) RCTs. The allocation concealment was ensured in 31 (47%) RCTs. The blinding of participants was assured in 8 RCTs (12%), the blinding of the outcomes assessment in 30 (46%) RCTs. The risk of incomplete data was low in 49 (75%) RCTs. In 32 (48%) RCTs, a study protocol was available, with corresponding published data, resulting in a low risk for selective reporting.

#### Effects on cardiovascular fitness

Sixteen RCTs investigated the effect of aerobic exercise on cardiovascular fitness (online supplementary table 2), including 1326 participants with a mean (SD) age of 51.4 (7.2) years. The MA showed a moderate effect of SMD 0.56 (95% CI 0.38 to 0.75) (figure 2). The overall heterogeneity of study effects was moderate (I<sup>2</sup>=42.2%); the funnel plot (online supplementary figure 11) showed an asymmetry, indicating a risk of small study bias.

Author (publicationyear)	SMD (95% CI)	Weight	
Rheumatoid Arthritis Harkcom 1985 Baslund 1993 Hansen 1993 Van den Ende 1996 Sanford Smith 1998 Neuberger 2007 Baillet 2009 Hisieh 2009 Westby 2000 Subtotal (I-squared = 52.3%, p = 0.033)	0.48 (-0.53, 1.49) 1.55 (0.46, 2.63) 0.14 (-0.47, 0.74) 0.92 (0.34, 1.51) -0.31 (-1.20, 0.57) 0.25 (-0.03, 0.53) 0.38 (-0.19, 0.95) 0.78 (0.04, 1.53) 1.42 (0.44, 2.40) 0.54 (0.22, 0.86)	2.87 2.54 6.15 6.46 3.55 12.65 6.65 4.63 3.03 48.54	
Osteoarthritis and Rheumatoid Arthritis Minor 1989 Subtotal (I-squared = .%, p = .)	0.98 (0.52, 1.44) 0.98 (0.52, 1.44)	8.46 8.46	
Osteoarthritis Ettinger 1997 Subtotal (I-squared = .%, p = .)	0.36 (0.11, 0.61) 0.36 (0.11, 0.61)	13.23 13.23	
Ankylosing Spondylitis Karapolat 2009 Niedermann 2013 Hsieh 2014 Sveaas 2014 Jennings 2015 Subtotal (I-squared = 0.0%, p = 0.628)	0.53 (-0.17, 1.23) 0.73 (0.32, 1.13) 1.12 (0.14, 2.11) 0.73 (-0.11, 1.57) 0.36 (-0.12, 0.83) 0.62 (0.36, 0.87)	5.08 9.56 3.00 3.86 8.27 29.77	
Overall (I-squared = 42.3%, p = 0.038)	0.56 (0.38, 0.75)	100.00	
NOTE: Weights are from random effects analysis			
-2 -1 0 1 2			
Eavours control Eavours cardiovascular exercise			

Figure 2. Forest plot showing the results of the meta-analysis on the effect of aerobic exercise on measured on VO2max in people with rheumatoid arthritis, spondyloarthritis and hip/knee osteoarthritis. SMD, standardised mean difference.

#### Effects on muscle strength

Thirty-one RCTs investigated the effect of muscle strength exercises on lower limb muscle strength in people with RA or HOA/KOA (online supplementary table 3), of which 28 provided data on M.quadriceps femoris strength. The measurement (60°/sec, 90°/sec) and units (Nm, kp, kg, N, S/kg) varied between the studies; therefore, analyses were performed on SMD. Three RCTs were not included as they reported data in median and range(43, 75, 91). Although these data could be recalculated into mean and SD, this may lead to imprecise data given the small sample sizes in these studies. If strength was described for both legs, values for the affected leg/joint were included in the analysis. If strength was measured in different positions, the velocity of 60°/sec was preferred for the MA. Twenty-five RCTs, including 936 participants with a mean (SD) age of 61.5 (5.0) years, were included in the MA, which showed a moderate effect of muscle strength exercise on M.quadriceps femoris strength with a SMD 0.54 (95% CI 0.35 to 0.72; figure 3) with substantial heterogeneity (I<sup>2</sup>=67%). The funnel plot shows minor asymmetry for both RA and HOA/KOA (online supplementary figure 12), thus the analyses might slightly overestimate the effect for both RA and HOA/KOA.

Author (publicationyear)	SMD (95% CI)	Weight
Osteoarthritis		
Börjesson 1996	0.28 (-0.23, 0.78)	4.28
Baker 2001	0.37 (-0.27, 1.01)	3.57
Mikesky 2006	-0.05 (-0.37, 0.26)	5.31
Wang 2007	0.73 (0.07, 1.39)	3.49
Jan 2008	0.69 (0.19, 1.20)	4.27
Lin 2009	0.96 (0.47, 1.45)	4.36
Weng 2009	0.56 (0.06, 1.06)	4.30
Bennell 2010	0.39 (-0.06, 0.84)	4.55
Lim 2010	-0.41 (-0.94, 0.12)	4.14
Salli 2010	1.28 (0.74, 1.81)	4.11
Foroughi 2011	0.79 (0.18, 1.40)	3.72
Juhakoski 2011	0.78 (0.40, 1.16)	4.97
Bruce-Brand 2012	-0.51 (-1.54, 0.52)	2.12
Sayers 2012	0.80 (0.05, 1.55)	3.08
Anwer 2014	<ul> <li>1.43 (0.74, 2.11)</li> </ul>	3.37
Jorge 2015	1.16 (0.61, 1.71)	4.04
Hermann 2016	0.71 (0.25, 1.16)	4.56
Svege 2016	0.08 (-0.31, 0.46)	4.94
Subtotal (I-squared = 72.6%, p = 0.000)	0.56 (0.33, 0.79)	73.21
Rheumatoid arthritis		
Van den Ende 1996	0.20 (-0.25, 0.66)	4 55
Komatireddy 1997	0.07 (-0.56, 0.71)	3 50
Häkkinen 2004	1 13 (0 59 1 66)	4 10
De Jong 2009	0.59 (-0.06, 1.24)	3 53
Lemmey 2009	0.59 (-0.17, 1.36)	3.04
Strasser 2011	0.65 (-0.04 1.34)	3 36
Sigueria 2017	0.21 (-0.23, 0.65)	4 62
Subtotal (Lsquared = 41.8% n = 0.112)	0.47 (0.19, 0.76)	26.79
	0.47 (0.10, 0.10)	20.10
Overall (I-squared = 67.0%, p = 0.000)	0.54 (0.35, 0.72)	100.00
NOTE: Weights are from random effects analysis		
-2 -1 0 1 2	2	

Figure 3. Forest plot showing the results of the meta-analysis on the effect of strength exercise on M.quadriceps femoris in people with rheumatoid arthritis and hip/knee osteoarthritis. SMD, standardised mean difference.

#### Effects on flexibility

Eight RCTs investigated the effect of flexibility exercises (ROM, stretching) combined with strength or aerobic exercises compared with no exercise or usual care, and measured the effect on ROM (passive or active) or the Bath Ankylosing Spondylitis Mobility Index. One RCT was not included in the MA as data was presented as median and range (71). None of the RCTs compared flexibility exercises alone with no exercises (online supplementary table 4). The MA of the seven RCTs, including 477 participants with a mean (SD) age of 49.3 (15.2) years, showed no effect of the exercise combination on flexibility either for RA or HOA/KOA showing an overall SMD 0.12 (95%CI, -0.16 to 0.41) (figure 4). The overall heterogeneity of study effects was moderate (I<sup>2</sup>=31.1%). Publication bias was not evaluated, due to the sparse number of included studies, especially when stratifying on patient group (HOA/KOA and SpA) in the MA (20).



Figure 4. Forest plot showing the results of the meta-analysis on the effect of flexibility exercises combined with strength or aerobic exercises on flexibility in people with ankylosing spondylitis and hip/knee osteoarthritis. Flexibility was measured as active or passive range of motion, and using the Bath Ankylosing Spondylitis Mobility Index (BASMI). SMD, standardised mean difference.

#### Effects on neuromotor performance

Studies evaluating neuromotor performance were scarce. Da Silva and colleagues (36) showed that a 16-weeks neuromotor exercise programme had a statistically positive effect in neuromotor performance in 91 people with RA, evaluated with Berg Balance Scale, Tinettis Test and Timed-Up-and-Go Test. Four RCTs in people with KOA who performed

strength exercises, evaluated neuromotor performance with the step test for standing balance, reposition error of the knee, Berg Balance Scale and one leg standing time with eyes open. Two RCTs showed a positive effect (63, 92), two found no effect (74, 93). Røgind and colleagues (71) investigated the effect of a programme combining strength, stretching, mobility and balance exercises on postural sway, but found no effects.

#### Effects on daily PA

Eleven RCTs investigated the effect of a daily PA promotion intervention on daily PA (online supplementary table 5) evaluated by accelerometer or questionnaire. In six RCTs, counselling based on BCTs was an essential component of the intervention (30, 31, 34, 58, 61, 69). Three RCTs did not include a counselling strategy as part of the intervention (46, 56, 70) and two studies described data with percentages (33, 49, 78, 82) and were therefore excluded from the analysis. The MA of the six studies, including 694 participants with a mean (SD) age of 59.5 (9.5) years, showed a small effect SMD 0.21 (95%CI 0.03 to 0.38) (figure 5) with no heterogeneity (0%) and no small study bias (online supplementary figures 13).



Figure 5. Forest plot showing the results of the meta-analysis on the effect of physical activity promotion strategies including counselling. SMD, standardised mean difference.

# Discussion

The four MAs overall included 49 studies with 1049 people with RA, 508 people with SpA, 497 people with HOA and 1855 people with KOA. Moderate short-term effects of aerobic and muscle strength exercises on cardiovascular fitness and muscle strength in people with RA, SpA and HOA/KOA were shown. No studies on flexibility exercises alone were found, whereas flexibility exercises combined with muscle strength or aerobic exercise had no effect on flexibility. PA had no detrimental side effects like increased disease activity in any exercise dimension. PA promotion with counselling based on BCTs showed a small beneficial effect on daily PA which has been shown to be a strong mediator for further health benefits. Overall, this MA confirms the increasing evidence for the multiple benefits of exercise in people with RA, SpA and HOA/KOA, and indicates that PA promotion interventions can increase daily PA within these individuals.

In this MA, we reported the effectiveness estimates for each PA dimension over the three conditions based on the effectiveness data available in the primary studies. Exercise interventions are effective if they achieve a physiological response, such as increased flexibility, muscular strength and cardiovascular fitness, which is determined by the right intervention in its right dose, as well as patient adherence (94). Only seven RCTs, included in the MA (13%), reported (minor) adverse events related to exercise like transient increased pain (30, 38, 65, 77, 81, 85, 92). Our findings regarding the safety of exercise and PA promotion interventions are in line with the literature (3). Even a small to moderate beneficial effect of exercise on joint damage, inflammation and symptoms (pain, stiffness, joint tenderness) was found (89). Bischoff and Roos (95) showed that aerobic and strength exercise was equally effective regarding pain and function in people with HOA/KOA. Indeed, the effectiveness of interventions cannot be appraised without taking account to feasibility. The feasibility of interventions can be evaluated by adherence to the intervention or the study protocol (reported as attendance rate). However, (self-)reported adherence to intervention might be overestimated due to recall bias or social desirability (96). For aerobic exercise interventions, 75% of the MA-included studies reported an attendance rate varying between 48 and 100%; for muscle strength exercises, 60% of the MA-included studies reported an attendance rate between 49 and 96%. Attendance rate tends to decline over time (38, 42), the compliance between those with supervised and home-exercises seems to vary (47, 60). This information underlines that a low compliance rate is one of the most compelling challenges for exercise and PA interventions. Long-term adherence implies a change in lifestyle, which can be supported by interventions based on BCTs (18, 97).

This MA was based on studies that followed the public health recommendation for PA according to ACSM (3) and thus achieved the minimal dose for effective PA interventions. The ACSM (98) provide clear recommendations regarding frequency and type of exercise in people with arthritis. Cardiovascular training was performed with no-impact sports like (Nordic) walking

(68), dance (86), cycling (76) or water-based (48) exercising. The recommended intensity varies from moderate to vigorous (3), but also high intensity training up to 90% of the predicted maximal heart rate or 80% one-repetition-movement appears to be effective and feasible in people with RA (60, 83, 99). Flexibility exercises were performed as usual care or combined part of the intervention. Most included studies investigated (semi-) supervised exercise interventions. 'Booster strategies' like phone calls (33, 34, 47, 82), devices (eg, pedometer (34, 46), wearable (56, 61)), home visits (27, 38), log book (34, 47, 86, 100), web-based instructions (49), written material (48, 70, 86), visual instructions (eq, video (70)) were reported. The interventions were always provided by physiotherapists, except in three studies (pilates trainer (24), physician (76), physiatrist (80)). Only if adherence to the FITT-VP components of the intervention (frequency, intensity, time, type, volume, and progression of exercise) is given and reported the best possible training effect can be achieved and the individual goal on physical performance level can be reached (101). The initiation and maintenance of PA are both challenging and meet different general and disease-specific barriers and facilitators (102-105). This highlights the need for further development in promising strategies, such as BCTs for the promotion of a long-term active lifestyle in people with RA, SpA and HOA/KOA. However, starting and maintaining regular exercise in all four dimensions is difficult, even for healthy people. Therefore, it is important to pinpoint that the public health recommendations for PA (3) state that exercise below the recommended levels may still be beneficial for people not able or not willing to fulfil the recommended amount of PA.

#### Strengths and Limitations

To our knowledge, this is the first MA gathering the current evidence on the effectiveness of exercising according to the ACSM principles (3) with regard to the dimensions and dose for exercise in the three rheumatic conditions RA, SpA and HOA/KOA. The aim of this MA was to evaluate the overall effect of exercise and PA promotion, and we do not have evidence to suspect heterogeneity in treatment response among these conditions. However, a disease-specific analysis could be beneficial.

The vast variability of used assessments and scales led us to indirect comparisons. Our analysis illustrates that a consensus on assessments in the field of PA could help to launch more research with comparative outcomes and findings. Gates and colleagues (106) described a consensus how to facilitate the comparison of data between heterogeneous studies in research on PA in HOA/KOA. One of the important points was to decide on MET minutes per week as a key method to assess daily PA and defined minimal requirements for future studies. Being aware of the indirect comparisons, the analyses were a good basis for the task force to decide on PA recommendations for people with RA, SpA and HOA/KOA. More studies with a larger number of participants and similar assessments are needed to clearly decide on the long-term effects of exercise, as the overall aim of PA promotion is a change in long-term lifestyle. Long-term supervised exercises seem to be more effective than short-term exercises (88, 107); however, the intervention duration and setting was not part of this analysis.

It is a limitation that only one reviewer screened the abstracts and decided on unclear abstracts together with a second reviewer, which is not fully in line with standard procedures of a SR (108). However, we applied a double check by experts to ensure that no relevant studies were missed.

# Conclusion/clinical implication

Physical activity according to public health recommendations following the ACSM principles (3) show moderate effectiveness on cardiorespiratory and muscle strength as well as PA promotion interventions based on counselling in people with with RA, SpA and HOA/KOA. No evidence was found for the effects of flexibility exercises and almost no literature was available evaluating neuromotor exercises. Many studies had to be excluded due to poor reporting on the intervention. Future studies should use the TIDieR-checklist (109) (Template for Intervention Description and Replication) or CERT (110) (Consensus on Exercise Reporting Template) to improve completeness of reporting. This will enable clinicians to implement effective interventions and researchers to replicate or merge findings.

# Funding

The manuscript is part of the EULAR project "2018 EULAR recommendations for physical activity in people with inflammatory Arthritis and Osteoarthritis".

# Acknowledgements

We thank Mrs. José Plevier, Walaeus Library, Leiden University Medical Center, Leiden, The Netherlands, for supporting our literature search, and Christian Horvath (CH), MSc student at ZHAW for his help in the Cochrane risk of bias assessment.

Many thanks to all Task Force Members of the EULAR endorsed project "2018 EULAR recommendations for physical activity in people with inflammatory arthritis and osteoarthritis", who all participated actively in the development of the recommendations and supported the idea of a MA.

## Contributions

A-KRO performed the SR and data extraction, supervised by KN and TPVV. JS developed the search strategy for all databases. CBJ double-checked data and performed the metaanalysis. A-KRO and KN drafted the manuscript. All authors of this manuscript were members of the EULAR Task Force, except JS, who developed the recommendations for physical activity in people with inflammatory arthritis and osteoarthritis. They revised this manuscript and approved the final version for publication.

## **Competing interests**

None

# Additional Files:

- Supplement 1: Search strategy
- Supplement 2: Table study characteristics
- Supplement 3: Risk of bias assessment (one document providing legends, Figs 6-10 separately)
- Supplement 4: Tables 2-5 with description of interventions
- Supplement 5: Funnel plots (Figures 11-13)
- Supplement 6: PRISMA checklist

# References

- 1. Caspersen CJ, Powell KE, Christenson GM. Physical activity, exercise, and physical fitness: definitions and distinctions for health-related research. Public Health Rep. 1985;100(2):126-31.
- Keogh JWL, Grigg J, Vertullo CJ. Is Home-Based, High-Intensity Interval Training Cycling Feasible and Safe for Patients With Knee Osteoarthritis?: Study Protocol for a Randomized Pilot Study. Orthopaedic Journal of Sports Medicine. 2017;5(3).
- Garber CE, Blissmer B, Deschenes MR, Franklin BA, Lamonte MJ, Lee IM, et al. American College of Sports Medicine position stand. Quantity and quality of exercise for developing and maintaining cardiorespiratory, musculoskeletal, and neuromotor fitness in apparently healthy adults: guidance for prescribing exercise. Medicine and science in sports and exercise. 2011;43(7):1334-59.
- 4. WHO. Global recommendations on physical activity for health. Geneve 2010.
- Jette M, Sidney K, Blumchen G. Metabolic equivalents (METS) in exercise testing, exercise prescription, and evaluation of functional capacity. Clin Cardiol. 1990;13(8):555-65.
- Swain DP, Franklin BA. Comparison of cardioprotective benefits of vigorous versus moderate intensity aerobic exercise. Am J Cardiol. 2006;97(1):141-7.
- Lee IM, Paffenbarger RS, Jr. Associations of light, moderate, and vigorous intensity physical activity with longevity. The Harvard Alumni Health Study. Am J Epidemiol. 2000;151(3):293-9.
- Lee IM, Shiroma EJ, Lobelo F, Puska P, Blair SN, Katzmarzyk PT, et al. Effect of physical inactivity on major noncommunicable diseases worldwide: an analysis of burden of disease and life expectancy. Lancet. 2012;380(9838):219-29.
- Warburton DE, Bredin SS. Reflections on Physical Activity and Health: What Should We Recommend? Can J Cardiol. 2016;32(4):495-504.
- Cooney JK, Law RJ, Matschke V, Lemmey AB, Moore JP, Ahmad Y, et al. Benefits of exercise in rheumatoid arthritis. J Aging Res. 2011;2011:681640.
- Sveaas SHBIJ, Provan SA. Semb AG, Hagen KB, Vollestad N, Fongen C, Olsen IC, Michelsen A, Ueland T, Aukrust P, Kvien TK, Dagfinrud H. Efficacy of high intensity exercise on disease activity and cardiovascular risk in active axial spondyloarthritis: a randomized controlled pilot study. PloS one. 2014;9(9):e108688.
- Bartholdy C, Juhl C, Christensen R, Lund H, Zhang W, Henriksen M. The role of muscle strengthening in exercise therapy for knee osteoarthritis: A systematic review and meta-regression analysis of randomized trials. Seminars in arthritis and rheumatism. 2017.
- 13. Combe B, Landewe R, Daien CI, Hua C, Aletaha D, Alvaro-Gracia JM, et al. 2016 update of the EULAR recommendations for the management of early arthritis. Annals of the rheumatic diseases. 2017;76(6):948-59.
- van der Heijde D, Ramiro S, Landewe R, Baraliakos X, Van den Bosch F, Sepriano A, et al. 2016 update of the ASAS-EULAR management recommendations for axial spondyloarthritis. Annals of the rheumatic diseases. 2017;76(6):978-91.
- Fernandes L, Hagen KB, Bijlsma JW, Andreassen O, Christensen P, Conaghan PG, et al. EULAR recommendations for the non-pharmacological core management of hip and knee osteoarthritis. Annals of the rheumatic diseases. 2013;72(7):1125-35.
- Rausch Osthoff AK, Vliet Vlieland TP, Braun J, Adams J, Brodin N, Dagfinrud H, et al. 2018 EULAR recommendations for physical activity in people with inflammatory arthritis and osteoarthritis. Annals of Rheumatic Diseases 2018; 77 (9):1251-1260
- 17. Larkin L, Kennedy N, Gallagher S. Promoting physical activity in rheumatoid arthritis: a narrative review of behaviour change theories. Disability and rehabilitation. 2015;37(25):2359-66.
- Demmelmaier I, Iversen MD. How Are Behavioral Theories Used in Interventions to Promote Physical Activity in Rheumatoid Arthritis? A systematic review. Arthritis care & research 2018;70(2):185-196
- van der Heijde D, Aletaha D, Carmona L, Edwards CJ, Kvien TK, Kouloumas M, et al. 2014 Update of the EULAR standardised operating procedures for EULAR-endorsed recommendations. Annals of the rheumatic diseases. 2015;74(1):8-13.
- 20. Higgins JG, S. Cochrane Handbook for Systematic Reviews of Interventions. Version 5.1.0. Collaboration TC, editor2011.
- 21. Cohen J. Statistical Power Analysis for the Behavioral Sciences. 2nd ed. Hillsdale, NJ: Erlbaum; 1988.
- 22. Higgins JTTSG. Quatifying heterogeneity in a meta-analysis. Statistics in Medicine. 2002;21:1539-58.
- Higgins JP, Thomson SG, Deeks JJ, Altman DG. Measuring inconsistancy in meta-analyses. BMJ 2003;327(7414):557-60.
- 24. Altan LKN. Dizdar M,Yurtkuran M. Effect of Pilates training on people with ankylosing spondylitis. Rheumatology international. 2012;32(7):2093-9.
- Anwer SAA. Effect of isometric quadriceps exercise on muscle strength, pain, and function in patients with knee osteoarthritis: A randomized controlled study. Journal of Physical Therapy Science. 2014;26(5):745-8.
- Baillet A, Payraud E, Niderprim VA, Nissen MJ, Allenet B, Francois P, et al. A dynamic exercise programme to improve patients' disability in rheumatoid arthritis: a prospective randomized controlled trial. Rheumatology. 2009;48(4):410-5.
- 27. Baker KRN. FelsonDT, Layne JE, Sarno R, Roubenoff R. The efficacy of home based progressive strength training in

older adults with knee osteoarthritis: A randomized controlled trial. Journal of Rheumatology. 2001;28(7):1655-65.

- Baslund BLK, Andersen V, Halkjaer Kristensen J, Hansen M, Klokker M, Pedersen BK. Effect of 8 wk of bicycle training on the immune system of patients with rheumatoid arthritis. Journal of applied physiology. 1993;75(4):1691-5.
- Bennell KLET, Pua YH, Abbott JH, Sims K, Metcalf B, McManus F, Wrigley TV, Forbes A, Harris A, Buchbinder R. Efficacy of a multimodal physiotherapy treatment program for hip osteoarthritis: A randomised placebo-controlled trial protocol. BMC musculoskeletal disorders. 2010;11 (238).
- Bennell KLCPK, Egerton T, Metcalf B, Kasza J, Forbes A, Bills C, Gale J, Harris A, Kolt GS, Bunker SJ, Hunter DJ, Brand CA, Hinman RS. Telephone Coaching to Enhance a Home-Based Physical Activity Program for Knee Osteoarthritis: A Randomized Clinical Trial. Arthritis Care and Research. 2017;69(1):84-94.
- Bieler TSV, Magnusson SP, Kjaer M, Beyer N. Even in the long run nordic walking is superior to strength training and home based exercise for improving physical function in older people with hip osteoarthritis-an RCT. Annals of the rheumatic diseases. 2016;75:55.
- Borjesson MRE, Weidenhielm L, Mattsson E, Olsson E. Physiotherapy in knee osteoarthrosis: effect on pain and walking. Physiotherapy research international : the journal for researchers and clinicians in physical therapy. 1996;1(2):89-97.
- Brodin NEE, Jensen I, Nisell R, Opava, C. Coaching patients with early rheumatoid arthritis to healthy physical activity: a multicenter, randomized, controlled study. Arthritis and rheumatism. 2008; 59(3):325-31.
- 34. Brosseau LWGA, Kenny GP, Reid R, Maetzel A, Tugwell P, Huijbregts M, McCullough C, De Angelis G, Chen L. The implementation of a community-based aerobic walking program for mild to moderate knee osteoarthritis: a knowledge translation randomized controlled trial: part II: clinical outcomes. BMC public health. 2012;12:1073.
- Bruce-Brand RAWRJ, Ong JC, Emerson BS, O'Byrne JM, Moyna NM. Effects of home-based resistance training and neuromuscular electrical stimulation in knee osteoarthritis: a randomized controlled trial. BMC musculoskeletal disorders. 2012;13:118.
- 36. da Silva KN, Teixeira LE, Imoto AM, Atallah AN, Peccin MS, Trevisani VF. Effectiveness of sensorimotor training in patients with rheumatoid arthritis: a randomized controlled trial. Rheumatology international. 2013;33(9):2269-75.
- 37. de Jong ZMM, Kroon HM van Schaardenburg D, Dijkmans BA, Hazes JM, Vliet Vlieland TPM. Long-term follow-up of a high-intensity exercise program in patients with rheumatoid arthritis. Clinical rheumatology. 2009;28(6):663-71.
- Ettinger WHBR, Messier SP, Applegate W, Rejeski WJ, Morgan T, Shumaker S, Berry MJ, Otoole M, Monu J, Craven T. A randomized trial comparing aerobic exercise and resistance exercise with a health education program in older adults with knee osteoarthritis - The Fitness Arthritis and Seniors Trial (FAST). JAMA. 1997;277(1):25-31.
- Fang HCW, Pan Y, Wu D, Liang L. Six-month home-based exercise and supervised training in patients with ankylosing spondylitis. International Journal of Clinical and Experimental Medicine. 2016;9(3):6635-41.
- 40. Flint-Wagner HG, Lisse J, Lohman TG, Going SB, Guido T, Cussler E, et al. Assessment of a sixteen-week training program on strength, pain, and function in rheumatoid arthritis patients. Journal of clinical rheumatology : practical reports on rheumatic & musculoskeletal diseases. 2009;15(4):165-71.
- Foroughi NSR, Lange AK, Baker MK, Fiatarone Singh MA, Vanwanseeele B. Lower limb muscle strengthening does not change frontal plane moments in women with knee osteoarthritis: a randomized controllled trial. Clinical biomechanics. 2011;26:167-74.
- 42. Hakkinen A. Effectiveness and safety of strength training in rheumatoid arthritis. Current Opinion in Rheumatology. 2004;16(2):132-7.
- Hansen TM, Hansen G, Langgaard AM, Rasmussen JO. Longterm physical training in rheumatoid arthritis. A randomized trial with different training programs and blinded observers. Scandinavian journal of rheumatology. 1993;22(3):107-12.
- 44. Harkcom TM, Lampman RM, Banwell BF, Castor CW. Therapeutic value of graded aerobic exercise training in rheumatoid arthritis. Arthritis and rheumatism. 1985;28(1):32-9.
- 45. Hermann AH-LA, Zerahn B, Mejdahl S, Overgaard S. Preoperative progressive explosive-type resistance training is feasible and effective in patients with hip osteoarthritis scheduled for total hip arthroplasty--a randomized controlled trial. Osteoarthritis and cartilage. 2016;24(1):91-8.
- 46. Hiyama Y, Yamada M, Kitagawa A, Tei N, Okada S. A four-week walking exercise programme in patients with knee osteoarthritis improves the ability of dual-task performance: a randomized controlled trial. Clinical rehabilitation. 2012;26(5):403-12.
- 47. Hsieh LC, Chuang C, Chai H, Chen W, He Y. Supervised aerobic exercise is more effective than home aerobic exercise in female chinese patients with rheumatoid arthritis. Journal of rehabilitation medicine. 2009; 41(5):332-7
- Hsieh LFCCC, Tseng CS, Wei JC, Hsu WC, Lin YJ. Combined home exercise is more effective than range-of-motion home exercise in patients with ankylosing spondylitis: a randomized controlled trial. BioMed research international. 2014;2014;398190.
- Hurkmans EJ, van den Berg MH, Ronday KH, Peeters AJ, le Cessie S, Vlieland TP. Maintenance of physical activity after Internet-based physical activity interventions in patients with rheumatoid arthritis. Rheumatology. 2010;49(1):167-72.
- Ince GST, Durgun B, Erdogan S. Effects of a multimodal exercise program for people with ankylosing spondylitis. Physical therapy. 2006;86(7):924-35.
- Jan MHLCH, Lin Y, Lin JJ, Lin DH. Effects of weight-bearing versus nonweight-bearing exercise on function, walking speed, and position sense in participants with knee osteoarthritis: a randomized controlled trial. Archives of physical medicine and rehabilitation. 2009;90(6):897-904.
- 52. Jennings FOHA, De Souza MC, Cruz VDG, Natour J. Effects of aerobic training in patients with ankylosing spondylitis.

Journal of Rheumatology. 2015;42(12):2347-53.

- 53. de Jorge MPS, de la Torre-Aboki J, Herrero-Beaumont G. Randomized clinical trials as reflexive-interpretative process in patients with rheumatoid arthritis: a qualitative study. Rheumatology International. 2015;35(8):1423-30.
- Juhakoski RTS, Malmivaara A, Kiviniemi V, Anttonen T, Arokoski JP. A pragmatic randomized controlled study of the effectiveness and cost consequences of exercise therapy in hip osteoarthritis. Clinical rehabilitation. 2011;25(4):370-83.
- Karapolat H, Eyigor S, Zoghi M, Akkoc Y, Kirazli Y, Keser G. Are swimming or aerobic exercise better than conventional exercise in ankylosing spondylitis patients? A randomized controlled study. European journal of physical and rehabilitation medicine. 2009;45(4):449-57.
- Katz P, Margaretten M, Gregorich S, Trupin L. Physical Activity to Reduce Fatigue in Rheumatoid Arthritis: A Randomized, Controlled Trial. Arthritis care & research. 2018; 70:1-10
- 57. Kjeken I, Bo I, Ronningen A, Spada C, Mowinckel P, Hagen KB, et al. A three-week multidisciplinary in-patient rehabilitation programme had positive long-term effects in patients with ankylosing spondylitis: randomized controlled trial. Journal of rehabilitation medicine. 2013;45(3):260-7.
- Knittle KDGV, Hurkmans E, Peeters A, Ronday K, Maes S, Vlieland TPM. Targeting motivation and self-regulation to increase physical activity among patients with rheumatoid arthritis: a randomised controlled trial. Clinical rheumatology. 2015;34(2):231-8.
- 59. Komatireddy GR, Leitch RW, Cella K, Browning G, Minor M. Efficacy of low load resistive muscle training in patients with rheumatoid arthritis functional class II and III. The Journal of rheumatology. 1997;24(8):1531-9.
- Lemmey ABMSM, Chester K, Wilson S, Casanova F, Maddison PJ. Effects of high-intensity resistance training in patients with rheumatoid arthritis: a randomized controlled trial. Arthritis and rheumatism. 2009;61(12):1726-34.
- Li LC, Sayre EC, Xie H, Clayton C, Feehan LM. A Community-Based Physical Activity Counselling Program for People With Knee Osteoarthritis: Feasibility and Preliminary Efficacy of the Track-OA Study. JMIR Mhealth Uhealth. 2017;5(6):e86.
- 62. Lim JYT, E.; Jang, S. N. Effectiveness of aquatic exercise for obese patients with knee osteoarthritis: a randomized controlled trial. PM & R : the journal of injury, function, and rehabilitation. 2010;2(8):723-31.
- Lin DH, Lin CH, Lin YF, Jan MH. Efficacy of 2 non-weight-bearing interventions, proprioception training versus strength training, for patients with knee osteoarthritis: a randomized clinical trial. The Journal of orthopaedic and sports physical therapy. 2009;39(6):450-7.
- Lyngberg KK, Harreby M, Bentzen H, Frost B, Danneskiold-Samsoe B. Elderly rheumatoid arthritis patients on steroid treatment tolerate physical training without an increase in disease activity. Archives of physical medicine and rehabilitation. 1994;75(11):1189-95.
- 65. Mikesky AEMSA, Brandt KD, Perkins SM, Damush T, Lane KA. Effects of strength training on the incidence and progression of knee osteoarthritis. Arthritis Care and Research. 2006;55(5):690-9.
- 66. Minor MA, Hewett JE, Webel RR, Anderson SK, Kay DR. Efficacy of physical conditioning exercise in patients with rheumatoid arthritis and osteoarthritis. Arthritis and rheumatism. 1989;32(11):1396-405.
- Neuberger GBA, Gajewski B, Embretson SE, Cagle PE, Loudon JK, Miller PA. Predictors of exercise and effects of exercise on symptoms, function, aerobic fitness, and disease outcomes of rheumatoid arthritis. Arthritis and rheumatism. 2007;57(6):943-52.
- Niedermann K, Muggli C, Dagfinrud H, Hermann M, Tamborrini G, Ciurea A, Bischoff-Ferrari H. Effect of cardiovascular training on fitness and perceived disease activity in people with ankylosing spondylitis. Arthritis care & research. 2013;65(11):1844-52.
- O'Dwyer T, Monaghan A, Moran J, O'Shea F, Wilson F. Behaviour change intervention increases physical activity, spinal mobility and quality of life in adults with ankylosing spondylitis: a randomised trial. Journal of physiotherapy. 2017;63(1):30-9.
- Rodriguez-Lozano C, Juanola X, Cruz-Martinez J, Pena-Arrebola A, Mulero J, Gratacos J, et al. Outcome of an education and home-based exercise programme for patients with ankylosing spondylitis: a nationwide randomized study. Clinical and experimental rheumatology. 2013;31(5):739-48.
- 71. Rogind HBNB, Jensen B, Moller HC, Frimodt-Moller H, Bliddal H. The effects of a physical training program on patients with osteoarthritis of the knees. Archives of physical medicine and rehabilitation. 1998;79(11):1421-7.
- Salli ASN, Baskent A, Ugurlu H. The effect of two exercise programs on various functional outcome measures in patients with osteoarthritis of the knee: A randomized controlled clinical trial. Isokinetics and Exercise Science. 2010;18(4):201-9.
- 73. Sanford Smith S, MaxcKay-Lyons M, Nunes-Clement S. Therapeutic benefits of aquaerobics for individuals with rheumatoid arthritis. Physiotherapy Canada. 1998;50:40.6.
- 74. Sayers SPGK, Cook CR. Effect of high-speed power training on muscle performance, function, and pain in older adults with knee osteoarthritis: A pilot investigation. Arthritis Care and Research. 2012;64(1):46-53.
- Seneca THEM, Maribo T. Comparable effect of partly supervised and self-administered exercise programme in early rheumatoid arthritis--a randomised, controlled trial. Danish medical journal. 2015;62(8):A5127.
- 76. Shapoorabadi YJVB, Salesi M, Ramezanian H. Effects of aerobic exercise on hematologic indices of women with rheumatoid arthritis: A randomized clinical trial. Journal of research in medical sciences. 2016;21:9.
- Siqueira USOVLG, de Mello MT, Szejnfeld VL, Pinheiro MM. Effectiveness of Aquatic Exercises in Women With Rheumatoid Arthritis: A Randomized, Controlled, 16-Week Intervention-The HydRA Trial. American journal of physical medicine & rehabilitation. 2017;96(3):167-75.
- 78. Sjöquist EBN, Lampa J, Jensen I, Opava C. Physical activity coaching of patients with rheumatoid arthritis in everyday

practice: a long-term follow-up. Musculoskeletal care. 2011; 9(2):75-85.

- Steinhilber BHG, Miller R, Janssen P, Krauss I. Exercise therapy in patients with hip osteoarthritis: Effect on hip muscle strength and safety aspects of exercise-results of a randomized controlled trial. Modern rheumatology. 2016:1-10.
- Strasser BLG, Strehblow C, Schobersberger W, Haber P, Cauza E. The effects of strength and endurance training in patients with rheumatoid arthritis. Clinical rheumatology. 2011;30(5):623-32.
- Svege IL, Nordsletten L, Holm I, Risberg M. A. Long-Term Effect of Exercise Therapy and Patient Education on Impairments and Activity Limitations in People With Hip Osteoarthritis: Secondary Outcome Analysis of a Randomized Clinical Trial. Physical therapy. 2016;96(6):818-27.
- van den Berg MH, Ronday HK, Peeters AJ, Voogt-van der Harst EM, Munneke M, Breedveld FC, et al. Engagement and satisfaction with an Internet-based physical activity intervention in patients with rheumatoid arthritis. Rheumatology. 2007;46(3):545-52.
- van den Ende CHHJM, le Cessie S, Mulder WJ, Belfor DG, Breedveld FC, Dijkmans BA. Comparison of high and low intensity training in well controlled rheumatoid arthritis. Results of a randomised clinical trial. Annals of the rheumatic diseases. 1996;55(11):798-805.
- 84. Wang TJBB, Thompson EF, Whitney JD, Bennett K. Effects of aquatic exercise on flexibility, strength and aerobic fitness in adults with osteoarthritis of the hip or knee. Journal of advanced nursing. 2007;57(2):141-52.
- Weng MCLCL, Chen CH, Hsu JJ, Lee WD, Huang MH, Chen TW. Effects of different stretching techniques on the outcomes of isokinetic exercise in patients with knee osteoarthritis. The Kaohsiung journal of medical sciences. 2009;25(6):306-15.
- Westby MDJP, Rangno KK, Berkowitz J. A randomized controlled trial to evaluate the effectiveness of an exercise program in women with rheumatoid arthritis taking low dose prednisone. The Journal of rheumatology. 2000;27(7):1674-80.
- 87. Hakkinen AHK, Hannonen P. Effects of strength training on neuromuscular function and disease activity in patients with recent-onset inflammatory arthritis. Scandinavian journal of rheumatology. 1994;23(5):237-42.
- Fransen MMS, Harmer AR, Van der Esch M, Simic M, Bennell KL. Exercise for osteoarthritis of the knee: a Cochrane systematic review. British journal of sports medicine. 2015;49(24):1554-7.
- Sveaas SH, Smedslund G, Hagen KB, Dagfinrud H. Effect of cardiorespiratory and strength exercises on disease activity in patients with inflammatory rheumatic diseases: a systematic review and meta-analysis. British journal of sports medicine. 2017;51(14):1065-72.
- Cramp FHS, Almeida C, Kirwan JR, Choy EHS, Chalder T, Pollock J, Christensen R. Non-pharmacological interventions for fatigue in rheumatoid arthritis. Cochrane Database of Systematic Reviews. 2013(8):77.
- Lyngberg KK, Ramsing BU, Nawrocki A, Harreby M, Danneskiold-Samsoe B. Safe and effective isokinetic knee extension training in rheumatoid arthritis. Arthritis and rheumatism. 1994;37(5):623-8.
- Bennell KLHM, Wrigley TV, Hunter DJ, McManus FJ, Hodges PW, Li L, Hinman RS. Hip strengthening reduces symptoms but not knee load in people with medial knee osteoarthritis and varus malalignment: A randomised controlled trial. Osteoarthritis and cartilage. 2010;18(5):621-8.
- Kim HST, Saito K, Kim M, Kojima N, Ishizaki T, Yamashiro Y, Hosoi E, Yoshida H. Effectiveness of exercise with or without thermal therapy for community-dwelling elderly Japanese women with non-specific knee pain: a randomized controlled trial. Archives of gerontology and geriatrics. 2013;57(3):352-9.
- Schiffer T, Knicker A, Hoffman U, Harwig B, Hollmann W, Struder HK. Physiological responses to nordic walking, walking and jogging. European journal of applied physiology. 2006;98(1):56-61.
- Bischoff HAR, E. M. Effectiveness and safety of strengthening, aerobic, and coordination exercises for patients with osteoarthritis. Current Opinion in Rheumatology. 2003;15(2):141-4.
- 96. Coughlin SS. Recall bias in epidemiologic studies. J Clin Epidemiol. 1990;43(1):87-91.
- 97. Larkin LGS, Cramp F, Brand C, Fraser A, Kennedy N. Behaviour change interventions to promote physical activity in rheumatoid arthritis: a systematic review. Rheumatology international. 2015;35(10):1631-40.
- American College of Sports M. ACSM's Guidelines for exercise testing and prescription. tenth edition ed. Philadelphia: Wolters Kluwer; 2018.
- De Jong ZMM, Zwinderman AH, Kroon HM, Jansen A, Ronday KH, Van Schaardenburg D, Dijkmans BAC, Van den Ende CM, Breedveld FC, Vliet Vlieland TPM, Hazes JMW. Is a long-term high-intensity exercise program effective and safe in patients with rheumatoid arthritis? Results of a randomized controlled trial. Arthritis and rheumatism. 2003;48(9):2415-24.
- Jan MHLJJ, Liau JJ, Lin YF, Lin DH. Investigation of clinical effects of high- and low-resistance training for patients with knee osteoarthritis: A randomized controlled trial. Physical therapy. 2008;88(4):427-36.
- 101. Baschung Pfister P, de Bruin ED, Tobler-Ammann BC, Maurer B, Knols RH. The relevance of applying exercise training principles when designing therapeutic interventions for patients with inflammatory myopathies: a systematic review. Rheumatology international. 2015;35(10):1641-54.
- 102. Veldhuijzen van Zanten JJCSR, Peter C, Hale ED, Ntoumanis N, Metsios GS, Duda JL, Kitas G D. Perceived barriers, facilitators and benefits for regular physical activity and exercise in patients with rheumatoid arthritis: A review of the literature. Sports Medicine. 2015;45(10):1401-12.
- O'Dwyer T, McGowan E, O'Shea F, Wilson F. Physical Activity and Exercise: Perspectives of Adults With Ankylosing Spondylitis. Journal of physical activity & health. 2016;13(5):504-13.

- 104. Dobson FBKL, French SD, Nicolson PJ, Klaasman RN, Holden MA, Atkins L., Hinman RS. Barriers and Facilitators to Exercise Participation in People with Hip and/or Knee Osteoarthritis: Synthesis of the Literature Using Behavior Change Theory. American journal of physical medicine & rehabilitation. 2016;95(5):372-89.
- Loeppenthin K, Esbensen B, Ostergaard M, Jennum P, Thomsen T, Midtgaard J. Physical activity maintenance in patients with rheumatoid arthritis: a qualitative study. Clinical rehabilitation. 2014;28(3):289-99.
- Gates LS, Leyland KM, Sheard S, Jackson K, Kelly P, Callahan LF, et al. Physical activity and osteoarthritis: a consensus study to harmonise self-reporting methods of physical activity across international cohorts. Rheumatology international. 2017;37(4):469-78.
- Juhl CC, Roos EM, Zhang W, Lund H. Impact of exercise type and dose on pain and disability in knee osteoarthritis: a systematic review and meta-regression analysis of randomized controlled trials. Arthritis & rheumatology (Hoboken, NJ). 2014;66(3):622-36.
- 108. Higgins JP, Green S. Cochrane Handbook for systematic reviews of interventions. http://handbook.cochrane.org: The Cochrane Collaboration; 2011
- Hoffmann TC, Glasziou PP, Boutron I, Milne R, Perera R, Moher D, et al. Better reporting of interventions: template for intervention description and replication (TIDieR) checklist and guide. BMJ. 2014;348:g1687.
- Slade SC, Dionne CE, Underwood M, Buchbinder R, Beck B, Bennell K, et al. Consensus on Exercise Reporting Template (CERT): Modified Delphi Study. Physical therapy. 2016;96(10):1514-24.

# 4

The perspective of people with axial spondyloarthritis regarding physiotherapy: room for the implementation of a more active approach

Anne-Kathrin Rausch Osthoff, Florus van der Giesen, André Meichtry, Beatrice Walker, Floris A van Gaalen, Yvonne P M Goekoop-Ruiterman, Andreas J Peeters, Karin Niedermann, Theodora P M Vliet Vlieland

> Rheumatology Advances in Practice 2019; 3 (2) doi.org/10.1093/rap/rkz043

Poster Presentations at EULAR congress in London (June 2016) and Swiss Health Professionals in Rheumatology Symposium in Interlaken (September 2016)

# Abstract

### Objectives

Physiotherapy is recommended in the management of people with axial spondyloarthritis (axSpA), with new insights into its preferred content and dosage evolving. The aim of this study was to describe the use and preferences regarding individual and group physiotherapy among people with axSpA.

## Methods

A cross-sectional survey was conducted among people with axSpA living in The Netherlands (NL) and Switzerland (CH).

## Results

Seven hundred and thirteen people with axSpA participated (56.7% male, median age 55 years, median Assessment of Spondyloarthritis International Society Health Index score 4.2). Response rates were 45% (n=206) in NL and 29% in CH (n=507). Of these participants, 83.3% of participants were using or had been using physiotherapy. Individual therapy only was used or had been used by 36.7%, a combination of individual plus land-and water-based group therapy by 29.1% and group therapy by only 5.3%. Fewer than half of the participants attending individual therapy reported active therapy (such as aerobic, muscle strength and flexibility exercises). Although the majority (75.9%) were not aware of the increased cardiovascular risk, participants showed an interest in cardiovascular training, either individually or in a supervised setting. If supervised, a majority, in CH (75.0%) more than in NL (55.7%), preferred supervision by a specialized physiotherapist.

## Conclusions

The majority of people with axSpA use or have used physiotherapy, more often in an individual setting than in a group setting. The content of individual therapy should be more active; in both therapy settings, aerobic exercises should be promoted. In particular, enabling people with axSpA to perform exercises independently would meet their needs and might enhance their daily physical activity.

#### key messages:

- The large majority of people with axSpA uses physiotherapy.
- Individual physiotherapy in people with axSpA consists of mainly passive modalities.
- Many people with axSpA are unaware of increased cardiovascular risk but interested in aerobic exercise.
# Introduction

Axial spondyloarthritis (axSpA) is a chronic, inflammatory rheumatic disease that affects the sacroiliac joints and spine, leading to structural and activity limitations (1). The prevalence in the general population is about 0.1-0.6% according to European disease prevalence data (2, 3). AxSpA affects male and female equally (1:1 ratio) (4). Disease onset usually is in the early adulthood (5), and therefore axSpA has a large impact on working ability and personal and societal costs are high (6, 7).

Drug treatment and physiotherapy, in particular exercise therapy, are the cornerstones of the appropriate management of the disease (1, 8). Especially the fact that people with axSpA have an increased risk of cardiovascular diseases (9), and evidence showing that axSpA affects flexibility (10), balance (11), muscle strength (12), and cardio-respiratory capacity (13) emphasize the need of exercise. In this respect it is important to underline that exercise is a subset of physical activity, and defined as 'planned, structured and repetitive [activity, that] has as a final or intermediate objective, the improvement or maintenance of one or more dimensions of physical activity' (14, 15). Therapeutic exercises are individual and/or disease specific, meant to improve or restore function or to prevent dysfunction.

Regarding exercise, a Cochrane systematic literature review (10) showed that exercise interventions have an effect on spinal mobility and physical function, with the most favourable results being seen with supervised group exercise. None of the 11 included studies in that systematic literature review reported harm as a result of exercising. Based on this evidence, exercise is generally recommended in professional guidelines, with the type (aerobic, muscle strengthening and flexibility), and the preferred mode of delivery [supervised, group exercise therapy (GET)] being defined (1, 16, 17). Recently, EULAR published recommendations on physical activity emphasizing the importance of adequate composition and dosage of activities according to American College of Sports Medicine (ACSM) principles (15) throughout the course of disease (18). Indeed, individual and GET meeting the frequencies, intensity, time, type, volume, progression (FITT-VP) principles described by ACSM (15) was shown to be effective in people with axSpA, by having a positive impact on disease activity, joint damage and cardiovascular risk factors (19-21). In addition, a number of trials investigated the effectiveness of cardiovascular training on disease activity and cardiovascular fitness (20-22). Despite these insights, in research and daily practice exercise may not meet the requirements described in the guidelines. It was found that only a small proportion of GET evaluated in clinical trials met the ACSM recommendations for flexibility, muscle strength or aerobic exercise capacity (23). Moreover, a small survey revealed that physiotherapists providing GET in Switzerland did not include elements of aerobic training in an adequate dose during the training sessions in people with axSpA (K.N., unpublished data). Apart from insufficient delivery, some patients may not exercise at all. The literature on barriers and facilitators to engage in exercise in patients with axSpA is, however, scanty (24).

Internationally, there are currently activities going on to develop an implementation strategy to optimize the usage and delivery of physiotherapy and exercise.

Therefore, we aimed to make an inventory of use, experiences, and preferences of people with axSpA regarding the delivery of individual physiotherapy and GET. As usage, content and preferences regarding physiotherapy may however vary among countries, the inventory was carried out in two countries, The Netherlands (NL) and Switzerland (CH).

# Methods

### Design and setting

This cross-sectional survey was conducted among people with axSpA living in the western region of NL and the German-speaking part of CH. The findings are reported in line with the STrengthening the Reporting of OBservational studies in Epidemiology (STROBE) guidelines (26). The study obtained ethical approval from the Leiden University Hospital Ethical committee (P14.326) and Ethics committee Canton Zurich (KEK-ZH-71-2015).

In both countries, NL and CH, supervised exercise therapy can be offered on an individual or group basis. The latter is usually water and/or land based, offered once a week, supervised by a physiotherapist, and yields an important social factor (26).

The amount of refund for both individual and group therapy differs between the two countries, because health insurance systems are different. In NL, direct access to physiotherapy was introduced in 2006, and most health-care insurers reimburse direct access therapy. However, axSpA GET is currently not reimbursed in NL. In CH, health-care insurers reimburse physiotherapy, including axSpA group exercise, but only if it is induced by a referral. In both countries, but based on different systems, patients have to pay an obligatory financial contribution. In both countries, health-care insurers have expressed the need for a proof of the effectiveness of exercise therapy.

### Participants

### Dutch patients

Four hundred and fifty-eight people with a confirmed diagnosis of axSpA who had visited the rheumatology outpatient clinic in the past 12 months were identified from the registries of three hospitals in The Netherlands: Leiden University Medical Center, Haga Teaching Hospital, The Hague, and Reinier de Graaf Gasthuis, Delft. Eligible patients received an invitation letter from their treating rheumatologist, an information leaflet, a paper survey and a pre-stamped envelope by regular mail. Returned questionnaires were scanned and analysed with the software Cardiff Software (CA, USA). No reminders were sent.

### Swiss patients

All 1742 German-speaking members of the Schweizerische Vereinigung Morbus Bechterew (SVMB) were invited by e-mail to complete an online survey (by use of SurveyMonkey) or a paper version. Representatives of SVMB, a rheumatologist and a researcher signed the invitation. Electronic data were collected with the Internet Protocol (IP) address inactive to preserve anonymity, and all paper questionnaires were collected by the SVMB and forwarded as anonymized versions for data analysis. No reminders were sent.

### Assessments

### Survey on exercise use and preferences

The survey was self-developed in Dutch by a team of researchers and, at a later stage, translated into German. The survey consisted of dichotomous- or multiple-choice questions, multiple-answer options (MAOs) and some with a free text field ('other'-option). The survey consisted of the following parts:

- Demographic and clinical information: age, sex, disease duration (in years) and use of medication (pain medication, NSAIDs, DMARDs, biologicals or no drugs; MAO).
- Use of Individual physiotherapy: usage (if ever/currently; if, frequency, duration, and way of referral) and contents of physiotherapy (active and passive exercises, home exercises, hydrotherapy, education, massage, thermotherapy, kinesiotaping, electrotherapy, US, dry needling, relaxation techniques either individual or group setting; by MAO). In addition, if patients had used physiotherapy but stopped, the reasons for stopping were queried (too hard, more complaints, motivation, no positive effect, too time consuming or no refund; by MAO); Unfortunately, in the online survey for the Swiss population, the option describing the content of the individual therapy as 'I perform exercises meant to strengthen my muscles by using my own weight or free weights or machines' vanished owing to a technical problem, which led to a bias (is this case, data collection is based on the free text field option).
- Use of group physiotherapy: usage of land-based or water-based GET (ever/ currently/no; frequency and duration) and, if patients had stopped it, the reasons were queried (too hard, more discomfort, motivation, no positive effect, too time consuming or no refund; by MAO).
- Patients' motivation and preferences regarding exercises: willingness/ability to exercising individually, knowledge of how to exercise without supervision, way of interaction with supervisior [e.g., (in)direct, via technology, group], preferred frequency and duration of organized activity (by MOA).

### Health status

In addition, the Assessment of Spondyloarthritis International Society Health Index (ASAS HI) was included. The self-reported questionnaire evaluates 17 aspects of function and health and 9 environmental factors in patients with SpA, providing a score on the individuals' health status (27, 28). The lower the score, the better the 'functioning' (29).

### Data analysis

Demographic and disease-specific data were presented as the mean and S.D. or median and associated range for continuous data or as frequencies (percentages) for categorical variables. To compare the characteristics of Dutch and Swiss patients, Student's unpaired t-tests or Mann-Whitney U-tests were used where appropriate for continuous data, and Chi Square or Fischer's exact tests for categorical data. In addition, logistic regression models with nationality as an independent variable were fitted to the data, adjusting for the effect of age, sex, disease duration, DMARD use and current health status. For some of these analyses, some levels of the dependent variable were grouped: current or past treatment by a physiotherapist combined to 'yes' vs 'no treatment'; current or past use of GET combined to 'yes' vs 'no'; referred by rheumatologist or referred by general practitioner combined to 'referral by doctor' vs 'direct access'; duration of treatment >5 years and 1-5 years combined to '1year+' versus <6 months and 6 months-1 year combined to 'once' versus twice or 3 times and more per week combined to 'twice+'.

The parameters of the logistic regression models are log odds ratios (LOR):  $\log O_{_{NL}}/O_{_{CH}} = \log O_{_{NL}} - \log O_{_{CH}}$  for the event given by the second level of the outcome variable, mostly 'yes'. We reported the exponentiated values (odds ratios).

The level of significance was set at  $\alpha$ = 0.05. The R language and environment for statistical computing (http://www.Rproject.org, 2018) was used for the statistical analyses.

# Results

### Demographics

In total, 713 people participated; 206 in NL (response rate 45%) and 507 in CH (response rate 29%; 0.5% used paper version). Approximately 57% of participants were male, with a median (range) disease duration of 16 (1-65) years and median (range) ASAS-HI score of 4.2 (0-14.2). The Dutch cohort was statistically different with regard to sex, age, disease duration, ASAS-HI score and the use of pain medication (Table 1).

Table 1.	Characteristics	of D	Outch	and	Swiss	people	with	axial	spondyloarthritis	participating	in	а
survey o	n physiotherapy											

Characteristic	Total (n=713)	NL (n=206)	CH (n=507)	p-value*
Sex, male, n (%)	404 (56.7)	142 (69.3)	262 (51.7)	<0.001
Age, years, median (range)	55.0 (21-94)	58.0 (24-94)	53.5 (21-85)	<0.001
Disease duration, years, median (range)	16 (1-65)	24 (1-58)	13 (1-65)	<0.001
Current drug treatment Pain medication (e.g. paracetamol), n (%)	206 (29.0)	82 (39.8)	124 (24.5)	<0.001
Anti-inflammatory pain medication (NSAIDs), n (%) DMARDs, n (%)	424 (59.5) 103 (14.7)	125 (60.7) 25 (12.1)	300 (59.2) 78 (15.4)	0.73 0.26
Biologic, n (%)	270 (38.0)	81 (39.3)	189 (37.3)	0.61
No axSpA-related drugs, n (%)	94 (13.3)	16 (7.8)	78 (15.4)	0.06
ASAS Health Index, median (range)	4.2 (0-14.9)	5.7 (0-14.9)	4.3 (0-14.9)	< 0.001

Abbreviations: NL=The Netherlands, CH=Switzerland, ASAS= Assessment of Spondyloarthritis International Society, NSAIDS= Nonsteroidal anti-inflammatory drugs, DMARDS= Disease modifying anti-rheumatic drug, axSpA= axial Spondyloarthritis. \*p-value of Chi Square or Mann-Whitney U-Test.

More than one-third (36.7%) of participants had or had been using individual physiotherapy but never attended a GET, 29.1% had used or had been using a combination of individual plus land- or water-based GET, and 5.3% had used or had been using land- and water-based GET only (see Table 2).

### Use of individual physiotherapy

In total, 83.3% of the patients were currently or had been treated by a physiotherapist individually (1:1) (Table 3). Direct access to physiotherapy was used by 17.1%. However, the chance of being referred to physiotherapy by a general practitioner or rheumatologist, in contrast to going on ones' own initiative, was 2.7 times higher in Switzerland than in NL (adjusted odds ratio 2.74, 95% Cl 1.57,4.83) (Table 3).

Regarding individual physiotherapy content (Table 3) most participants receive a combination of active (70.4%) or (assisted) passive (75.2%) flexibility interventions, massage (53.6%), and instructions of home exercises (67.7%).

Setting	Total n=713 No (%)	NL n=205ª No (%)	CH n=506ª No (%)
Individual therapy only	262 (36.7)	102 (49.7)	160 (31.6)
GET only			
- Land-based GET only	14 (1.9)	3 (1.4)	11 (2.1)
- Water-based GET only	3 (0.4)	O (O)	3 (0.5)
- combination land- and water-based GET	38 (5.3)	5 (2.4)	33 (6.5)
Combination of individual and GET			
- Combination individual with land-based GET	105 (14.7)	32 (15.6)	73 (14.4)
- Combination individual with water-based GET	19 (2.6)	9 (4.3)	10 (1.9)
- Combination individual with land and water-based GET	208 (29.1)	37 (18.0)	171 (33.7)
Never used any kind of therapy	62 (8.6)	17 (8.8)	45 (8.8)

### Table 2. Use of individual and group exercise therapy by people with axial spondyloarthritis

GET= group exercise therapy, NL= the Netherlands, CH= Switzerland, "one individual did not answer those questions

Table 3.	Use	and	content	of	individual	physiotherapy	by	Dutch	and	Swiss	people	with	axial
spondylo	arthri	tis											

		Total	NL	СН	p-value*	Adjusted odds ration (95% CI)
Cu tre	rrent or past individual physiotherapy eatment	n=713 No (%)	n=206 No (%)	n=507 No (%)	yes vs. no	yes vs. no
•	Currently	233 (32.6)	90 (43.7)	143 (28.2)	0.07	1.41 (0.86-2.39)
•	In the past	362 (50.7)	90 (43.7)	272 (53.6)		
•	Never	118 (16.5)	26 (12.0)	92 (18.1)		
Re	ferral	n=437 No (%)	n=169 No (%)	n=268 No (%)		Direct vs. referral
•	Direct access	75 (17.1)	44 (26.0)	31 (11.6)	<0.001	2.74 (1.57-4.83)
•	Referral by GP	130 (29.7)	32 (18.9)	98 (36.6)		
•	Referral by rheumatologist or rheumatology nurse specialist	226 (51.7)	89 (52.7)	137 (51.1)		
•	Other	6 (1.3)	4 (3.4)	2 (0.7)		
Dı	ration of treatment	n=232 No (%)	n=89 No (%)	n=143 No (%)		<1y vs.1y+
•	>5 years	132 (56.8)	67 (75.2)	65 (45.5)	0.06	0.57 (0.23-1.34)
•	1-5 years	63 (27.1)	13 (14.6)	50 (34.8)		
•	6 months- 1 year	11 (4.7)	4 (4.4)	7 (4.9)		
•	< 6 months	26 (11.2)	5 (5.6)	21 (14.7)		
Fr	equency	n=230 No (%)	n= 89 No (%)	n=141 No (%)		≤once vs. twice+
•	< 1 per week	99 (43.0)	45 (50.5)	54 (38.3)	0.08	0.58 (0.31-1.06)
•	Once per week	106 (46.0)	31 (34.8)	75 (53.2)		
٠	Twice per week	23 (10.0)	13 (14.6)	10 (7.1)		
•	Three times or more per week	2 (0.8)	0 (0)	2 (1.4)		
Co	ontent	n=598 No (%)	n=180 No (%)	n=418 No (%)		
Ea	lucation					
Ec	lucation on coping with limitations	128 (21.4)	60 (33.3)	68 (16.3)	<0.001	2.11 (1.35-3.27)
Ec	lucation on sports and physical activity	206 (34.4)	72 (40.0)	134 (32.0)	0.13	1.4 (0.99-2.19)

### Table 3. Continued

	Total	NL	СН	p-value*	Adjusted odds ration (95% CI)
Instruction on home exercises	405 (67.7)	121 (67.2)	284 (67.9)	0.51	0.87 (0.58-1.30)
Exercises					
Cardiovascular (Aerobic) exercises	105 (17.5)	40 (22.2)	65 (15.6)	0.08	1.26 (0.77-2.03)
Muscle strengthening exercises	262 (43.8)	76 (42.2)	186 (44.5)	0.93	0.83 (0.56-1.22)
Active range of motion / flexibility exercises	275 (70.4)	70 (38.8)	205 (49.0)	0.01	0.58 (0.39-0.85)
Balance exercises	94 (15.7)	31 (17.2)	63 (15.0)	0.62	1.09 (0.64-1.83)
Relaxation exercises	21 (3.5)	6 (3.3)	15 (3.6)	1.00	0.94 (0.32-2.45)
Passive range of motion exercises	262 (43.8)	99 (55.0)	163 (38.9)	0.00	2.13 (1.45-3.15)
Passive assisted range of motion exercises	188 (31.4)	54 (30.0)	134 (32.0)	0.50	0.98 (0.65-1.46)
Other physiotherapy treatment					
Heat treatment	126 (21.0)	17 (9.4)	109 (26.0)	<0.001	0.28 (0.15-0.49)
Cold treatment	13 (2.1)	3 (1.6)	10 (2.4)	0.76	0.61 (0.13-2.10)
Massage	321 (53.6)	90 (50.0)	231 (55.3)	0.11	0.80 (0.55-1.18)
Kinesiotaping	64 (10.7)	3 (1.6)	61 (15.1)	< 0.001	0.14 (0.03-0.41)
US	97 (16.2)	33 (18.3)	64 (15.3)	0.47	1.19 (0.71-1.97)
Dry needling	29 (4.8)	6 (3.3)	23 (5.5)	0.30	0.67 (0.23-1.65)
Reasons for stopping (if applicable)	n=362 No (%)	n= 90 No (%)	n=272 No (%)		
Not necessary anymore	73 (20.1)	16 (17.8)	57 (20.9)	0.64	0.70 (0.37-1.25)
Being able to do the exercises at home	202 (55.8)	52 (57.7)	150 (55.1)	0.10	0.80 (0.53-1.20)
No perceived effect	75 (20.7)	17 (18.8)	58 (21.3)	0.65	0.66 (0.34-1.23)
More discomfort	37 (10.2)	12 (13.3)	25 (9.1)	0.32	1.04 (0.456-2.247)
Inadequate reimbursement (any more)	65 (17.9)	17 (18.8)	48 (17.6)	1.00	0.897 (0.46-1.65)
Other	59 (16.2)	13 (14.4)	46 (16.9)	0.40	0.41 (0.19-0.80)

NL= the Netherlands, CH= Switzerland, n= numbers, MC=multiple choice, GP = general practitioner, GET=group exercise therapy, \*p-value of Mann Whitney U, Chi Square or Fischer Exact tests

### Use of land- or water-based GET

Participants usually met once a week (median 4 times a month) for 60 min land-based or 45 min for water-based exercise. The most frequent reason for discontinuation was 'too time consuming' (22.5% for land-based and 22% for water-based GET; Table 4).

Table 4. Use and content of land- of	r water-based group	exercise therapy by [	Dutch and Swiss	patients
with axial spondyloarthritis				

	Total	NL	СН	p-value*	Adjusted odds ration (95% CI)
Land-Based GET					
Current or past land-based GET	n=712 No (%)	n=205 No (%)	n =507 No (%)	Yes vs. no	Yes vs. no
Currently	171 (24.0)	18 (8.8)	153 (30.2)	< 0.001	0.28 (0.18-0.42)
In the past	193 (27.1)	59 (28.8)	134 (26.4)		
Never	348 (48.8)	128 (62.4)	220 (43.4)		
Frequency per months (n)	n=168	n=18	n=150		
Median (range)	4 (1-10)	4 (1-4)	4 (1-10)	0.95	n.c.

#### Table 4. Continued

	Total	NL	СН	p-value*	Adjusted odds ration (95% CI)
Duration of session (minutes)	n=170	n=18	n=152		
Median (range)	60 (0-150)	90 (30-150)	60 (20-90)	< 0.001	n.c.
Reasons for stopping (if applicable, MC)	n=191 No (%)	n=57 No (%)	n=134 No (%)		
Too hard	28 (14.6)	17 (29.8)	11 (8.0)	< 0.001	3.60 (1.42-9.36)
More discomfort	22 (11.5)	8 (14.0%)	14 (10.3)	0.62	1.28 (0.42-3.63)
No motivation	37 (19.3)	12 (21.0)	25 (18.4)	0.84	0.94 (0.39-2.12)
No perceived effect	34 (17.8)	10 (17.5)	24 (17.6)	0.83	0.69 (0.26-1.69)
Too time consuming	43 (22.5)	10 (17.5)	33 (24.3)	0.25	0.92 (0.36-2.19)
Inadequate reimbursement (any more)	10 (5.2)	8 (14.0)	2 (1.5)	< 0.001	13.48 (2.00-157.03)
Water-based GET					
Current or past water-based GET?	n=712 No (%)	n=205 No (%)	n=507 No (%)	yes vs. no	yes vs. no
• currently	117 (16.4)	16 (7.8)	101 (19.9)	< 0.001	0.28 (0.18-0.43)
• in the past	150 (21.0)	34 (16.6)	116 (22.8)		
• never	445 (62.5)	155 (75.6)	290 (57.2)		
Frequency per months (n)	n=114	n=14	n=100		
Median (range)	4 (1-10)	4 (1-8)	4 (1-10)	0.05	n.c.
Duration of session (minutes)	n= 116	n=16	n=100		
Median (range)	45 (20-135)	45 (30-135)	45 (20-90)	0.38	n.c.
Reasons for stopping water-based GET (if applicable, MC)	n=150 No (%)	n=34 No (%)	n=116 No (%)		
Too hard	17 (11.3)	9 (26.5)	8 (6.8)	<0.001	9.27 (2.57-39.09)
More discomfort	12 (8.0)	4 (11.7)	8 (6.8)	0.47	2.16 (0.32-13.53)
No motivation	26 (17.3)	8 (23.5)	18 (15.4)	0.29	2.22 (0.76-6.31)
No perceived effect	25 (16.6)	11 (32.3)	14 (11.9)	<0.001	3.10 (1.09-8.77)
Too time consuming	33 (22.0)	6 (17.6)	27 (23.0)	0.63	0.81 (0.24-2.36)
Inadequate reimbursement (any more)	10 (6.6)	5 (14.7)	5 (4.3)	0.04	2.27 (0.42-10.64)

NL=The Netherlands, CH= Switzerland, n= numbers, GET=group exercise therapy, MC=multiple choice, n.c.= not calculated; \*p-value of Mann Whitney U, Chi Square or Fischer Exact tests

### Participants' motivation and preferences regarding exercise

A large proportion of participants (75.9%) were not aware of the extra risk of cardiovascular disease and osteoporosis caused by axSpA (see Table 5). However, more than two-thirds of the participants were motivated to carry out exercises to improve fitness (82.7%; see Table 5). Reasons for being unwilling or unable to exercise were 'I don't feel like it' (44.8%) for being unwilling and 'I get more discomfort' (72%) for being unable (*please Supplementary Table S1*, available at *Rheumatology advances in Practice* online).

The proportion of the participants who felt self-responsible and able to conduct an unsupervised program themselves was 42.4%. Of those participants preferring supervised exercising, 28% liked having an individual programme with face-to-face supervision by a

physiotherapist (see Table 5). Two-thirds of the participants (67.9%) preferred the supervising physiotherapist to be specialized in axSpA (see Table 5), with significantly more Swiss than Dutch participants finding this important. In contrast, 20.2% preferred exercising in a regular fitness club without specialized supervision. The ideal organized exercising setting would take place once per week, for a duration of about 1h, in the evening, but not at weekends (Table 5).

Knowledge about disease and exercise	Total	NL	СН	p-value*	Adjusted odds ration (95% CI)
Knowledge how to get information on axSpA	n=651 No (%)	n=153 No (%)	n=498 No (%)		
yes	574 (88.1)	121 (79.0)	453 (90.3)	< 0.001	n.c.
no	77 (11.8)	32 (20.9)	45 (9.0)		
Awareness of extra risk of cardiovascular diseases and osteoporosis	n=708 No (%)	n=201 No (%)	n= 507 No (%)		
yes	161 (22.7)	60 (29.8)	101 (21.7)	0.025	n.c.
no	538 (75.9)	141 (70.1)	397 (78.3)		
Willingness to improve fitness	n=704 No (%)	n=197 No (%)	n=507 No (%)		
No	49 (6.9)	29 (14.7)	20 (3.9)	< 0.001	n.c.
Yes, but not able to	72 (10.2)	33 (16.8)	39 (7.7)		
Yes, I do my best already	416 (59.0)	93 (47.2)	323 (63.7)		
Yes	167 (23.7)	42 (21.3)	125 (24.7)		
Preferences for delivery of unsupervised or	supervised exe	ercising? (Mult	tiple-Answer-O	ption)	
Unsupervised exercises preferred (MC)	n=685 No (%)	n=199 No (%)	n=486 No (%)		
General instructions via leaflet or website	128 (18.6)	37 (18.5)	91 (18.7)	0.99	1.05 (0.66-1.66)
General instructions via DVD or APP	130 (18.9)	34 (17.0)	96 (19.7)	0.84	1.01 (0.62-1.59)
Personalized program	250 (36.4)	50 (25.1)	200 (41.1)	< 0.001	0.62 (0.41-0.91)
Personal program with guidance by an expert by email, Internet or app	102 (14.8)	20 (10.0)	82 (16.8)	0.02	0.70 (0.39-1.21)
I am self-responsible and able to conduct an unsupervised exercise program	291 (42.4)	71 (35.6)	220 (45.2)	0.02	0.74 (0.51-1.06)
Supervised exercise preferred (MC)	n=670 No (%)	n=190 No (%)	n=480 No (%)		
Individual exercise programme with face-to- face supervision by PT	188 (28.0)	35 (18.4)	153 (31.2)	<0.001	0.60 (0.38-0.92)
Individual exercise with internet-based guidance (e.g. webcam)	57 (8.5)	6 (3.1)	51 (10.6)	<0.001	0.32 (0.12-0.72)
Group exercise program for axSpA patients	233 (34.7)	36 (18.9)	197 (41.0)	< 0.001	0.34 (0.22-0.52)
Regular sport activities (sport club or fitness center) supervised by sports instructor	136 (20.2)	26 (13.6)	110 (22.9)	<0.001	0.54 (0.32-0.87)
Duration per session	n=445 No (%)	n= 90 No (%)	N=355 No (%)		
• <1h	47 (10.3)	11 (12.2)	36 (10.1)	n.c.	n.c.
• 1h	313 (70.3)	50 (55.5)	263 (74.0)		
• 1.5h	61 (13.7)	17 (18.9)	44 (12.4)		
• >1.5h	24 (5.3)	12 (13.3)	12 (3.3)		

Table 5. Preferences of people with axial spondyloarthritis for content and design of education and exercise

### Table 5. Continued

Knowledge about disease and exercise	Total	NL	СН	p-value*	Adjusted odds ration (95% CI)
Frequency per week	n=440 No (%)	n= 85 No (%)	n=355 No (%)		
Once	239 (54.3)	42 (49.4)	197 (55.5)	n.c.	n.c.
• Twice	151 (34.3)	34 (40.0)	117 (32.9)		
• 3 times	50 (11.3)	7 (8.2)	43 (12.1)		
• >3 times	9 (2.0)	2 (2.3)	7 (1.9)		
Time of the day	n=503 No (%)	n=84 No (%)	n= 419 No (%)		
Morning	138 (27.4)	27 (32.1)	111 (25.9)	n. c.	n.c.
Afternoon	61 (12.1)	12 (14.3)	49 (11.4)		
Evening	241 (47.9)	35 (41.7)	206 (48.0)		
Does not matter	73 (14.5)	10 (11.9)	63 (14.7)		
During Weekends	n=440 No (%)	n=84 No (%)	n= 356 No (%)		
• Yes	123 (27.9)	27 (32.1)	96 (26.9)	n.c.	n.c.
• No	239 (54.3)	50 (59.5)	189 (53.1)		
• I don't know	78 (17.7)	7 (1.2)	71 (19.9)		
Supervisor should be expert in	n=439 No (%)	n=85 No (%)	n =354 No (%)		
Sports	76 (17.3)	24 (28.2)	52 (14.7)	<0.001	n.c.
Bone and joints	115 (26.1)	29 (34.1)	86 (24.3)	0.07	n.c.
Bone and joints and rheumatic diseases	368 (83.8)	62 (72.9)	306 (86.4)	<0.001	n.c.
How important is that supervising PT is expert specifically in axSpA?	n=515 No (%)	n=159 No (%)	n=356 No (%)		
(Very) important	350 (67.9)	83 (55.7)	267 (75.0)	<0.001	0.43 (0.28-0.67)
Not important	165 (32.0)	76 (44.3)	89 (25.0)		

axSpA: axial spondyloarthritis; CH: Switzerland; MC: multiple choice; n.c.: not calculated; NL: The Netherlands. \*p-value of Mann Whitney U, Chi Square or Fischer Exact tests, n.c.= not calculated, PT= physiotherapist

## Discussion

This survey among a sample of people with axSpA found that physiotherapy was frequently used, in both individual and GET settings. Individual therapy, mostly initiated by doctoral referral, was more often used than GET. The patients in this study seemed to be motivated to exercise in either a supervised or non-supervised, individually tailored programme; for both settings, the majority of patients found that guidance by a specialist would be required. Currently, individual therapy seemed to be based on passive interventions combined with instructions for (home) exercises. If active interventions were included in the therapy sessions, which appeared to be the case in <50%, mainly muscle strengthening and flexibility exercises were used; aerobic exercises and balance exercises, which are also recommended for people with axSpA (18), were less often promoted. Counselling or advice seemed to play only a subsidiary role.

Recently, a Dutch guideline specific for physiotherapy in axSpA was launched (30) but given that this guideline is only available in Dutch, physiotherapists may work according to international general management recommendations for axSpA (1, 17, 31-33) and use experiences from other rheumatic conditions like OA (34) or RA (35). Some axSpA guidelines clearly state that active therapy is more effective than passive therapy (16) whereas RA recommendations state that passive interventions may be considered for only a limited period (35). Physiotherapy interventions with therapeutic exercises or exercise training should be structured, i.e., incorporating goals, a treatment plan, and regular assessments (36). The patients' needs and preferences and the presence of facilitators and barriers regarding exercising should be taken into account (18). Known facilitators are higher education level, belief in the benefits of exercise, and intrinsic motivation, whereas barriers are being physically inactive, fatigue, lack of time or tiring exercises (37). Therefore, priority should be given to patients' preferences in exercise choice and conditions. A Cochrane review evaluated the effect of physiotherapeutic interventions for axSpA showed that GET was superior home exercise (10). Moreover, a group setting was found to foster adherence to exercising (17). Indeed, the social aspect of GET is well-known ('moving with friends') and was also appreciated by the Dutch and Swiss participants in the survey. However, group therapy in NL was not as often attended as in CH (8.8% vs 30.2% were currently attending GET) with the numbers and sizes of the groups declining, and people in the groups ageing (oral communication). These observations could imply that in the future we need to find alternative modes to obtain the added effect of exercising in a group, e.g., by web-based physiotherapy (38) and establishment of digital communities. But costs must also be considered, because GET was not being refunded fully for many Dutch patients, or its availability was limited, unlike the situation in CH.

Irrespective of the mode of delivery, it should be ensured that the intervention is not underdosed according to ACSM principles (15). It must be emphasized that exercising once per week, i.e. usual frequency of group exercise interventions, is not enough to fulfil the public health recommendations for physical activity. In this respect, it is noteworthy that two-thirds of the participants from both countries were not aware that regular exercising might help to reduce the extra risk of cardiovascular diseases. Although aerobic exercise is highly recommended (39), this was part of the individual physiotherapeutic intervention in only 17.5% in our study. Unfortunately, we do not know the extent to which aerobic training was performed during GET, despite the fact that this setting is ideal to promote aerobic exercises. With respect to balance exercises, these were reported by only 15.7% of patients, although people with axSpA more often have impaired balance compared with healthy controls and a higher risk of falls (11, 40).

Overall, our data underscore that the traditional focus on strength and flexibility exercises still dominates the physiotherapeutic interventions for people with axSpA and that consideration of cardiovascular and neuromotor exercises should be emphasized. Recalling the aforementioned multiplicity of physical activity recommendations, we believe that people with axSpA need more guidance to fulfil every aspect (i.e. cardiovascular, muscle strength, balance, and flexibility training). Future physiotherapy interventions should be based on physical activity recommendations in addition to patients' needs.

Regarding the patient perspective on the delivery of exercise interventions in axSpA in both countries, 67.9% of the sample thought that it is 'important' and 'very important' that the supervising physiotherapist is specialized in their condition (i.e. a specialization in rheumatic conditions/axSpA was more valued than a specialization in sports). This finding clearly underpins the need for the specialized physiotherapist.

A large proportion of the people participating in the survey in both countries signalled awareness of self-responsibility to exercise, in particular in a non-supervised setting but with tailored instructions. It should also be noted that 42.4% preferred unsupervised (non-GET setting) exercise. This need requires interventions to counsel and help a patient managing axSpA 'from a distance'. For this purpose, physiotherapists' knowledge and skills regarding counselling strategies and long-term exercise promotion need to be evaluated and, presumably, improved. Findings showed that 21% of the Dutch and 9% of the Swiss population surveyed did not know how to find information about their condition (Table 5). Physiotherapists also bear responsibility in providing information and support in disease management.

### Limitations

This study has a number of limitations. One limitation was that the selection of patients was different in NL and CH, which might explain some of the differences observed between the two groups. Another limitation was that the survey questionnaire was self-developed, and we did not ask for the content of group exercise interventions, because it was assumed that standard programmes would be used.

In addition, the process of data collection differed between countries (i.e. paper vs onlinesurvey and one question being posted differently). Nevertheless, we believe a comparison between the two nations is still useful to appraise common and different issues.

Concerning the reported differences between NL and CH in terms of the use and preferences of people with axSpA related to exercising, the comparisons were adjusted for potential confounders, such as differences in case mix or settings. However, we cannot rule out the possibility that there were other factors influencing the observed differences in habits and attitudes towards exercising.

Further research should assess the perspective of physiotherapists of the content and structure of interventions in people with axSpA. Guidelines for the physiotherapeutic management of people with axSpA, including recommendations on (long-term) exercise promotion, in addition to an implementation strategy for both nations, are needed urgently.

### Conclusions

Exercises are a commonly used intervention in people with axSpA, in both the individual and the group setting. There is an international need for implementing active exercises at appropriate doses, especially with more focus on cardiovascular exercising in the individual or GET setting. Our findings may help to develop further the patient-centred services independent of insurance systems. In particular, enabling people with axSpA to perform exercises independently would meet their needs and might enhance their daily physical activity.

### Funding

The development of the questionnaire was financially supported by the Dutch Arthritis Foundation (ReumaNederland; formerly named Reumafonds), Amsterdam (Project number: BP 14-1-161).

### Acknowledgement

We would like to thank René Braem (CEO Swiss Ankylosing Spondylitis Association) for supporting the survey among SVMB members, and Markus Ernst (ZHAW) for supporting the progress of the manuscript.

### Authors' contributions:

F.G. and T.P.M.V.V. initiated the project. F.G. developed the questionnaire BEVER and edited the Dutch data. A.-K-R.O. translated the questionnaire to German, edited the Swiss data and performed the data analysis. B.W. organised the data collection in Switzerland, F.A.G., Y.P.M.G.-R and A.J.P. organized the data collection in the Netherlands. T.P.M.V.V. and K.N. were involved in the study conceptualisation and supervised the study process. A.M. conducted the logistic regression analyses. A.-K.R.O. drafted the manuscript, but all authors reviewed it, provided comments on each draft and approved the final version.

### Conflict of interest

The authors declare no conflict of interest.

### Additional Files:

• Supplementary Table S1: Reasons of people with axSpA for not being able or willing to exercise

# References

- van der Heijde D, Ramiro S, Landewe R, Baraliakos X, Van den Bosch F, Sepriano A, et al. 2016 update of the ASAS-EULAR management recommendations for axial spondyloarthritis. Annals of the rheumatic diseases. 2017;76(6):978-91.
- Bohn R, Cooney M, Deodhar A, Curtis JR, Golembesky A. Incidence and prevalence of axial spondyloarthritis: methodologic challenges and gaps in the literature. Clin Exp Rheumatol. 2018;36(2):263-74.
- Hamilton L, Macgregor A, Toms A, Warmington V, Pinch E, Gaffney K. The prevalence of axial spondyloarthritis in the UK: a cross-sectional cohort study. BMC musculoskeletal disorders. 2015;16:392.
- Rusman T, van Vollenhoven RF, van der Horst-Bruinsma IE. Gender Differences in Axial Spondyloarthritis: Women Are Not So Lucky. Curr Rheumatol Rep. 2018;20(6):35.
- 5. Braun J, Sieper J. Ankylosing spondylitis. Lancet. 2007;369(9570):1379-90.
- Ramonda R, Marchesoni A, Carletto A, Bianchi G, Cutolo M, Ferraccioli G, et al. Patient-reported impact of spondyloarthritis on work disability and working life: the ATLANTIS survey. Arthritis research & therapy. 2016;18:78.
- 7. Strand V, Singh JA. Patient Burden of Axial Spondyloarthritis. J Clin Rheumatol. 2017;23(7):383-91.
- Noureldin B, Barkham N. The current standard of care and the unmet needs for axial spondyloarthritis. Rheumatology. 2018;57(suppl\_6):vi10-vi7.
- Mathieu S, Gossec L, Dougados M, Soubrier M. Cardiovascular profile in ankylosing spondylitis: a systematic review and meta-analysis. Arthritis care & research. 2011;63(4):557-63.
- Dagfinrud H, Kvien TK, Hagen KB. Physiotherapy interventions for ankylosing spondylitis. The Cochrane database of systematic reviews. 2008(1):CD002822.
- 11. Dursun N, Sarkaya S, Ozdolap S, Dursun E, Zateri C, Altan L, et al. Risk of falls in patients with ankylosing spondylitis. J Clin Rheumatol. 2015;21(2):76-80.
- Sahin N, Ozcan E, Baskent A, Karan A, Kasikcioglu E. Muscular kinetics and fatigue evaluation of knee using by isokinetic dynamometer in patients with ankylosing spondylitis. Acta reumatologica portuguesa. 2011;36(3):252-9.
- Peters MJ, Visman I, Nielen MM, Van Dillen N, Verheij RA, van der Horst-Bruinsma IE, et al. Ankylosing spondylitis: a risk factor for myocardial infarction? Annals of the rheumatic diseases. 2010;69(3):579-81.
- Caspersen CJ, Powell KE, Christenson GM. Physical activity, exercise, and physical fitness: definitions and distinctions for health-related research. Public health reports. 1985;100(2):126-31.
- Garber CE, Blissmer B, Deschenes MR, Franklin BA, Lamonte MJ, Lee IM, et al. American College of Sports Medicine position stand. Quantity and quality of exercise for developing and maintaining cardiorespiratory, musculoskeletal, and neuromotor fitness in apparently healthy adults: guidance for prescribing exercise. Medicine and science in sports and exercise. 2011;43(7):1334-59.
- Ward MM, Deodhar A, Akl EA, Lui A, Ermann J, Gensler LS, et al. American College of Rheumatology/Spondylitis Association of America/Spondyloarthritis Research and Treatment Network 2015 Recommendations for the Treatment of Ankylosing Spondylitis and Nonradiographic Axial Spondyloarthritis. Arthritis Rheumatol. 2016;68(2):282-98.
- 17. Millner JR, Barron JS, Beinke KM, Butterworth RH, Chasle BE, Dutton LJ, et al. Exercise for ankylosing spondylitis: An evidence-based consensus statement. Seminars in arthritis and rheumatism. 2016;45(4):411-27.
- Rausch Osthoff AK, Niedermann K, Braun J, Adams J, Brodin N, Dagfinrud H, et al. 2018 EULAR recommendations for physical activity in people with inflammatory arthritis and osteoarthritis. Annals of the rheumatic diseases. 2018;77(9):1251-60.
- Sveaas SH, Smedslund G, Hagen KB, Dagfinrud H. Effect of cardiorespiratory and strength exercises on disease activity in patients with inflammatory rheumatic diseases: a systematic review and meta-analysis. British journal of sports medicine. 2017;51(14):1065-72.
- Sveaas SHB, Provan SA, Semb AG, Hagen KB, Vollestad N, Fongen C, Olsen IC, Michelsen A, Ueland T, Aukrust P, Kvien TK, Dagfinrud H. Efficacy of high intensity exercise on disease activity and cardiovascular risk in active axial spondyloarthritis: a randomized controlled pilot study. PloS one. 2014;9(9):e108688.
- Niedermann K, Muggli C, Dagfinrud H, Hermann M, Tamborrini G, Ciurea A, Bischoff-Ferrari H. Effect of cardiovascular training on fitness and perceived disease activity in people with ankylosing spondylitis. Arthritis care & research. 2013;65(11):1844-52
- Berg IJ, van der Heijde D, Dagfinrud H, Seljeflot I, Olsen IC, Kvien TK, et al. Disease activity in ankylosing spondylitis and associations to markers of vascular pathology and traditional cardiovascular disease risk factors: a cross-sectional study. The Journal of rheumatology. 2015;42(4):645-53.
- Dagfinrud H, Halvorsen S, Vollestad NK, Niedermann K, Kvien TK, Hagen KB. Exercise programs in trials for patients with ankylosing spondylitis: do they really have the potential for effectiveness? Arthritis care & research. 2011;63(4):597-603.
- Niedermann K, Nast I, Ciurea A, Vliet Vlieland T, van Bodegom-Vos L. Barriers and facilitators of vigorous cardiorespiratory training in axial Spondyloarthritis: Surveys among patients, physiotherapists, rheumatologists. Arthritis care & research. 2018.
- von Elm E, Altman DG, Egger M, Pocock SJ, Gotzsche PC, Vandenbroucke JP, et al. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. J Clin Epidemiol. 2008;61(4):344-9.
- 26. Demmelmaier I, Lindkvist A, Nordgren B, Opava CH. "A gift from heaven" or "This was not for me". A mixed methods

approach to describe experiences of participation in an outsourced physical activity program for persons with rheumatoid arthritis. Clinical rheumatology. 2015;34(3):429-39.

- Kiltz U, van der Heijde D, Boonen A, Cieza A, Stucki G, Khan MA, et al. Development of a health index in patients with ankylosing spondylitis (ASAS HI): final result of a global initiative based on the ICF guided by ASAS. Annals of the rheumatic diseases. 2015;74(5):830-5.
- 28. SpondyloArthritis international Society https://www.asas-group.org/clinical-instruments/asas-health-index/(21 May 2018, date last accessed).
- Kiltz U, van der Heijde D, Boonen A, Braun J. The ASAS Health Index (ASAS HI) a new tool to assess the health status of patients with spondyloarthritis. Clin Exp Rheumatol. 2014;32(5 Suppl 85):S-105-8.
- van Weely SFE, Van der Giesen FJ, van Gaalen F, Van der Horst-Bruinsma I, Ramiro S, Weel AEAM, et al. Aanbevelingen voor fysiotherapie bij mensen met axiale spondyloartritis https://www.nhpr.nl/wp-content/uploads/2019/05/ Aanbevelingen-voor-fysiotherapie-axiale-SpA-2018.pdf: KNGF; 2018 (30 March 2019, date last accessed).
- Ozgocmen S, Akgul O, Altay Z, Altindag O, Baysal O, Calis M, et al. Expert opinion and key recommendations for the physical therapy and rehabilitation of patients with ankylosing spondylitis. International journal of rheumatic diseases. 2012;15(3):229-38.
- Wendling D, C. L, Paccou J, al. e. Recommendations of the French Society for Rheumatology (SRF) on the everyday management of patients with spondylarthrtis. joint Bone Spine. 2014;81:6-14.
- Braun J, van den Berg R, Baraliakos X, Boehm H, Burgos-Vargas R, Collantes-Estevez E, et al. 2010 update of the ASAS/EULAR recommendations for the management of ankylosing spondylitis. Annals of the rheumatic diseases. 2011;70(6):896-904.
- Zhang W, Moskowitz RW, Nuki G, Abramson S, Altman RD, Arden N, et al. OARSI recommendations for the management of hip and knee osteoarthritis, Part II: OARSI evidence-based, expert consensus guidelines. Osteoarthritis Cartilage. 2008;16(2):137-62.
- 35. Hurkmans EJ, Van der Giesen FJ, Bloo H et al. Therapie RDSfP. KNGF-Guideline for Physical Therapy in patients with rheumatoid arthritis. Practice guidelines. 2008. https://www.kngf.nl/binaries/content/assets/kennisplatform/onbeveiligd/ guidelines/rheumatoid\_arthritis\_practice\_guidelines\_2008.pdf (2 August 2015, date last accessed).
- WCPT. WCPT guideline for standrads in physical therapy practice. In: WCPT, editor. http://www.wcpt.org/guidelines/ standards2011.
- Fongen C, Sveaas SH, Dagfinrud H. Barriers and Facilitators for Being Physically Active in Patients with Ankylosing Spondylitis: A Cross-sectional Comparative Study. Musculoskeletal care. 2015;13(2):76-83.
- Paul L, McDonald MT, Coulter E, Brandon M, McConnachies A, Siebert S. Adherence to web-based physiotherapy in people with axial spondyloarthrtis. Rheumatology. 2019;58(3).
- Agca R, Heslinga SC, Rollefstad S, Heslinga M, McInnes IB, Peters MJ, et al. EULAR recommendations for cardiovascular disease risk management in patients with rheumatoid arthritis and other forms of inflammatory joint disorders: 2015/2016 update. Annals of the rheumatic diseases. 2017;76(1):17-28.
- Murray HC, Elliott C, Barton SE, Murray A. Do patients with ankylosing spondylitis have poorer balance than normal subjects? Rheumatology. 2000;39(5):497-500.

# 5

Understanding beliefs related to physical activity in people living with axial spondyloarthritis

- a theory-informed qualitative study

Anne-Kathrin Rausch Osthoff, Irina Nast, Karin Niedermann

BMC Rheumatology 2022; 6 (1):40

# Abstract

### Background

People living with axial spondyloarthrtis (axSpA) have an increased risk of cardiovascular diseases, which can be reduced by regular physical activity (PA) and its subset of cardiorespiratory training (CRT). To fulfil their crucial role in PA promotion, physiotherapists and other health professionals need to understand the beliefs that people living with axSpA possess concerning general PA and CRT. The aim of this study is to explore these behavioural, normative and control beliefs.

### Methods

A qualitative descriptive design approach was chosen. Five semi-structured focus group interviews with 24 individuals living with axSpA were performed. Data was analysed using structured thematic qualitative content analysis.

### Results

People with axSpA possessed multifaceted behavioural, normative and control beliefs concerning general PA and CRT. Behavioural beliefs revealed a positive attitude towards general PA, with participants mentioning numerous physical, psychological, and social benefits and only few risks. However, the conceptual difference between general PA and CRT, and the relevance of CRT, was unclear to some participants. Normative beliefs were expressed as the beliefs of significant others that influenced their motivation to comply with such beliefs, e.g. spouses, other people living with axSpA, rheumatologists. Regarding control beliefs, general PA and CRT were both mentioned as effective self-management strategies to control the disease. From experience, a high level of self-discipline, as well as technology, were shown to be useful.

### Conclusions

General PA is understood to be an important self-management strategy for people with axSpA and most participants build general PA into their daily routines. They believe that general PA beneficially impacts personal health and wellbeing. However, some participants are unaware of the difference between general PA and CRT and the important impact that this difference could have on their health. The consequences of CRT promotion for people living with axSpA should be the subject of further research.

## Introduction

Axial Spondyloarthritis (axSpA) is a chronic, inflammatory rheumatic disease that leads to structural impairments and functional limitations (1). AxSpA increases the risk of cardiovascular diseases (2, 3) and has an impact on flexibility (4), balance (5), muscle strength (6) and cardio-respiratory capacity (7).

As the cornerstones of disease management, exercise and effective drug treatment are recommended (1, 8). Correspondingly, the EULAR (European Alliance of Association for Rheumatology) physical activity (PA) recommendations for individuals with rheumatic and musculoskeletal diseases (RMDs) (9) strongly emphasize the importance of PA promotion. Evidence shows that exercising, according to the public health recommendations for health-enhancing PA, is effective, safe, and feasible for individuals with RMDs (10), and that it should be performed throughout the course of the disease (9). Exercise, particularly cardio-respiratory training (CRT), can be effective medicine reducing cardiovascular risk when the dose, i.e. intensity, duration and frequency, is adequately applied (11-14). Current disease management recommendations state that patients should be continuously integrated into exercise programs, rather than taking rest or inactive periods during or after a flare up (9, 15).

Despite existing evidence, individuals with axSpA are often physically active at low-tomoderate intensity levels and spend less time performing vigorous PA than healthy people (16, 17). In Switzerland, although 88% of individuals with axSpA are willing to improve their fitness, almost 80% are unaware of the increased risk of cardiovascular diseases (18). The best facilitator of general PA is to build intrinsic motivation, guided by enjoyment and personal interest (19). The most commonly reported barriers to general PA are diseasespecific symptoms, notably fatigue and pain (20). Previous research has shown that the most frequent facilitators, particularly to CRT, are knowledge, homogenous group composition, and high perceived motivation. Contrarily, the most frequent barriers to CRT are lack of motivation or information, hindering disease symptoms, and problem of timing in the daily routine (21).

The EULAR recommendations on PA state that all health professionals should provide advice on PA, however, interventions should be delivered by health professionals competent in PA principles and RMDs (9). Physiotherapists (PTs) are experts in PA promotion (22) and support people with RMDs to live an active lifestyle and exercise at a correct dose. PTs need to be precise in their instructions and use clear definitions and terms to achieve this. There is a difference between the concepts of being 'physically active' and 'exercising' (23), which often becomes blurred in communication with patients. Inconsistencies arise through the over-reporting or under-reporting of behavioural activity because the understanding of terms varies between the individuals involved (23).

PA is defined as, 'any bodily movement produced by skeletal muscles that results in energy expenditure above resting levels. PA broadly encompasses exercise, sports, and physical activities done as part of daily living, occupation, leisure, and active transportation' (24). Exercise is a subcategory of 'PA that is planned, structured, and repetitive and [that] has, as a final or intermediate objective, the improvement or maintenance of one or more dimensions of physical fitness' (25). Public health recommendations for health-enhancing PA encompass four exercise dimensions (aerobic, strength, neuromotor, flexibility) (24).

Numerous theoretical models have been applied that aim to change PA behaviour in people with RMDs (26, 27). Changes in awareness, beliefs, attitudes, motivation or knowledge may influence our PA behaviour. These psychological constructs are not open to direct observation, but models, such as the Theory of Planned Behaviour (TPB), support research in this field. The TPB has been used widely in research on PA behaviour (see Figure 1). It postulates that behaviour (e.g. PA) depends on intention (motivation) and ability (behavioural control) (28). In this respect, three constructs are relevant: (1) attitude toward the behaviour (a function of behavioural beliefs), defined as 'the psychological tendency that is expressed by evaluating a particular entity with some degree of favour or disfavour'; (2) subjective norm (arising from normative beliefs), defined as 'the beliefs of significant others and the extent that one wishes to comply with such beliefs of [other] people'; (3) perceived behavioural control (a function of control beliefs), which is 'the perceived ease or difficulty of performing behaviour' and is assumed 'to reflect past experience as well as anticipated impediments and obstacles' [all citations are obtained from (Chapter 3, (28)]. An individual's behavioural intention is determined by his or her attitude toward the behaviour, subjective norms, and perceived behavioural control. The latter is a central component in predicting behaviour: behavioural control co-determines behaviour either indirectly by shaping behavioural intention, or directly in cases where behavioural control is objectively limited by environmental circumstances.

Experience from clinical practice suggests that patients often are not aware of the distinction between PA and CRT and their specific health benefits. Therefore, the objective of this study was to explore the behavioural, normative and control beliefs concerning general PA and CRT in people living with axSpA. Understanding their beliefs would allow PTs and other health professionals to fulfil their crucial role in PA promotion.



**Figure 1.** Adapted pathway of Theory of Planned Behaviour (29). *Abbreviations: PA = physical activity; CRT = cardiorespiratory training.* 

## Methods

### Design

A qualitative research design with semi-structured focus groups of individuals with axSpA was chosen to explore patient beliefs concerning CRT and physical activity. The study applied a descriptive qualitative approach, as described by Sandelowski (30, 31). Focus groups are a valid qualitative method to explore people's knowledge and experiences, making use of their interaction (32, 33).

Three of the five focus groups were conceptualized primarily to learn more about the importance of PA and barriers/facilitators to CRT in individuals with axSpA (21). The other two focus groups were used to understand more about PA behaviour and technology-based CRT. All focus group interviews were performed during the early stage of a project (34) aimed at implementing PA recommendations within an axSpA exercise group concept. The data were analysed a second time with focus on beliefs.

### Ethical considerations

This study was approved by the Ethics Commission of Canton Zurich (June 2016, BASEC-Nr. 2016-00316) and all participants gave their written informed consent. The principles of good clinical practice and the declaration of Helsinki were followed (35). Reporting accords with the Consolidated Criteria for Reporting Qualitative Research (COREQ, Supplement A1) (36). The characteristics of the research team are described in Supplement A2.

### Procedures / patient selection / data collection

The Ankylosing Spondylitis Association of Switzerland (Schweizerische Vereinigung Morbus Bechterew, SVMB) recruited the participants of the focus groups by sending an invitation to all their members in the German-speaking part of Switzerland. For the barrier/ facilitator study (21), patients were purposefully sampled based on their self-reported PA participation levels (according to the International PA Questionnaire, short version (37)) and on their participation in a SVMB exercise group. Final selection was dependent upon achieving maximum variability of group participants in terms of sex, age, and part-time or full-time employment. For two focus groups, a convenience sample of people interested in technology-based exercising participated. Demographic data of the participants are provided in Table 1.

A few days before the focus groups took place, detailed participant information and an informed consent form were sent via e-mail to the participants. The same information was explained once again verbally at the beginning of the focus group meetings and participants gave their written informed consent. The focus group interviews took place either in a meeting room at the SVMB or at the Zurich University of Applied Sciences (ZHAW). A representative of the SVMB welcomed their participant members but were not present during the conduct of the focus group interview. The researchers explained to the participants that the aim of

the study was to evaluate and further develop the SVMB exercise groups, that their opinions were of great value to the project, and that the study was independent of their engagement in the SVMB exercise groups or their current physical activity level. Furthermore, participants were assured that all data would be processed confidentially and that the transcripts of the group discussions would be anonymised. The interviewers (IN, KN, AR) were unknown to the participants prior to the study and introduced themselves, together with their professional functions at the ZHAW.

Characteristics	Participants (n)
Total n	24 (100 %)
Gender Male female	15 (62.5 %) 9 (37.5 %)
Age mean±SD (range)	53±14 (28-86)
Occupation Employed Retired	18 (75.0 %) 6 (25.0 %)
Member of SVMB exercise group	8 (33.0 %)
Participation in sports club	9 (37.5 %)
Performance of regular CRT (at least 2x/week)	15 (62.5 %)

Table 1. Characteristics of	focus g	group p	participants
-----------------------------	---------	---------	--------------

SD, standard deviation; CRT, cardiorespiratory training; SVMB, Schweizerische Vereinigung Morbus Bechterew

The semi-structured focus group interview guides are detailed in Supplement B. At the beginning, participants were invited to introduce themselves with their name and, for example, age, job, recreational activities, or onset and course of disease. The interview guide contained leading and detailed open-ended questions on the following topics: appraisal of own health; personal significance of PA; and attitudes to and thoughts on PA and CRT. Three focus groups focused on the potential barriers and facilitators to PA/CRT, while two focus groups concentrated on the use of technology for PA/CRT. In addition, SVMB group exercising and expectations on the SVMB were discussed. Each focus group lasted approximately 90 minutes.

### Data analysis

Data was analysed using structured thematic qualitative content analysis, as described by Kuckartz (38). The focus group audio data was transcribed and proofread by the researchers IN (psychologist) and AR (physiotherapist). Subsequently, data was imported into the Qualitative Data Analysis Software Atlas.ti (39). Qualitative content analysis was initiated through attentive reading of the transcripts and the highlighting of important passages. Thematic main categories were defined deductively, according to the TPB's three determinants of behaviour attitudes, norms, and perceived behavioural control, as well as their associated behavioural, normative and control beliefs. In the next step, all data were coded and quotations assigned to these three main categories. In the following step, all quotations assigned to a category were compiled. Within these, subcategories were developed inductively to describe the material within each category (IN, AR). The two researchers discussed the developed subcategories and a consensus-based detailed categorisation system was defined (Table 2). Applying this system, all the data was analysed accordingly. Finally, thematic summaries were described for all three categories of the TPB. Example quotations of patients were translated from (Swiss-)German into English.

Category	Subcategories
Behavioural Beliefs	<ul> <li>Conceptualization of physical activity and cardiovascular training</li> <li>Exercise benefits</li> <li>Exercise associated risks</li> </ul>
Normative Beliefs	<ul> <li>Motivation of significant others to exercise</li> <li>Mutual commitment</li> <li>Performance pressure</li> <li>Image of people with axSpA</li> </ul>
Control Beliefs	<ul> <li>Discipline, tricks</li> <li>Key experience</li> <li>Limitation due to disease</li> <li>Pain control</li> <li>Exercise requirements in daily living (work, leisure activities)</li> <li>Exercise opportunities in daily living (work, leisure activities)</li> <li>Exercise opportunities offered by the SVMB</li> <li>Expectations on course instructors</li> <li>Support from technology</li> </ul>

axSpA, axial Spondyloarthritis; SVMB, Schweizerische Vereinigung Morbus Bechterew

### Measures of trustworthiness

The trustworthiness of the study, as described by Steinke (38), has been strengthened by various measures taken during the study process: intersubjective comprehensibility was enhanced by stringent documentation of the research process, discussion of the results within the research group, and application and documentation of the categorisation system. Sampling, maximum variation strategy, as well as methods of data collection and analysis, have been chosen as indicated by the research question. Results are highlighted based on statements to allow empirical anchoring. Finally, memos were used to reflect on experiences, perceptions, and expectations of the topic, and the beliefs of the participants in terms of reflected subjectivity. Transcripts and findings were returned to participants for comments.

# Results

Results reflect the behavioural, normative, and perceived control beliefs of individuals with axSpA concerning general PA and CRT. Table 3 (Supplement C) displays example anchor quotations that underpin the following results.

### Attitudes toward behaviour: performing general PA and CRT

All participating individuals living with axSpA had a positive attitude towards PA, reporting various important benefits: physical (improving fitness, physical well-being, active control of disease); psychological (joy, self-management strategy, improvement in self-esteem, enabling a sense of achievement); and social (reinforcing exercise group, activities with healthy people or other people living with axSpA). See Table 3 Supplement C, *e.g. "I think* axSpA also has advantages. You stay active." C6 (physical benefit); "I also feel better when I move. You have the same pain but then you know it's just a good feeling, when you don't do anything and you have pain, then it's more of a negative feeling. (...) I agree, physical activity is simply most important." D3 (psychological); "For us, group therapy is first and foremost a social gathering. More than anything else. That's the good thing about group therapy." E6 (social).

Some participants also mentioned the perceived risks of PA, such as the fear of increasing symptoms (e.g. pain) or physical overload prompting an injury or flare up. Some mentioned that they had had to discontinue their preferred sport since their diagnosis (e.g. "I'm not allowed to do contact sports anymore." B1).

For many participants, regular PA was part of their self-concept (e.g. "*I am very sporty.*" C6) and an important part of their daily routine (e.g. "*I miss something if I cannot exercise.*" C5). However, the relevance of CRT was new to some participants (e.g. "*Why are people so into cardiorespiratory training, recently, it always comes up..?*" B3) and only a few participants expressed the same positive attitude towards CRT as they did towards general PA.

Those participants who were members of a SVMB exercise group reported that CRT was taken into account too little (e.g. "*I quickly realised that the exercise group neglects cardiovascular training. Quite clearly. And that is almost more important than flexibility. You have to take care of your fitness yourself.*" A2).

Furthermore, the conceptual difference between general PA and CRT seemed to be unclear to some participants. If they were asked which kind of CRT they perform, answers were very diverse, covering a range from unspecific descriptions of general, often transport-related, activities (e.g. *"I just walk a lot, I don't have a car (...) I walk relatively fast." C2)* to detailed knowledge of CRT by ambitious athletes (e.g. *".. if I want to improve my fitness, then I have to go beyond this limit and see what pulse rate I train at. Otherwise, I lose fitness instead of improving it." D2*).

### Normative beliefs

For normative beliefs, no distinction between general PA and CRT was observable. Participants described several beliefs of significant others (e.g. spouses, friends) who had an influence on their motivation to comply with such beliefs. Further relevant people mentioned were other patients participating in axSpA exercise groups, the rheumatologist, and the physiotherapist. Many individuals with axSpA like to be experienced by others as being positive and sportive people (e.g. "The doctor at the hospital said that he likes axSpA people best because they are consistently nice and positive people. They can be devastated, but as soon as they can walk again, they laugh again. People with axSpA are basically positive." B3). This might provoke resistance in people unable to fulfil this ideal (e.g. "And especially when I'm in pain, I can't motivate myself to go to the gym because I'd embarrass myself. I'll just watch the others (..), in the fitness centre, everyone is watching you." C3). Some made a point of being independent and competent in self-management (e.g. "After that, I had a physiotherapist who helped very little through physical treatment, but developed exercises with me that I can do myself at home. I don't want her doing something with her magic hands and afterwards I feel great, I want her to show me how I can help myself. And that's the crucial thing - I don't want to spend my whole life running after a physiotherapist or a rheumatologist." C1). Others saw the weekly physiotherapy session as a matter of course (e.g. "I see my physiotherapist once a week" E5).

Exercising together with other people was perceived as fun and motivating. Arranging joint physical activity or exercise sessions helped to overcome the "weaker self". The competing atmosphere of an exercise group was perceived as both motivating (e.g. "*drill and fun*", B1; "you can compare, you can bolster each other up", A4) and a deterrent (e.g. "*I like more to exercise according to my possibilities without having to justify myself*", C5). Many participants described the axSpA exercise group as socially important (e.g. "*The social component should not be underestimated*", C1; "We have a good group, half of it goes for a *drink after exercising*", A5). Others preferred exercise groups not specifically designed for individuals with axSpA, because they did not need to think or talk about their condition (e.g. "When I go to handball I'm just like any other and I don't have to deal with my back problem", B4).

### **Control beliefs**

Both, general PA and CRT, were mentioned as effective self-management strategies to control their disease (e.g. "One must have axSpA under control, not the other way around" B1). Some reported a "key experience" that helped them to realise the importance of PA (e.g. "There is a sequence that is firmly in my memory. I could hardly move. Nevertheless, I wanted to go horse riding... when you're in so much pain, you can't move. So, I somehow got my leg up and got into the saddle. I struggled onto the horse with tears in my eyes and started riding. And the longer the horse kept moving, the better the pain got. That was a key experience for me. Although everything hurt, in the end the movement did me good." A3). Regular PA and specific exercising (mostly CRT) were perceived as a matter of discipline,

especially when someone did not enjoy it. Many individuals with axSpA are experts in how to integrate general PA and regular exercises into their daily routine (e.g. "*I walk the stairs, I work on the 5th floor, I walk in the morning, at noon, in the evening and in between when I have to go to the construction site.*" *B2*). Almost all participants of the focus groups had examples (e.g. "*I was not born an athletic person, I had to acquire that. That I really do a lot and incorporate it into everyday life.*" *A1; "I have a home trainer (...) but I think riding a bike is terrible. I leave the TV on and then I know how many minutes... and then I think, that's terrible! However, I just do it then.*" *A3*). A positive attitude and the perceived benefits of PA and specific exercise help individuals to create time and material resources for exercising (e.g. buying a home trainer bike). Less active individuals did not describe strategies of how to integrate general PA or specific exercises into their daily routine. Some participants underlined the importance of setting an exercise goal (e.g. "*When I have a goal, I do it. Then it's fun. But I need a goal. If I don't have one, I don't do anything.*" *A5*). Reaching the goal reinforced their self-efficacy.

Nonetheless, the disease is associated with obstacles to exercise, since it limits the choice of sport and can also force a change in occupational career, which is sometimes hard to accept (e.g. *"I used to be an aerobics instructor (..) and had to give that up" C3*). But participants also experienced facilitators to exercise: regular participation in group exercise and the group-leading PT are often perceived as supportive (e.g. *"Physiotherapists always have a bunch of ideas [how to exercise]" B3*).

Technology-support for general PA was viewed as both motivating and demotivating. A reminder, through a tracker function that controls steps per day, was perceived as stimulating, especially for beginners or light intensity exercisers (e.g. "*I think tracking steps per day is good for a start. Everyone has a different motivation.*" *D2; "When I get up in the morning, I plan it (training), and when I remember it, it's in the evening. In between, the day is so packed that I forget. But when I am reminded, I incorporate it more. Because I want to do it, because I know it does me good. I do it 2 times a week ... most of the time. But never without a reminder." E2).* Others felt demotivated (e.g. "*I got a tracking watch from my wife as a birthday present. But I was so bad that I gave it back to her.*" E5).

Participants agreed that if a specific CRT goal is to be reached, technology-based pulsecontrol is required (e.g. "And then there are the athletes who are training for a goal. And if you want to train properly it has to be pulse-oriented." D3).

## Discussion

The individuals with axSpA participating in our study showed multifaceted behavioural, normative and control beliefs concerning CRT and general PA. However, the conceptual difference between general PA and CRT, and the relevance of CRT, was unclear to some. General PA is acknowledged to have a beneficial impact on personal health and wellbeing. For most participants, general PA was part of their daily routine and was understood to be an important self-management strategy. Despite the importance of CRT in reducing the increased cardiovascular risk of people with axSpA, only a small number of participants seemed to know how to perform regular CRT and did so. The engagement in CRT requires skills and empowerment, such as detailed knowledge, social support, and a strong motivation to exercise in contempt of barriers, in particular, pain or fatigue.

Some implications can be derived from this finding for clinical practice, education, and research.

PA is behaviourally complex and increasing PA as a lifestyle change is challenging (28, 40). Facilitating factors for individuals with axSpA include, but are not limited to, (intrinsic) motivation, good organisational conditions, improvement of disease symptoms and health (21, 41). Barriers to an active lifestyle include, but are not limited to, low motivation, problems with timing in the daily routine, hindering disease symptoms (notably pain or fatigue) and high disease activity (21, 41, 42). In CRT, motivation is a top facilitator and top barrier (21). PTs, but also other health professionals, need to understand the complexity of PA behaviour and learn how to promote change using appropriate (behavioural change) techniques (43). For instance, general practitioners and rheumatologists tend to have a long-term relationship with their patients. However, in the context of exercise prescription, general practitioners think that nurses or PTs are more skilled to promote PA (44). Word Physiotherapy claims that physiotherapy excellence lies in the promotion, guidance, and prescription of safe PA across the life span (45). Although this might be the case for well-informed PTs (46), there is literature describing cases of PTs with misconceptions regarding knowledge, attitudes and beliefs about PA (47), and not themselves fulfilling PA recommendations (48, 49). Rethorn and colleagues (50) highlight that addressing individual (e.g. increasing PTs' knowledge, skills) and organisational factors (e.g., touching upon perceived lack of time and increasing organizational support) may improve PTs' PA promotion interventions. Verbal information on the benefits of PA and specific exercises is not usually sufficient to change behaviour. Prior to starting an exercise intervention, a PT should explore a patient's beliefs regarding PA and exercise, detect individual barriers and define a goal, according to individual preferences and through shared decisions. Sufficient coaching, exercise guidance and correctly dosed exercise could enhance the adherence to long-term PA (51). Our data show how diverse the understanding and personal approaches to PA and CRT can be, e.g. for some patients pain is a barrier, while for others it is the main reason for exercising. Tailored goals and patientoriented communication strategies are especially important in handling these differences.

However, as yet unpublished data on PTs working in Switzerland show that only a small number of behaviour change techniques are used, which underpins the need for the continuous training of PTs in communication and behavioural change techniques.

Patients also require further education and knowledge, since the power and effectiveness of correctly dosed exercise as medicine is still undervalued. Gossec and colleagues (52) described the belief that PA had negative effects on the disease as being associated with poor education and psychological issues, such as anxiety. Some participants in our study also mentioned their anxiety that performing PA could worsen a flare up or symptoms. However, the contrary is true. People living with axSpA benefit from both low and high intensity exercise (53-56). PA is associated with better function and greater exercise capacity, while sedentary behaviour is associated with poor quality of life and lower exercise capacity (57).

Further research should focus on the development of strategies on how to best implement educational interventions for health professionals and people living with axSpA, and thus provide the knowledge and skills on how to promote the recommended performance of general PA and CRT (9).

The lack of distinction between the concepts of PA and exercise is common, not only in individuals with axSpA (23, 58). If people cannot distinguish between these concepts, they see no need to engage in formal exercise and simply integrate more PA into their daily routine, increasing the duration rather than the intensity of PA. However, intensity matters in combatting the increased risk of cardiovascular disease. It is particularly important for individuals with axSpA to perform regular, moderate to vigorously dosed CRT. Regardless of disease activity, individuals with axSpA tend to perform less outdoor activities but with higher intensity compared to healthy controls (59). However, both aspects, outdoor activity and the intensity of PA, should be taken into account for axSpA management. Interestingly, there is some evidence to show that people are more likely to perform intense exercises outdoors, but – paradoxically – perceive it as less exhausting (60). The natural outdoor environment generally increases physical and psychological health (61), unsurprisingly, since weather conditions are associated with outdoor PA in people with arthritis (62).

The findings of our study are in line with others (63), underpinning the conclusion that group exercise is perceived as supportive in terms of motivation (encouraging) and peer reinforcement. However, there is no evidence that group exercise is more effective than individual exercise. Group supervisors need to be aware that a 'one-size fits all' approach is not useful and that the dose of exercises needs to be tailored to the fitness state of the individual (9). Independent of setting, patients report that continual guidance and acknowledgement, and receiving advice and feedback from a supervisor is important – especially during high-intensity exercising (55). Further research should consider how the promising area of technology-based counselling (64) could be integrated into axSpA

exercise groups to promote individually-tailored CRT.

A further focus of future research could be the objective assessment of daily PA in conjunction with PA/exercise beliefs to better understand behaviour patterns. The resulting information could be used to design a composite tool to measure core attitudes and beliefs to PA/exercise in people living with axSpA. This would be useful as a standardized tool for use across studies and to develop specific interventions. Finally, it is important to investigate whether changing the attitudes and beliefs of people has a positive impact on long-term PA/ exercise behaviour. In this regard, other relevant behaviour change theories (27), such as self-efficacy, should also be considered.

In terms of generalisability, the results of this study should be interpreted with caution. As commonly is the case in focus groups, participants had an interest in the topic under study (65). Therefore, insight was gained only into the beliefs of those individuals who are interested in PA/CRT. The study results might not be representative for those individuals with no interest in PA/CRT. However, the study participants living with axSpA presented a wide range of PA behaviour, ranging from no regular PA engagement at all to exercising in a performance-oriented manner. Furthermore, the study covered the complete range of population, in terms of age, gender and membership of the SVMB exercise group, which is an important prerequisite for the generalisability of focus groups findings (66). Additionally, since the focus groups were conducted only in the German language, generalisability is limited for those people living with axSpA in Switzerland who are not proficient in the German language.

Although the main themes of the focus groups were the barriers and facilitators to CRT and technology-based CRT, our data revealed further interesting material on the general beliefs of participants concerning CRT and PA. During analysis of results, however, it became apparent that it was unclear whether these statements referred to CRT or to general PA. In conclusion, it can be assumed that the participants' beliefs regarding CRT and PA may differ. A detailed elaboration of these differences should be the subject of future research.

# Conclusion

People living with axSpA show multifaceted behavioural, normative and control beliefs concerning CRT and general PA that underpin the beneficial impact on personal health and wellbeing (behavioural belief). For most participants, general PA is part of their daily routine and it is understood to be an important self-management strategy (control belief). However, not all participants were aware of the importance of CRT and only a few possessed the knowledge and skills to perform CRT. PTs play a key role in PA promotion and influence normative beliefs, which could be used to a greater extent in the future to promote active lifestyle competencies in people living with axSpA. Concrete training and implementation strategies at the level of patients and PTs need to be developed in the future through further research.

### Funding

The project was supported financially by physioswiss.

### Acknowledgements

We thank all participants for their openness, their willingness to discuss the issues and the time given to the focus groups. We are grateful for the support of SVMB management and particular thanks go to Bea Topalidis and René Braehm. Last but not least, our gratitude goes to Leah Reicherzer for controlling the coding of the focus groups on technology-based CRT.

### Authors' contribution:

AR drafted the manuscript, both co-authors reviewed it for important intellectual content, and both co-authors approved the final version to be published. All three authors (AR, IN, KN) were involved in all stages of the study (initialisation and execution, conception and design, acquisition of data, analysis and interpretation of data).

### Additional Files:

- Supplement A1: COREQ checklist
- Supplement A2: characteristics of the research team
- Supplement B: Key questions of semi-structured focus group guides
- Supplement C: Results supporting citations

# References

- van der Heijde D, Ramiro S, Landewe R, Baraliakos X, Van den Bosch F, Sepriano A, et al. 2016 update of the ASAS-EULAR management recommendations for axial spondyloarthritis. Annals of the rheumatic diseases. 2017;76(6):978-91.
- Mathieu S, Gossec L, Dougados M, Soubrier M. Cardiovascular profile in ankylosing spondylitis: a systematic review and meta-analysis. Arthritis Care Res. 2011;63(4):557-63.
- Nigil Haroon N, Paterson MJ, Li P, Inman RD, Haroon N. SAT0272 Increased Risk of Vascular Mortality in Axial Spondyloarthritis. Annals of the Rheumatic Disease. 2015;74(2).
- Dagfinrud H, Kvien TK, Hagen KB. Physiotherapy interventions for ankylosing spondylitis. Cochrane Database Syst Rev. 2008(1):CD002822.
- Dursun N, Sarkaya S, Ozdolap S, Dursun E, Zateri C, Altan L, et al. Risk of falls in patients with ankylosing spondylitis. J Clin Rheumatol. 2015;21(2):76-80.
- Sahin N, Ozcan E, Baskent A, Karan A, Kasikcioglu E. Muscular kinetics and fatigue evaluation of knee using by isokinetic dynamometer in patients with ankylosing spondylitis. Acta Reumatol Port. 2011;36(3):252-9.
- Peters MJ, Visman I, Nielen MM, Van Dillen N, Verheij RA, van der Horst-Bruinsma IE, et al. Ankylosing spondylitis: a risk factor for myocardial infarction? Ann Rheum Dis. 2010;69(3):579-81.
- Noureldin B, Barkham N. The current standard of care and the unmet needs for axial spondyloarthritis. Rheumatology. 2018;57(suppl\_6):vi10-vi7.
- Rausch Osthoff AK, Niedermann K, Braun J, Adams J, Brodin N, Dagfinrud H, et al. 2018 EULAR recommendations for physical activity in people with inflammatory arthritis and osteoarthritis. Annals of the rheumatic diseases. 2018;77(9):1251-60.
- Rausch Osthoff AK, Juhl CB, Knittle K, Dagfinrud H, Hurkmans E, Braun J, et al. Effects of exercise and physical activity promotion: meta-analysis informing the 2018 EULAR recommendations for physical activity in people with rheumatoid arthritis, spondyloarthritis and hip/knee osteoarthritis. RMD Open. 2018;4(2):e000713.
- 11. Services USDoHaH. Physical Activity Guidelines for Americans, 2nd edition. 2020.
- 12. Nystoriak MA, Bhatnagar A. Cardiovascular Effects and Benefits of Exercise. Front Cardiovasc Med. 2018;5:135.
- Sveaas SH, Smedslund G, Hagen KB, Dagfinrud H. Effect of cardiorespiratory and strength exercises on disease activity in patients with inflammatory rheumatic diseases: a systematic review and meta-analysis. British journal of sports medicine. 2017;51(14):1065-72.
- Sveaas SH, Provan,SA, Semb AG, Hagen KB, Vollestad N, Fongen C, Olsen IC, Michelsen A, Ueland T, Aukrust P, Kvien TK, Dagfinrud H. Efficacy of high intensity exercise on disease activity and cardiovascular risk in active axial spondyloarthritis: a randomized controlled pilot study. PloS one. 2014;9(9):e108688.
- Ward MM, Deodhar A, Gensler LS, Dubreuil M, Yu D, Khan MA, et al. 2019 Update of the American College of Rheumatology/Spondylitis Association of America/Spondyloarthritis Research and Treatment Network Recommendations for the Treatment of Ankylosing Spondylitis and Nonradiographic Axial Spondyloarthritis. Arthritis & rheumatology. 2019;71(10):1599-613.
- O'Dwyer T, O'Shea F, Wilson F. Decreased physical activity and cardiorespiratory fitness in adults with ankylosing spondylitis: a cross-sectional controlled study. Rheumatology international. 2015;35(11):1863-72.
- van Genderen S, Boonen A, van der Heijde D, Heuft L, Luime J, Spoorenberg A, et al. Accelerometer Quantification of Physical Activity and Activity Patterns in Patients with Ankylosing Spondylitis and Population Controls. The Journal of rheumatology. 2015;42(12):2369-75.
- Rausch Osthoff AK, van der Giesen F, Meichtry A, Walker B, Van Gaalen F, Goekoop-Ruitermans YPM, et al. The perspective of people with axial spondyloarthritis regarding physiotherapy: room for the implementation of a more active approach. Rheumatology Advances in Practice. 2019;0:1-10.
- 19. Brophy S, Cooksey R, Davies H, Dennis MS, Zhou SM, Siebert S. The effect of physical activity and motivation on function in ankylosing spondylitis: a cohort study. Seminars in arthritis and rheumatism. 2013;42(6):619-26.
- 20. Rasmussen JO, Primdahl J, Fick W, Bremander A. Physical activity in people with axial spondyloarthritis and the impact of overall attitudes, barriers, and facilitators: A cross-sectional study. Musculoskeletal care. 2020;18(4):510-8.
- Niedermann K, Nast I, Ciurea A, Vliet Vlieland T, van Bodegom-Vos L. Barriers and Facilitators of Vigorous Cardiorespiratory Training in Axial Spondyloarthritis: Surveys Among Patients, Physiotherapists, and Rheumatologists. Arthritis care & research. 2019;71(6):839-51.
- 22. Verhagen E, Engbers L. The physical therapist's role in physical activity promotion. British journal of sports medicine. 2009;43(2):99-101.
- 23. Dasso NA. How is exercise different from physical activity? A concept analysis. Nurs Forum. 2019;54(1):45-52.
- 24. Garber CE, Blissmer B, Deschenes MR, Franklin BA, Lamonte MJ, Lee IM, et al. American College of Sports Medicine position stand. Quantity and quality of exercise for developing and maintaining cardiorespiratory, musculoskeletal, and neuromotor fitness in apparently healthy adults: guidance for prescribing exercise. Medicine and science in sports and exercise. 2011;43(7):1334-59.
- Caspersen CJ, Powell KE, Christenson GM. Physical activity, exercise, and physical fitness: definitions and distinctions for health-related research. Public Health Rep. 1985;100(2):126-31.
- 26. Knittle KDG, V.; Maes, S. Lifestyle- and behaviour-change interventions in musculoskeletal conditions. Best Practice and

Research: Clinical Rheumatology. 2012;26(3):293-304.

- 27. Larkin L, Kennedy N, Gallagher S. Promoting physical activity in rheumatoid arthritis: a narrative review of behaviour change theories. Disability and rehabilitation. 2015;37(25):2359-66.
- 28. Biddle SJH, Mutrie N. Psychology of Physical Activity. London: Routledge; 2001.
- 29. Kan MPH, Fabrigar LR. Theory of Planned Behavior. Zeigler-Hill V, Shackelford TK, editors. Cham: Springer; 2017.
- 30. Sandelowski M. Focus on research methods: whatever happened to qualitative description? Research in nursing & health. 2000;23:334-40.
- 31. Sandelowski M. What's in a name? Qualitative description revisited. Research in nursing & health. 2010;33(1):77-84.
- Kitzinger J. Focus group research: using group dynamics to explore perceptions, experiences and understandings. In: Holloway I, editor. Qualitative Research in Health Care. Maidenhead: open university press; 2005. p. 56-70.
- 33. Kitzinger J. Qualitative research: Introducing focus group. British Medical Journal. 1995;311(299).
- Rausch Osthoff AK, Vliet Vlieland TPM, Meichtry A, van Bodegom-Vos L, Topalidis B, Buchi S, et al. Lessons learned from a pilot implementation of physical activity recommendations in axial spondyloarthritis exercise group therapy. BMC Rheumatology. 2022;6(12).
- General Assembly of the World Medical A. World Medical Association Declaration of Helsinki: ethical principles for medical research involving human subjects. J Am Coll Dent. 2014;81(3):14-8.
- Tong A, Sainsbury P, Craig J. Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups. Int J Qual Health Care. 2007;19(6):349-57.
- Craig CL, Marshall AL, Sjostrom M, Bauman AE, Booth ML, Ainsworth BE, et al. International physical activity questionnaire: 12-country reliability and validity. Medicine and science in sports and exercise. 2003;35(8):1381-95.
- Steinke I. Gütekritierien qualitativer Forschung. In: Flick U, von Kardoff E, Steinke I, editors. Qualitative Forschung Ein Handbuch. Reinbek bei Hamburg: Rowohlt Taschenbuch Verlag GmbH; 2009.
- 39. GmbH SSD. Atlas.ti Qualitative Data Analysis. 7-9 ed. https://atlasti.com/ 2021.
- 40. Gabriel KKP, Morrow JR, Woolsey ALT. Framework for Physical Activity as a Complex and Multidimensional Behavior. Journal of physical activity & health. 2012;9:S11-S8.
- 41. Liu SH, Morais SA, Lapane KL, Kay J. Physical activity and attitudes and perceptions towards physical activity in patients with spondyloarthritis: A systematic review. Seminars in arthritis and rheumatism. 2020;50(2):289-302.
- 42. Fongen C, Sveaas SH, Dagfinrud H. Barriers and Facilitators for Being Physically Active in Patients with Ankylosing Spondylitis: A Cross-sectional Comparative Study. Musculoskeletal care. 2015;13(2):76-83.
- 43. Kunstler BE, Cook JL, Freene N, Finch CF, Kemp JL, O'Halloran PD, et al. Physiotherapists use a small number of behaviour change techniques when promoting physical activity: A systematic review comparing experimental and observational studies. Journal of Science and Medicine in Sport. 2018;21(6):609-15.
- Persson G, Brorsson A, Ekvall Hansson E, Troein M, Strandberg EL. Physical activity on prescription (PAP) from the general practitioner's perspective - a qualitative study. BMC family practice. 2013;14:128.
- World\_Physiotherapy. Physical therapists as exercise and physical activity experts across the life span. Policy statement. https://worldphysio/sites/default/files/2020-04/PS-2019-Exercise-expertspdf: World Confederation for Physical Therapy; 2019.
- Shirley D, van der Ploeg HP, Bauman AE. Physical activity promotion in the physical therapy setting: perspectives from practitioners and students. Physical therapy. 2010;90(9):1311-22.
- Mounton A, Mugnier B, Demoulin C, Cloues M. Physical Therapists' Knowledge, Attitudes, and Beliefs about Physical Activity: A Prerequisite to their Role in Physical Activity Promotion? Journal of Physical Therapy Education. 2014;28(3):120-7.
- Neil-Sztramko SE, Ghayyur A, Edwards J, Campbell KL. Physical Activity Levels of Physiotherapists across Practice Settings: A Cross-Sectional Comparison Using Self-Report Questionnaire and Accelerometer Measures. Physiother Can. 2017;69(2):152-60.
- Lowe A, Littlewood C, McLean S, Kilner K. Physiotherapy and physical activity: a cross-sectional survey exploring physical activity promotion, knowledge of physical activity guidelines and the physical activity habits of UK physiotherapists. BMJ Open Sport Exerc Med. 2017;3(1):e000290.
- Rethorn ZD, Covington JK, Cook CE, Bezner JR. Physical Therapists' Knowledge, Skills, Beliefs, and Organizations Impact Physical Activity Promotion: A Systematic Review and Meta-Analysis. Physical therapy. 2022;102(3): pzab291
- 51. Bilberg A, Dagfinrud H, Sveaas SH. Supervised intensive Exercise strengthen Exercise Health Beliefs in Patients with Axial Spondyloarthritis: A Multicentre Randomized Controlled Trial. Arthritis care & research. 2022; 74(7):1196-1204.
- Gossec L, Berenbaum F, Chauvin P, Hudry C, Cukierman G, de Chalus T, et al. Development and application of aquestionnaire to assess patient beliefs in rheumatoid arthritis and axial spondyloarthritis (vol 37, pg 2649, 2018). Clinical Rheumatology. 2018;37(10):2659-2657.
- O'Dwyer T, McGowan E, O'Shea F, Wilson F. Physical Activity and Exercise: Perspectives of Adults With Ankylosing Spondylitis. Journal of physical activity & health. 2016;13(5):504-13.
- Demmelmaier I, Lindkvist A, Nordgren B, Opava CH. "A gift from heaven" or "This was not for me". A mixed methods approach to describe experiences of participation in an outsourced physical activity program for persons with rheumatoid arthritis. Clinical Rheumatology. 2015;34(3):429-39.
- 55. Bilberg A, Sveaas SH, Dagfinrud H, Mannerkorpi K. How Do Patients With Axial Spondyloarthritis Experience High-Intensity Exercise? Acr Open Rheumatol. 2020;2(4):207-13.
- Sveaas SH, Bilberg A, Berg IJ, Provan SA, Rollefstad S, Semb AG, et al. High intensity exercise for 3 months reduces disease activity in axial spondyloarthritis (axSpA): a multicentre randomised trial of 100 patients. British journal of sports medicine. 2020;54(5):292-297.
- 57. Coulter EH, McDonald MT, Cameron S, Siebert S, Paul L. Physical activity and sedentary behaviour and their associations with clinical measures in axial spondyloarthritis. Rheumatology International. 2020;40(3):375-81.
- Hurley M, Dickson K, Hallett R, Grant R, Hauari H, Walsh N, et al. Exercise interventions and patient beliefs for people with hip, knee or hip and knee osteoarthritis: amixedmethods review. Cochrane Database of Systematic Reviews. 2018(4).
- Fongen C, Halvorsen S, Dagfinrud H. High disease activity is related to low levels of physical activity in patients with ankylosing spondylitis. Clinical Rheumatology. 2013;32(12):1719-25.
- 60. Focht BC. Brief walks in outdoor and laboratory environments: effects on affective responses, enjoyment, and intentions to walk for exercise. Res Q Exerc Sport. 2009;80(3):611-20.
- 61. Aliyas Z. Physical, mental, and physiological health benefits of green and blue outdoor spaces among elderly people. Int J Environ Health Res. 2021;31(6):703-14.
- Timmermans EJ, van der Pas S, Dennison EM, Maggi S, Peter R, Castell MV, et al. The Influence of Weather Conditions on Outdoor Physical Activity Among Older People With and Without Osteoarthritis in 6 European Countries. Journal of physical activity & health. 2016;13(12):1385-95.
- 63. Hilberdink B, van der Giesen F, Vliet Vlieland T, Nijkamp M, van Weely S. How to optimize exercise behavior in axial spondyloarthritis? Results of an intervention mapping study. Patient education and counseling. 2020;103(5):952-9.
- 64. Li LDC, Feehan LM, Xie H, Lu N, Shaw CD, Gromala D, et al. Efficacy of a Physical Activity Counseling Program With Use of a Wearable Tracker in People With Inflammatory Arthritis: A Randomized Controlled Trial. Arthritis care & research. 2020;72(12):1755-65.
- 65. Acocella I. The focus groups in social research: advantages and disadvantages. Qual Quant. 2012;46:1125-36.
- 66. Bloor M, Frankland J, Thomas M, Robson K. Conclusions. Focus Groups in Social Research. Introducing Qualitative Methods 2001.London, SAGE Publications

# 6

# Reliability of an adapted core strength endurance test battery in individuals with axial spondylarthritis

Anne-Kathrin Rausch, Philipp Baltisberger, André Meichtry, Beatrice Topalidis, Adrian Ciurea, Theodora P M Vliet Vlieland, Karin Niedermann

> Clinical Rheumatology 2021; 40 (4): 1352-1360 doi.org/10.1007/s10067-020-05408-6

Research Award of the Swiss Ankylosing Spondylitis Foundation, video interview (June 2021), presentation at Swiss Rheumatology Congress Interlaken (2022)

# Abstract

#### Objectives

To adapt the core strength endurance test battery (aCSE), previously used for testing athletes, to a target group of patients with axial spondylarthritis (axSpA), to evaluate its intra-tester reliability and its associations with disease-specific factors.

#### Methods

A cross-sectional study was conducted at axSpA exercise therapy groups, including both axSpA patients and the physiotherapist group leaders (PTs). The aCSE was used to measure the isometric strength endurance of the ventral, lateral and dorsal core muscle chains (measured in seconds), as well as to assess the disease-specific factors of functional status, self-reported pain, and perceived strength performance. The aCSE was repeated after 7-14 days to measure intra-tester reliability for the same rater (PT group leader). Reliability was calculated as an intra-class correlation coefficient (ICC) using a nested design. The associations between ventral, lateral, and dorsal strength endurance and the disease-specific factors were calculated using Pearson correlation coefficients.

#### Results

Study participants were 13 PT group leaders and 62 axSpA patients. The latter were all capable of performing the aCSE, with the exception of one individual. A moderate to substantial intra-rater reliability (ICCs (95%CI)) was found for the ventral (0.54 (0.35, 0.74)), lateral (0.52 (0.33, 0.70)) and dorsal (0.71 (0.58, 0.86)) core muscle chains. None of the aCSE measures correlated with the disease-specific factors.

#### Conclusion

The aCSE was found to be a reliable test battery for assessing core strength endurance in axSpA patients. Interestingly, aCSE performance was not associated with any disease-specific factors.

#### Key points

- The adapted core strength endurance test battery measures the isometric strength of the ventral, lateral and dorsal core muscle chains.
- The adapted core strength endurance test battery showed a moderate to substantial intra-rater reliability for all three muscle chains tested in axSpA patients.
- No correlations were found between the adapted core strength endurance test battery and the disease-specific factors of self-reported pain, functional status and perceived strength performance.

# Introduction

Axial spondylarthritis (axSpA) is a chronic inflammatory rheumatic condition that mainly affects the axial skeleton, iliosacral joints, and spine. It can lead to structural and functional impairments and can have a significant impact on an individual's quality of life and working ability (1, 2).

The cornerstones of non-pharmacological treatment are patient education and regular exercise (1). Current recommendations for physical activity in individuals with inflammatory arthritis emphasize that the general physical activity recommendations, comprising the four exercise domains of cardiorespiratory fitness, muscle strength, flexibility and neuromotor performance, are effective, safe and feasible for axSpA patients (3).

In addition to peripheral muscle strength, core muscle strength (sometimes referred to as the "powerhouse") is especially important for people with axSpA (4), since inflammation and reduction in mobility affect the dynamic stabilization of the spine. A certain level of core stability is needed, in terms of strength and muscle fatigue resistance, for the activities of daily living (improved posture, enhanced balance and proprioception) and sports performance (such as correct and safe barbell-based exercising) (5). Core muscle performance is a complex and multivariable construct, with core stability not yet having been clearly defined (6, 7). Experts agreed in a Delphi-project that core stability is, "the ability to achieve and sustain control of the trunk region at rest and during precise movement" and depends on the components of muscle strength and neuromotor control (8). Three common categories to assess core muscle strength can be differentiated according to this concept: maximal strength, power (9, 10) and strength endurance, the latter defined as the ability to sustain a given level of force production over time (11).

No randomized controlled studies have been published to date investigating the effect of muscle strength exercises alone on lower limb and trunk muscle strength performance in axSpA patients (12). Training the trunk muscles does seem, however, to have a positive effect on flexibility in healthy people (13). Moreover, the combination of core strength and flexibility exercises has been found to have a positive effect on disease activity and flexibility in axSpA patients (assessed by the Bath Ankylosing Spondylitis Mobility Index, BASMI) (14). The physical activity recommendations for individuals with inflammatory arthritis (3) underline the fact that regular evaluation of physical fitness, including all afore mentioned four domains, should be part of standard care for axSpA patients (3). The assessment of strength, however, is challenging. Dagfinrud et al. found that the monitoring of muscle strength parameters in group exercise therapy (GET) studies was essentially nonexistent (15). A survey of the axSpA GET in Switzerland and the Netherlands confirmed this finding for clinical practice (16). However, there is a need to regularly assess all fitness dimensions to identify those with the potential for improvement. This could help healthcare providers and patients in the process of promoting physical activity, e.g. giving advice, setting goals,

planning and performance of exercises.

The Ankylosing Spondylitis Association of Switzerland (Schweizerische Vereinigung Morbus Bechterew, SVMB) currently has more than 4000 members and organizes more than 60 exercise groups across Switzerland. Participants exercise weekly in groups led by physiotherapists (PTs) in land-based or water-based settings. The objectives of the GET are to minimize the progressive spinal mobility restriction (17), reduce cardiovascular and biological risk factors (18), and maintain or increase muscle strength. According to the recommendations (3), annual fitness assessments are also part of the SVMB's quality concept. In clinical settings, such as the axSpA GET, the fitness dimensions should be assessed by easy-to-use and inexpensive devices. In this target group, strength assessment should focus on core muscle strength endurance rather than peripheral muscle strength.

In the absence of an existing gold standard, the isometric core strength endurance test battery (CSE), originally developed by the Swiss Olympic Medical Centers for use with athletes, seems to be the only available tool for use in an axSpA GET setting. It was found to be easy to perform, inexpensive to use and to present with good psychometric properties (19). The Biering-Sorensen test, which is part of the aCSE, showed good to excellent interrater and test-retest reliability (20), but no prior data on intra-rater reliability were found. The CSE was designed to evaluate the "basic" core strength of athletes, meaning the minimum strength required for the performance of sports (19). With the aid of a reference table including norm data of Swiss athletes, assessors judge whether core strength is sufficient or insufficient.

The aim of this study was to adapt the CSE for axSpA patients (aCSE), evaluate intra-tester reliability and the associations with disease-specific factors.

# Methods

#### Design

For the analysis of reliability, a cross-sectional study with a nested design was conducted, given that across the groups, a different numbers of individuals were measured by the PTs (raters). The findings are reported in line with the GRRAS (Guidelines for Reporting Reliability and Agreement Studies) guidelines (21).

#### Participants

#### Physiotherapists (PT)

The PTs leading the GET in the German-speaking region of Switzerland (n=45) were invited to participate in this study and to perform and rate the aCSE test battery with their group participants. They were requested to attend a 2h practical training session (lead by PB). In addition, they received detailed step-by-step explanations in the form of videos, photos, and handouts. Thirteen PTs registered for the aCSE training.

#### AxSpA patients

An information letter was sent to the 206 participants of the SVMB exercise groups of the 13 PTs registered for the aCSE training. The letter explained the purpose and procedures of the study and described the aCSE test battery. Inclusion criteria were age of over 18 years and sufficient German language skills. Eligible individuals also had to be capable of getting down onto the floor and lying in the prone, supine and side positions. Exclusion criteria were heart diseases of class three or four, according to the New York Heart Association (22), diagnosed osteoporosis of grade two or more (23) or surgical spondylodesis performed on the entire lumbar spine.

#### Assessments

The aCSE test battery was performed by each axSpA patient twice with a period of 1 to 2 weeks between the tests (T0 and T1). A period of 1 week allowed for full recovery of the muscles after performing the first aCSE test battery at T0, while the maximum of 2 weeks minimized the possible effect of disease-related changes. The measurements took place during the regular GET. At T0, as well as recording patient characteristics, measurements of pain status, disease activity functional restriction, and physical fitness were made. The same group PT leader conducted the aCSE measurements at both time points with each patient. No encouragement (in terms of cheering) was allowed.

#### Participant characteristics

Gender, age, disease duration, and self-reported exercise hours per week were recorded for axSpA patients. Gender, age, and work experience in years were documented for the PTs.

#### Core strength measurements

Adaptation and performance of the core strength endurance test battery (aCSE) Given the large number of exercise groups that could potentially implement the aCSE, the criteria for test selection were validity, reliability, and low cost. A further important criterion was feasibility, i.e. the test had to be able to be performed in sports halls with little specific equipment and by individuals with a broad range of fitness and health.

Previous field testing had shown that the CSE used with athletes (19) was too demanding for some SVMB exercise group participants. They reported joint pain while in positions of lateral and ventral muscle chains. The following adaptations were consequently made to the CSE for testing axSpA patients, resulting in an adapted version named the aCSE:

- The starting position for the ventral plane was changed from plank to quadruped position.
- Tests were performed statically instead of dynamically (24) to ensure a constant load (this modus reduced the risk of injury and was easier to standardize).
- Rods were used for standardization, i.e., the participant was asked to keep contact with the horizontal rod during the test (see supplement, Fig 1).

The time that the subject was able to remain in each of three positions was measured in seconds. Time recording stopped whenever the participant lost contact with the rod for the third time. The participant could get into the required position for a maximum of three attempts. Before the next position was measured, the participant rested for at least 30s. Detailed instructions on the performance of aCSE can be found in the Supplement.

#### Additional outcomes and assessments

#### Pain intensity

Pain intensity was measured using the Numeric Rating Scale (NRS), which is acknowledged as a reliable and valid measurement tool in clinical practice and research (25). The NRS measures pain on a scale from 0 to 10 (0 = no pain at all and 10 = worst pain ever). In this study, the tool was used to assess average pain immediately before and after the aCSE test batteries were performed.

#### The Bath Ankylosing Disease Activity Index (BASDAI)

AxSpA disease activity was measured using the BASDAI, which is a valid and reliable selfreporting questionnaire (26). The questionnaire consists of six items to determine pain in the peripheral joints and spine, fatigue, morning stiffness and pain sensitivity to touch (27). The BADAI results in a mean score of 0-6 points (0= no disease activity, 6= highest disease activity) (27).

#### The Bath Ankylosing Spondylitis Functional Index (BASFI)

Functional limitations experienced by axSpA patients during ten everyday tasks were evaluated using the BASFI, which is a valid and reliable self-reporting questionnaire (26).

The BASFI evaluation results in a mean score between 0 and 10 (0= no handicap and 10= highest possible degree of functional limitation).

#### Physical fitness questionnaire (FFB-Mot.)

Self-perceived motor performance ability during everyday tasks, with regard to cardiorespiratory fitness, strength, flexibility and coordination, was evaluated using The Physical Fitness Questionnaire (FFB-Mot., German: Fragebogen zur Erfassung des motorischen Funktionsstatus) (28). It has been shown that the FFB-Mot. is a valid instrument to determine physical fitness in a healthy adult population (29). The outcome of the 28 items on the questionnaire results in a total score ranging from 5 to 140; the strength questions include seven items for which the score can range from 0 to 30. For example, a total score of 35/140 for a 58 year old healthy man would indicate a rather poor overall physical fitness status, while a score of 100/140 would indicate a rather good physical fitness status (29). It has not yet been used as an assessment for evaluation, so no reliability data are available.

#### Weekly exercise

Questions on weekly hours of planned exercise, together with the focus of the exercise dimension (muscle strength, cardiovascular, neuromotor or flexibility), were asked by the PT, together with the patient characteristics (age, gender, year of diagnosis and disease duration).

#### Sample size and statistical analysis

Sample size was estimated based on the precision of the reliability estimate following Giraudeau and Mary (30). With a targeted width of the confidence interval (two margin of errors or four standard errors) of w = 0.3, 57 subjects were needed for (a conservative) ICC = 0.65. For larger ICCs, the required sample size would be smaller or the precision in the analysis would be larger.

All statistical analyses were performed using the R statistical software R version 3.5.3 (31).

#### Reliability

Generalizability theory (32) was used as the framework to estimate the reliability of the raters' time-keeping while assessing core muscle strength. To estimate the intra-rater reliability, a linear mixed model for "(strength endurance) Time"  $Y_i$  of Subject *i* nested in Rater *j* measured at time point *k* was fitted to the data:

$$Y_{i(j),k} = \mu + S(R)_i + R_j + T_k + RT_{ik} + \epsilon_{i,j,k}$$

With  $Y_{i(j), k}$  representing strength endurance time,  $\mu$  representing the global mean,  $S(R)_i$  corresponding to  $S_i + S(R)_{ij}$  which cannot be disentangled in a nested design and  $\varepsilon_{ijk}$  the independent and normal distributed errors.

The intra-rater reliability was calculated as the intra-class correlation coefficient

$$corr\left(Y_{i(j),k}, Y_{i(j),k'}\right) = \frac{\sigma_{S(R)}^2}{\sigma_{S(R)}^2 + \sigma_T^2 + \sigma_{RT}^2 + \sigma_\epsilon^2},$$

with the  $\sigma^2$  representing the corresponding variance components. Bootstrapped confidence intervals for the intra-class correlation coefficient (ICC) were computed. Values less than 0.2 demonstrate a slight reliability, values between 0.2 and 0.39 indicate a fair reliability, values between 0.4 and 0.59 describe moderate reliability, values between 0.6 and 0.79 indicate substantial reliability, and values greater than 0.80 indicate almost perfect reliability between measurements (33). The lower limit of the 95% confidence interval of the ICC to estimate the level of reliability was used (34).

#### Associations between the aCSE outcomes and disease-specific factors

To evaluate the relationship between aCSE outcomes and patient-reported strength and the disease-specific outcomes, the time measurements for aCSE ventral, lateral and dorsal were correlated with the FFB-Mot. subscales of strength, NRS-measured pain, BASDAI and BASFI, using the Pearson correlation coefficient. The size of a correlation coefficient can be interpreted as negligible ( $\rho < 0.3$ ), low ( $0.3 < \rho < 0.5$ ), moderate ( $0.5 < \rho < 0.7$ ), high ( $0.7 < \rho < 0.9$ ) and very high ( $\rho > 0.9$ ), with both positive/negative correlations (35). A priori, it was hypothesized that the following would provide evidence of an association between aCSE performance and disease-related factors: a) a positive Pearson correlation ( $r_s$ ) >0.3 between aCSE and FFB-Mot. subscale strength and/or self-reported hours exercise per week; and/or b) a negative correlation <0.5 between aCSE and pain (NRS) and/or disease activity (BASFI, BASDAI). The lower limits of 95% confidence intervals adjusted for multiple testing were used.

# Results

Thirteen PTs (28.8%) and 62 group participants (30.0%) provided informed consent and were included in the study; three of the latter could not participate at T1 (n = 2 due to acute influenza, n = 1 due to pain after the first measurement). Descriptive data for the study participants are shown in Table 1.

#### Intra-rater reliability of aCSE

A moderate to substantial intra-rater reliability was found for all three test positions that tested the strength of the ventral, lateral, and dorsal core muscle chains (Table 2).

#### Associations of aCSE outcomes with disease-specific factors

Data shown in Table 3 indicate no positive or negative associations between aCSE outcomes and any disease-specific factors measured. Thus, the a priori hypotheses were not confirmed.

Characteristics of axSpA patients	n=62
Gender, female, n (%)	29 (46)
Age, years	54.6 [11.2]
BASDAI (0-10)	3.5 [2]
BASFI (0-10)	1.8 [1.5]
FFB-Mot. total (0-140)	85.8 [18]
FFB subscale strength (0-35)	21.5 [5.4]
NRS pre aCSE at T0	2.2 [2.2]
NRS post aCSE at T0	1.6 [2.4]
Disease symptoms since, years	27.5 [12.9]
Disease duration since diagnosis, years	20.1 [13]
Self-reported training per week, hours	3.1 [2.9]
Characteristics of physiotherapists	n=13
Gender, female, n (%)	9 (69)
Age, years	46 [11]
Work experience, years	6.3 [3.5]
Number of participants per rater (median, range)	4 [2-9]

#### Table 1. Participants' characteristics

Data are mean (standard deviation), if not stated otherwise. BASDAI, Bath Ankylosing Disease Activity Index; BASFI, Bath Ankylosing Spondylitis Functional Index; FFB-Mot., Physical Fitness Questionnaire; NRS, Numeric Rating Scale; aCSE, adapted Core Strength Endurance test battery; T0 and T1, time point 1 (baseline) and 2, respectively; GET, group exercise therapy.

Table 2. Intra-rater reliability of the acous		
Test position	ICC	95% CI
aCSE ventral	0.54	0.35, 0.74
aCSE lateral	0.52	0.33, 0.70
aCSE dorsal	0.71	0.58, 0.86

#### Table 2. Intra-rater reliability of the aCSEs

aCSE, adapted Core Strength Endurance test battery; CI: Confidence Interval; ICC: Intraclass Correlation Coefficient.

Association	r	95%Cl*
aCSEv vs.FFB-Mot. strength	0.25	-0.19, 0.61
aCSEI vs. FFB-Mot. strength	0.34	-0.11, 0.67
aCSEd vs. FFB-Mot. strength	0.30	-0.15, 0.65
aCSEv vs. self-reported training hrs	-0.06	-0.44, 0.34
aCSEI vs. self-reported training hrs	-0.06	-0.44, 0.34
aCSEd vs. self-reported training hrs	-0.01	-0.32, 0.30
aCSEv vs. pain	-0.11	-0.50, 0.31
aCSEI vs. pain	-0.07	-0.44, 0.33
aCSEd vs. pain	0.06	-0.33, 0.43
aCSEv vs. BASFI	-0.03	-0.39, 0.33
aCSEI vs. BASFI	-0.16	-0.54, 0.27
aCSEd vs. BASFI	-0.35	-0.68, 0.10
aCSEv vs. BASDAI	-0.23	-0.59, 0.22
aCSEI vs. BASDAI	-0.26	-0.62, 0.19
aCSEd vs. BASDAI	0.03	-0.33, 0.37

#### Table 3. Evaluation of associations between aCSE and disease-specific factors

r=observed correlation coefficient, CI = confidence interval (\*uncertainty adjusted for multiple testing (Holm procedure)), BASDAI, Bath Ankylosing Disease Activity Index; BASFI, Bath Ankylosing Spondylitis Functional Index; FFB-Mot.strength, physical fitness questionnaire subscale strength; NRS, Numeric Rating Scale; aCSE, adapted Core Strength Endurance test battery; v,ventral; I, lateral; d, dorsal

# a reliable assessment for people with axSpA in a group setting.

Discussion

Possible associations between aCSE performance and disease-specific factors, according to the a priori hypotheses, were not confirmed. This is an interesting finding, suggesting that the aCSE can be performed by an individual with axSpA irrespective of their perceived strength performance, functional status and self-reported pain. This may be an aspect worthy of further investigation.

The aCSE values for the intra-rater reliability of the ventral, lateral, and dorsal planes show a moderate to substantial level of agreement. The results show that PTs are able to use the aCSE reliably, even with little experience of aCSE testing. Furthermore, nearly all participants were able to perform the aCSE, only one (1.6%) was unable to. In conclusion, the aCSE is

However, we were unable to confirm that pain did not influence aCSE performance. Pain was reported using NRS prior to and post testing, but no statistically significant negative relationship was found that would imply that more pain is related to less strength. Other than pain (36), the factors of motivation and effort are determinants of strength performance (37). Midgley and colleagues hypothesized that verbal encouragement could make a difference to test outcomes, through having a positive impact on intrinsic motivation and physical performance (38). The rating of perceived exertion together with consistent verbal encouragement should be incorporated into the future use of the aCSE. Consistent verbal encouragement could be usefully included in the GET assessment situation.

Due to the complexity of the core strength construct, there has been no gold standard for assessing core strength in the past, or core strength endurance in particular. Core strength is vital for maintaining an upright posture and is especially important for axSpA patients, who are affected by spinal inflammation and decreased spinal mobility. It is therefore appropriate to assess core strength endurance for axSpA patients. The physical position for assessing these individuals poses an additional challenge. It could be argued that to assess strength endurance of the dorsal muscle chain, the testing position described by Ito and colleague (39) (lying in prone position performing back extension) would be preferable to the testing position described by Biering Sorensen (40) (lying in prone position half of the body fixed on a treatment bench holding neutral position), because the Biering Sorensen occasionally activates more hip extensor muscles (24, 41). However, since the restricted flexibility of the spine is a major impediment in axSpA patients, the test procedure described by Biering Sorensen was considered to be more appropriate for covering the complete range of impairments from low to high. In future, it would be useful to develop a score for strength that encompasses all three planes, or to investigate whether there is a score from one plane that would serve as a sufficient proxy for all three planes. Ultimately, a database containing norm core strength values of individuals both with and without axSpA could ensure a meaningful interpretation of test results.

The original CSE is used as part of a sports performance test battery for athletes (19). With the aid of a reference table containing norm data of Swiss athletes, assessors judge whether core strength is sufficient or insufficient. Currently, no reference table including norm data of people with axSpA is available. This aspect should be taken into consideration when reporting test results to participants. However, until such a set of norm data is established, intra-individual comparisons might give an orientation.

Surprisingly, no relationship between aCSE outcomes and perceived strength performance was found, although there is evidence that people of all ages have a moderately accurate perception of their physical fitness (42). In contrast, other studies (43) provide evidence that people tend to be unrealistic in assessing their abilities, such as physical fitness or level of physical activity. A further reason for our finding could be that the aCSE measures core stability and not strength endurance. Core stability was not explicitly assessed in the patient-reported outcome measure. However, there is a consensus that core stability is dependent on the components of muscle strength and neuromotor control (8). Majewski-Schrage and colleagues (8) asked experts to provide assessment techniques that were specific to the components of core stability. The top three answers were: timed side bridge, Sorenson test, and timed prone bridge. All three of these are elements of the aCSE.

Low-cost was prioritized in our study setting. However, future projects could investigate the use of appropriate, objective, strength assessment devices, such as a hand-held dynamometer (44). This has been used previously to measure core strength indirectly (albeit maximal strength, not strength endurance) and has been found to be reliable and valid (7, 45). It might be less time-consuming to incorporate, e.g. easy to perform handgrip, instead of core strength, especially in a group setting.

This study has strengths and limitations. To our knowledge, this was the first study to evaluate the psychometric properties of a core strength assessment in axSpA patients. Movement assessment by visual observation and time-keeping appears to be feasible for PTs. However, it is a limitation of this study that reliability was tested by the clinically working usual GET PT leaders rather than by raters in a laboratory setting. Although the 2h assessment training was designed to standardize the test procedures and was appreciated by the PTs, their performance was likely to have been influenced by their motivation and understanding of the importance of performing the aCSE in a standardized way. PTs were not asked to perform the additional, disease-specific assessments, such as the Bath Ankylosing Mobility Index (BASMI), since the participants needed to receive specific instructions on how to hold the correct positions and to exert themselves. Additionally, the PTs emphasized that it was challenging to perform the assessments with each individual in the group during the usual GET sessions, in terms of time management and the planned exercise program. These factors (i.e., PTs' adherence to standardization, PTs' motivation, complexity of time-consuming tests) may have had an impact on the accuracy and reliability of the measurement outcomes. The reliability values could potentially be improved with improved test conditions.

# Conclusion

Regular fitness assessment on an individual basis is part of the SVMB's concept of quality and is recommended (3). The choice of assessment used in this study was influenced mainly by feasibility criteria, such as low-cost, easy-to-use for axSpA patients, and the ability for people with different health and fitness status to perform in both land-based and water-based GET settings. Appropriate assessments with good psychometric quality are necessary. In summary, the aCSE was found to be a feasible assessment instrument to measure core strength in axSpA patients reliably. It could be potentially combined with other assessments of aerobic fitness, flexibility and neuromotor performance to establish an inexpensive and practical assessment battery, covering all exercise dimensions, for axSpA patients. Future research should establish a reference norm data table including axSpA patients, to enable an appropriate interpretation of test results. In addition, a less timeconsuming alternative to implementing aCSE testing should be investigated, e.g. using only one representative plane instead of three separate planes.

#### Funding

Open access funding provided by ZHAW Zurich University of Applied Sciences. The measurement equipment was funded by physioswiss in 2017.

#### Authors' contributions

PB, AKR and KN designed the project. PB organised the data collection as part of his master's thesis. KN supervised the progress of the study. AM conducted the statistical analysis. All authors contributed significantly to the interpretation of results. AR drafted the manuscript. All authors provided comments on each draft and approved the final version.

#### Acknowledgements

We thank all participating physiotherapists and participants of the SVMB exercise groups for supporting the project with great enthusiasm and effort. Furthermore, our grateful thanks go to René Bräm, CEO SVMB, for promoting the project.

#### **Conflicts of Interest**

The authors declare no conflict of interest.

#### Compliance with ethical standards

Ethical approval and consent to participate The study obtained ethical approval from the Ethics committee of Canton Zurich (BASEC nr. 201700833). All participants (patients and PTs) gave written informed consent.

#### **Additional Files**

• Supplement A: Description of the core strength assessment battery

# References

- van der Heijde D, Ramiro S, Landewe R, Baraliakos X, Van den Bosch F, Sepriano A, et al. 2016 update of the ASAS-EULAR management recommendations for axial spondyloarthritis. Ann Rheum Dis. 2017;76(6):978-91.
- Mangone M, Paoloni M, Procopio S, Venditto T, Zucchi B, Santilli V, et al. Sagittal spinal alignment in patients with ankylosing spondylitis by rasterstereographic back shape analysis: an observational retrospective study. Eur J Phys Rehabil Med. 2020;56(2):191-6.
- Rausch Osthoff AK, Niedermann K, Braun J, Adams J, Brodin N, Dagfinrud H, et al. 2018 EULAR recommendations for physical activity in people with inflammatory arthritis and osteoarthritis. Ann Rheum Dis. 2018;77(9):1251-60.
- Fayh A, Brodt GA, Souza C, Loss JF. Pilates instruction affects stability and muscle recruitment during the long stretch exercise. J Bodyw Mov Ther. 2018;22(2):471-5.
- Borghuis J, Hof AL, Lemmink KA. The importance of sensory-motor control in providing core stability: implications for measurement and training. Sports Med. 2008;38(11):893-916.
- Hibbs AE, Thompson KG, French D, Wrigley A, Spears I. Optimizing performance by improving core stability and core strength. Sports Med. 2008;38(12):995-1008.
- Huyghe M, Sercu H, Vervaeke O. Evaluating Core Stability: Reliability and validity of hand-held dynamometry in measuring core strength: Ghent University 2016.
- Majewski-Schrage T, Evans TA, Ragan B. Development of a core-stability model: a delphi approach. J Sport Rehabil. 2014;23(2):95-106.
- Zemkova E, Poor O, Jelen M. Between-side differences in trunk rotational power in athletes trained in asymmetric sports. J Back Musculoskelet Rehabil. 2019;32(4):529-37.
- 10. Sapega AA, Drillings G. The definition and assessment of muscular power. J Orthop Sports Phys Ther. 1983;5(1):7-9.
- 11. Strand SL, Hjelm J, Shoepe TC, Fajardo MA. Norms for an isometric muscle endurance test. J Hum Kinet. 2014;40:93-102.
- Rausch Osthoff AK, Juhl CB, Knittle K, Dagfinrud H, Hurkmans E, Braun J, et al. Effects of exercise and physical activity promotion: meta-analysis informing the 2018 EULAR recommendations for physical activity in people with rheumatoid arthritis, spondyloarthritis and hip/knee osteoarthritis. RMD open. 2018.
- Kim K-J. Effects of Core Muscle Strengthening Training on Flexibility, Muscular Strength and Driver Shot Performance in Female Professional Golfers. International Journal of Applied Sports Sciences. 2010;22(1).
- 14. Fang HY, Cai WZ, Pan YH, Wu DC, Liang LM. Six-month home-based exercise and supervised training in patients with ankylosing spondylitis. Int J Clin Exp Med. 2016;9(3):6635-41.
- Dagfinrud H, Halvorsen S, Vøllestad NK, Niedermann K, Kvien TK, Hagen KB. Exercise programs in trials for patients with ankylosing spondylitis: Do they really have the potential for effectiveness? Arthritis Care & Research. 2011;63(4):597-603.
- Rausch Osthoff AK, van der Giesen F, Meichtry A, Walker B, Van Gaalen F, Goekoop-Ruiterman YPM, et al. The perspective of people with axial spondyloarthritis regarding physiotherapy: room for the implementation of a more active approach. Rheumatology Advances in PRactice. 2019.
- 17. Van den Berg R, Baraliakos X, Braun J, van der Heijde D. First update of the current evidence for the management of ankylosing spondylitis with non-pharmacological treatment and non-biologic drugs: a systematic literature review for the ASAS/EULAR management recommendations in ankylosing spondylitis. Rheumatology. 2012;51(8):1388-96.
- Sveaas SH, Bilberg A, Berg IJ, Provan SA, Rollefstad S, Semb AG, et al. High intensity exercise for 3 months reduces disease activity in axial spondyloarthritis (axSpA): a multicentre randomised trial of 100 patients. Br J Sports Med. 2019.
- Sveaas SH, Bilberg A, Berg IJ, Provan SA, Rollefstad S, Semb AG et al. High intensity exercise for 3 months reduces disease activity in axial spondyloarthritis (axSpA): a multicentre randomised trial of 100 patients. Br J Sports Med 2020, 54(5): 292-297
- AbilityLab SR. Rehabilitation Measures: Beiring Sorensen Test https://www.sralab.org/rehabilitation-measures/beiringsorensen-test 2015 [11.03.2020].
- 21. Kottner J, Audige L, Brorson S, Donner A, Gajewski BJ, Hrobjartsson A, et al. Guidelines for Reporting Reliability and Agreement Studies (GRRAS) were proposed. Int J Nurs Stud. 2011;48(6):661-71.
- 22. Committee NYHAC, Association NYH. Nomenclature and criteria for diagnosis of diseases of the heart and great vessels: Little, Brown Medical Division; 1979.
- 23. Kiltz U, Rudwaleit M, Sieper J, Braun J. Evidenzbasierte Empfehlung zur Diagnostik und Therapie der axialen Spondyloarthritis. Zeitschrift für Rheumatologie. 2016:1-6.
- Latimer J, Maher CG, Refshauge K, Colaco I. The reliability and validity of the Biering-Sorensen test in asymptomatic subjects and subjects reporting current or previous nonspecific low back pain. Spine. 1999; 24(20):2085-9.
- 25. Williamson A, Hoggart B. Pain: a review of three commonly used pain rating scales. Journal of Clinical Nursing. 2005;14(7):798-804.
- 26. Moncur C. Ankylosing spondylitis measures: The Ankylosing Spondylitis Quality of Life (ASQOL) Scale, Bath Ankylosing Spondylitis Disease Activity Index (BASDAI), Bath Ankylosing Spondylitis Functional Index (BASFI), Bath Ankylosing Spondylitis Global Score (BAS-G), Bath Ankylosing Spondylitis Metrology Index (BASMI), Dougados Functional Index (DFI), Health Assessment Questionnaire for the Spondyloarthropathies (HAQ-S), and Revised Leeds Disability

Questionnaire (RLDQ). Arthritis Care & Research. 2003;49(S5):S197-S209.

- Brandt J, Westhoff G, Rudwaleit M, Listing J, Zink A, Braun J, et al. Validierung einer deutschen Version des Fragebogens BASDAI zur Messung der Krankheitsaktivität bei ankylosierender Spondylitis. Zeitschrift für Rheumatologie. 2003;62(3):264-73.
- 28. Bös K. Handbuch Motorische Tests, 2.: Hogrefe; 2001.
- 29. Bos K, Abel T, Woll A, Niemann S, Tittlbach S, Schott N. The Physical Fitness Questionnaire (FFB-Mot). Diagnostica. 2002;48(2):101-11.
- Giraudeau B, Mary JY. Planning a reproducibility study: how many subjects and how many replicates per subject for an expected width of the 95 per cent confidence interval of the intraclass correlation coefficient. Stat Med. 2001;20(21):3205-14.
- 31. R CT. Language and Environment for Statistical Computing. In: Computing RFfS, editor. Vienna, Austria 2018.
- 32. Brennan RL. Generalizability theory. New York2001.
- 33. Landis JR, Koch GG. The measurement of observer agreement for categorical data. Biometrics. 1977;33(1):159-74.
- Koo TK, Li MY. A Guideline of Selecting and Reporting Intraclass Correlation Coefficients for Reliability Research. J Chiropr Med. 2016;15(2):155-63.
- 35. HInkle DE, Wiersma W, Jurs SG. Applied Statistics for the Behavioral Sciences. Edition t, editor. Boston: Houghtin Mifflin; 2003.
- Riddle DL, Stratford PW. Impact of pain reported during isometric quadriceps muscle strength testing in people with knee pain: data from the osteoarthritis initiative. Phys Ther. 2011;91(10):1478-89.
- Marcora SM, Staiano W. The limit to exercise tolerance in humans: mind over muscle? Eur J Appl Physiol. 2010;109(4):763-70.
- Midgley AW, Marchant DC, Levy AR. A call to action towards an evidence-based approach to using verbal encouragement during maximal exercise testing. Clin Physiol Imaging. 2018;38:547-53.
- Ito T, Shirado O, Suzuki H, Takahashi M, Kaneda K, Strax TE. Lumbar trunk muscle endurance testing: an inexpensive alternative to a machine for evaluation. Archives of physical medicine and rehabilitation. 1996;77(1):75-9.
- 40. Biering-Sorensen F. Physical measurements as risk indicators for low-back trouble over a one-year period. Spine (Phila Pa 1976). 1984;9(2):106-19.
- Latikka P, Battié MC, Videman T, Gibbons LE. Correlations of isokinetic and psychophysical back lift and static back extensor endurance tests in men. Clinical Biomechanics.10(6):325-30.
- 42. Germain JL, Hausenblas HA. The relationship between perceived and actual physical fitness: a meta-analysis Journal of applied sport psychology. 2006;18(4).
- Yu CA, Rouse PC, Veldhuijzen Van Zanten JJ, Ntoumanis N, Kitas GD, Duda JL, et al. Subjective and objective levels of physical activity and their association with cardiorespiratory fitness in rheumatoid arthritis patients. Arthritis Res Ther. 2015;17:59.
- 44. Stark T, Walker B, Phillips JK, Fejer R, Beck R. Hand-held dynamometry correlation with the gold standard isokinetic dynamometry: a systematic review. PM R. 2011;3(5):472-9.
- 45. Harding AT, Weeks BK, Horan SA, Little A, Watson SL, Beck BR. Validity and test-retest reliability of a novel simple back extensor muscle strength test. SAGE Open Med. 2017; 5:2050312116688842.

# 7

Lessons learned from a pilot implementation of physical activity recommendations in axial Spondyloarthritis exercise group therapy

Anne-Kathrin Rausch Osthoff, Theodora P M Vliet Vlieland, André Meichtry, Leti van Bodegom-Vos, Beatrice Topalidis, Stefan Büchi, Irina Nast, Adrian Ciurea, Karin Niedermann

> BMC Rheumatology (2022) 6:12 doi: 10.1186/s41927-021-00233-z

Oral presentation and poster presentation at EULAR congress in Madrid (June 2019)

## Abstract

#### Background

The Ankylosing Spondylitis Association of Switzerland (SVMB) aimed to implement physical activity recommendations (PAR) within their exercise groups (EGs). The PAR promote exercise in all fitness dimensions at the correct dose. To implement the PAR within EGs, they were translated into a new EG concept with five key activities: (a) training for supervising physiotherapists (PTs), (b) correctly dosed exercises in all fitness dimensions, (c) exercise counselling, (d) bi-annual fitness assessments, and (e) individual exercise training, in addition to EG. All these activities were realized in close coordination with SVMB management.

#### Objectives

To analyse the implementation success by evaluating adherence/fidelity, feasibility, and satisfaction at the patient, PTs, and organisational level.

#### Methods

The five key activities of the new EG concept were developed, executed, and assessed after 6 months. The primary outcomes for implementation success were adherence of patients to the recommended exercise behaviour, self-reported by electronic diary; fidelity of PTs to the new concept, self-reported by diary; SVMB organisational changes. Secondary outcomes were feasibility and satisfaction with the new EG concept at all three levels. The tertiary outcome, to evaluate the effectiveness of PAR, was patient fitness, assessed through fitness assessments.

#### Results

30 patients with axSpA (ten women, mean age 58  $\pm$ 9 years) and four PTs (three women, mean age 46  $\pm$ 9 years) participated. The patients' self-reporting of adherence to the PAR was insufficient (43%), possibly due to technical problems with the electronic dairy. The PTs' fidelity to the new EG concept was satisfactory. On all levels, the new concept was generally perceived as feasible and useful for supporting personalised exercise. The frequency of exercise counselling and the fitness assessments was found by patients and PTs to be too high and rigid. Patients' cardiorespiratory fitness [ES 1.21 (95%CI 0.59,1.89)] and core strength [ES 0.61 (95%CI 0.18, 1.069] improved over the 6 months.

#### Conclusions

The pilot implementation of PAR showed acceptance and satisfaction to be sufficient, thus confirming the need for evidence-based EGs, provided by a patient organisation in order to support active PA behaviour. However, adaptations are necessary to increase its feasibility for nationwide implementation.

#### **Trial Registration**

SNCTP000002880 (registered 31 May 2018)

# Introduction

Patients with axial spondyloarthritis (axSpA) have an increased risk of cardiovascular diseases (1). In addition, evidence shows that axSpA has an impact on flexibility (2), balance (3), muscle strength (4), and cardio-respiratory capacity (5).

Exercise and effective drug treatment are recommended as the cornerstones of disease management (6, 7). Indeed, the promotion of physical activity (PA) and specific exercise (cardiovascular, muscle strength, flexibility, and neuromotor exercise) should be an integral part of standard care throughout the course of disease (8). Regarding cardiovascular exercise, evidence underpins that the appropriate intensity and frequency are relevant to reduce the risk of cardiovascular diseases in healthy people effectively; this might be valid for people with rheumatic and musculoskeletal diseases (RMDs) as well (9, 10). Furthermore, evidence shows that exercise according to public health recommendations for health-enhancing PA is effective, safe, and feasible in people with RMDs (11). Consequently, PA recommendations should be implemented, establishing a standard for PA and exercise promotion and education of health professionals for effectively delivering PA promotion and exercise accross Europe.

The Ankylosing Spondylitis Association of Switzerland (Schweizerische Vereinigung Morbus Bechterew, SVMB) decided to implement the PA recommendations within their exercise groups. It was further decided to involve the SVMB members who currently do not participate in exercise groups at a later stage of time.

Previously, the groups, each consisting of between 5 and 25 patients with axSpA, met weekly to conduct 45-60 min of spinal flexibility and strength exercises, mostly land-based, few water-based, or combined, under the supervision of a physiotherapist (PT). Health insurers refunded the costs for the attendance of these exercise groups.

The overall aim of this implementation project was to establish the PA recommendations for patients with axSpA participating in exercise groups to perform exercises regularly and adequately dosed (12). The PA recommendations were translated into a new SVMB exercise group concept, named "moveSVMB" (an acronym for *iMplementation Of physical actiVity rEcommendations in SVMB*). It incorporated the following key implementation activities: (a) 2 days of training for supervising PTs, (b) exercises in all fitness dimensions, according to the exercise guidelines (13), (c) quarterly individual exercise counselling with a supervising PT to support more individual exercise training, (d) bi-annual fitness-assessments, and e) individual exercise training in addition to group exercise. All implementation activities were closely coordinated by SVMB management and study staff.

It is well known that the implementation of innovations in clinical routine is a great challenge (14, 15), requiring a planned and systematic approach with clear strategies. Knowledge

does not necessarily lead to changed behaviour (16-18). In this project, the implementation process was determined following the cyclical approach of Grol & Wensing' implementation of change model (19) and included the following steps (and publications): (A) analysis of the current clinical practice and settings (as briefly outlined above, (20)); (B) exploration of facilitators and barriers towards PA and exercise from the stakeholders' perspective, i.e. patients with axSpA, PTs and rheumatologists (21), including the patients' perspective on technology-based exercise (yet unpublished data); (C) development and application of a tailored implementation strategy; (D) pilot testing and adjustment of the applied strategy, and E) nationwide implementation.

The aim of the present pilot implementation study was to perform steps (C) and (D) and evaluate the adherence to recommended exercise behaviour on the levels of patients with axSpA, the fidelity to the new concept on the level of PTs, and the feasibility and satisfaction regarding the new concept, on the levels of patients, PTs and organisation. An additional objective was to investigate the effectiveness of PA recommendations at the level of patients. Furthermore, the long-term implementation fidelity was evaluated to gain lessons learned and inform the future nationwide implementation (step E).

# Methods

The Standards for Reporting Implementation Studies (STaRI) checklist (22) were followed to report the implementation steps (additional file 1), as well as the recommendations for reporting implementation strategies (23). The study was registered (SNCTP000002880) and approved by the ethical board of Canton Zurich (BASEC-Nr. 2018-00145). PTs and patients gave written informed consent.

#### Study design

The pilot implementation study had a hybrid type 3 design (24). It was executed from June (T0) to November 2018 (T1), with evaluations at T1 and at T2 (T2 after 1 year). An implementation strategy, nested implementation activities and evaluation procedures for each of the relevant target groups were developed, i.e. patients with axSpA, PTs, and SVMB management. For details refer to Table 1 and "implementation strategy" section below. The outcomes and assessments used to evaluate the implementation strategy and the effectiveness of the PA recommendations on the level of patients with axSpA, are described in Table 2 and below in "Evaluations" section.

#### Implementation strategy

#### Targeting patients and PTs: the new exercise group concept "moveSVMB"

A new EG concept encompassing the following key elements was introduced:

a. Training of supervising PTs

The supervising PTs attend a mandatory 2 days of training to learn how to apply the new concept within their exercise groups. The training consists of: gaining knowledge and awareness of the general and EULAR PA recommendations and improving exercise counselling and motivational support skills. The monitoring of quality is performed by diary and peer-involvement through regular exchange between the PTs and study staff. Besides communication and psychological theory, PTs learn the practical use of the instrument PRISM (Pictorial Representation of Illness and Self Measure, described in Additional file 2) (25) and, behavioural change techniques, such as goal setting and strengthening self-efficacy using role plays. Furthermore, study staff and PTs discuss the study processes. During the pilot intervention a refresher event and regular support is provided, e.g. "help desk" for current issues.

b. Correctly dosed exercises in all fitness dimensions

PTs instruct and perform exercises in all fitness dimensions during the exercise group sessions. According to the current knowledge of exercise guidelines, the dosage of exercises follows the FITT (Frequency, Intensity, Type, Time) principles (8, 13). Patients are additionally informed about the importance and elements of adequate exercise through information provided by the SVMB (website, SVMB journal). Both activities enable patients to perform exercises alone.

Individual and group exercise counselling
PTs are asked to perform three individual exercise counselling sessions with each

patient and at least three group interventions to improve patients PA behaviour.

An individual counselling session lasts about 30 min. PTs are encouraged to use an empathic, supportive, yet directive counselling style, according to the principles of Motivational Interviewing (26). PTs are asked to start each individual session by applying the PRISM, which is a visual-tactile instrument that fosters communication, collaboration and mutual confidence (25, 27). PRISM can also be used to get to know a person better, by defining dimensions relevant to his/her life and, thus, enhancing intrinsic motivation and self-management activities (28). After using PRISM and, based on the assessments of the patient's fitness and his/her personal needs, the PT discusses with the patient how he/she can integrate more exercise into his/her everyday life. The fitness assessments help to identify the exercise dimension(s) with potential for improvement. In this regard, counselling may focus on raising consciousness, reinforcing the patients' PA behaviour, avoidance of relapse, or maintenance strategies. For example, some patients may need support in increasing the active time spent on light activities and reducing sedentary behaviour, while others may need a structured exercise plan to participate in a half-marathon. With each patient an individual goal and action plan is developed in a shared decision process.

Group counselling interventions may vary in time (5-45 min) and can be designed differently, according to the needs and interests of the group. Examples are, (a) theoretical and practical training on heart rate self-monitoring during aerobic exercise, and (b) group discussion on how to cope with barriers.

#### d. Fitness assessments

During a regular exercise group session, a team of one senior (researcher) and several junior assessors (PT students) performs the fitness assessments with each participant, described as tertiary outcomes in Table 2 and Additional file 2. After the physical assessments, each patient is instructed to fill out six questionnaires at home. Furthermore, an exercise diary is explained and highly recommended, enabling feedback and reporting. Within 1 week, the individual assessment results are sent to the supervising PT for exercise counselling.

#### e. Individual, unsupervised exercise training

To fulfil the recommended amount of PA, one supervised exercise group a week is insufficient. The elements mentioned above should enable group members to engage in additional, unsupervised exercise. Based on the fitness assessments and personal preferences, the patient and PT agree on a training goal in one, or several, of the exercise dimensions and create an action plan. During the following counselling session, progress is evaluated and the goal reviewed.

Physiotherap	iists							
Element(s) of EG concept	Activity	Actors	Actions	Targets of change	Temporality (frequency)	Dose	Implementation outcomes affects; assessment	Justification
ଅ	Information	Study staff, self-study	Learn/recall characteristics of effectively dosed PA and exercise, purpose and methods of PA and exercise in line with actual guidelines/EULAR recommendations.	Knowledge and awareness	Workshop before implementation in EGs	Zh	Knowledge, feeling of confidence in dealing with topic and promoting exercises; interview	Consciousness raising, Self-Reevaluation (both Transtheoretical model)
ଜ	Education	PRISM expert	Aim, methodology and practical application of PRISM [27] in 1:1 counselling situation; communication skills to motivate and support groups, manage difficult situations, and encourage participants to perform (more) exercise.	Skils	Workshop before implementation in EGs	කි	Use of and adherence to the counselling elements (PRISM, MI); diary	Guided practice (Social Cognitive Theory)
a)	Education	Study staff, self-study	Learned/recalled how to elaborate a progressive exercise plan	Skills	Workshop before implementation in EGs	÷	Knowledge and development of exercise plans; diary	Guided practice (Social Cognitive Theory)
) Ĵ	Support	PT peer	PTs know each other and ask each other for support, exchange experiences	Peer-factor	.p.u	'n	Feeling supported; interview	Enhance network linkages (Theory of social networks)
ତ	Supervision	Study staff	A «help desk» is provided by a study staff member to support in current issues. Regular refresher/ supervision events with a PRISM expert are provided.	Motivation to ensure quality of intervention	As needed	As needed	Beeing motivated ; interview	Feedback (Goal-Setting Theory, Social Cognitive Theory)

Table 1. Elements of implementation strategy and implementation intervention activities according to Proctor (23)

	Justification	Consciousness -raising, Self-Reevaluation (both Transtheo-retical model)	Modelling (social cognitive theory)	Individualisation (Transtheo-retical model) Tailoring (Transtheoretical model) Planning coping responses (Flelapse Prevention Theory) Goal Setting (Goal Setting Theory) Set graded tasks (Social Cognitive Theory)
	Implementation outcomes affects; assessment	Awareness of how to perform effective exercises for each exercise diary <sup>1</sup> exercise diary <sup>1</sup>	Feeling supported; survey	Performance in assessments and (frequency and dose) training in each exercise diany <sup>1</sup> diany <sup>1</sup>
	Dose	Ca. 5min as needed Ca.30min reading time	n.d.	45min. 30min.
	Temporality (frequency)	Weekly (PT) As needed (website) 4x/year (SVMB journal)	n.d.	Bi-annual Assessments Quarterly counselling sessions sessions
	Targets of change	Knowledge and awareness of PAR	Empower peer-factor	Setting goals, action planning, evaluation goal attainment
	Actions	PTs distribute handouts to patients and explain the four exercise appropriately. Questions are discussed in the EGs. SVMB supports the knowledge transfer by proving information on website and SVMB journals.	Suitable peers, who successfully perform regular exercise, act as role models to support peers from their EG	Assessments help to identify active and non-active participants and which of the exercise dimensions have the potential for improvement. An individual goal and action plan is developed during the 1:1 sessions in a shared decision manner.
	Actors	PTs, SVMB	Peer patients	PTs
axSpA	Strategy	Information	Support	Individualisation
Patients with	Element(s) of EG concept	(þ) ; e)	b); e)	c); c)

Table 1. Continued

	Justification	Feedback (Goal-Setting Theory, Social Cognitive Theory) Reinforcement (Social Cognitive Theory) Goal Setting (Goal Setting Theory) Set graded tasks (Social Cognitive Theory)	Implementation intensions (Theories of Goal Directed behaviour) Feedback (Goal-Setting Theory, Social Cognitive Theory) Reinforcement (Social Cognitive Theory)
	Implementation outcomes affects; assessment	Goal and action plan; exercise diary, activity tracker Changes according to PRISM in the patients perceived burden of disease and importance of PA; PRISM	Exercise is part of the daily routine, exercise diary, activity tracker diary, activity tracker
	Dose	30min.	Ca. 2-5 mins.
	Temporality (frequency)	5 sessions within 18months (3 sessions within first six months/ pilot phase, 2 booster sessions)	Daily reporting
	Targets of change	Motivation and coping, self-efficacy	Organisation in daily life / Routine
	Actions	PTs and each patient have five 1:1 exercise counselling sessions, to define exercise goals, identify and deal with barriers and utilisation of facilitators, supporting self- efficacy, relapse prevention. Furthermore, regular counselling sessions within the whole group are performed.	Patients keep a diary about their exercise-activities, polar watches are provided for self-monitoring of the heart rate (not for data analysis)
	Actors	PTs	Patients
tinued	Strategy	exercise exercise counselling	Self-monitoring
Table 1. Con	Element(s) of EG concept	ΰ	ō

	L
Q	L
Ð	L
5	L
	L
₽	V
Ξ.	ľ
0	l
0	l
	L
-	L
Φ	l
	- 1

Organisation							
Strategy	Actors	Actions	Targets of change	Temporality (frequency)	Dose	Implementation outcomes affects; assessment	Justification
Regular information exchange and discussion	Study staff	Study staff explains elements of the EG concept, which are discussed with SVMB management.	Awareness, decision to change routine care	Every three months	30min	Accept and implement new EG concept as usual care; vote of SVMB management	Sense-making (Organizational Development Theory)
Acquisition of third-party funds Support negotiations with insurers	CEO, SVMB project leader	Responsible members of SVMB management establish personal and financial resources to continue all elements of the EG concept after the pilot phase.	Embedding of new care in organi-sation	D.d	n.d.	Establish finances and structures to maintain concept; staff positions, working hours/ week	Technical Assistance (Organizational Development Theory)
Integration of SVMB- stakeholder in adaptation process	CEO, SVMB project leader	Study staff and SVMB management were in continuous exchange to ensure the feasibility of the new EG concept and satisfaction with its integration into routine care.	Acceptance EG concept	Every three months	30min	Feasibility and satisfaction; continous contact	Increasing stakeholder influences (Stakeholder Theory)
n.d., not defined; SVMB, Ankylosing Spon	dylitis Associati	on of Switzerland; PT, physiotherapist; E0	3= exercise group;	PRISM, Pictorial Respre	esentation o	of Illness and Self Mea	sure; <sup>1</sup> paper or free

i. app « Trainungstagebuch» by Johannes Tscholl

7

Table 1. Continued

#### Targeting the SVMB

Regular meetings are held between SVMB management and study staff to ensure that the implementation activities align with the structures and processes of the SVMB. In addition, to increase the acceptance and feasibility of "moveSVMB", a significant effort is put into establishing a long-term financial strategy and personnel resources.

#### **Evaluations**

#### Evaluation at T1

The primary outcome for the evaluation at the patient level was adherence to the recommended exercise behaviour, assessed by the number of reported training sessions per exercise dimension in the electronic exercise diary for over 6 months. The secondary outcomes at T1 were feasibility and satisfaction with the new exercise group concept, rated on a 0-10 numeric rating scale (0 = not satisfied, 10 = very satisfied; if > 50% of patients rate > 5, the concept was evaluated as "satisfying") and changes in the perceived burden of disease or importance of PA, measured with PRISM. The tertiary outcome to evaluate the effectiveness of the individualised PA recommendations was the measured changes in the fitness levels of patients (T0 - T1), assessed by fitness parameters (Table 2).

The primary outcome for the evaluation at PT level was fidelity to the program design concept, assessed using a diary. Fidelity was rated as sufficient when 75% of PTs fulfilled the program. The secondary outcomes were treatment quality, feasibility and satisfaction, evaluated through semi-structured interviews.

The evaluation at the SVMB level focused on acceptance of the new exercise group concept and its establishment within organisational structures. In addition, regular progress meetings between study staff researchers and SVMB management were used to appraise the feasibility and effectiveness of the implementation interventions.

#### Evaluation at T2

Maintenance was monitored at T2 (1 year after T1) by assessing the number of patients and supervising PTs still adhering to the new group concept.

#### Adaptations of "moveSVMB"

After 6 months, the findings of evaluations on all three levels were considered while deciding on adaptations of the program concept of "moveSVMB". Both SVMB management and study staff agreed on findings that negatively impacted the feasibility. Accordingly, they developed adaptations needed for a successful nationwide implementation of the new exercise group concept.

#### Study procedures

The four pilot exercise groups, comprising the supervising PTs and 5-12 patients with axSpA, from the German-speaking region of Switzerland were asked by the SVMB management

to volunteer. Criteria were a location close to the study center and the motivation and availability of PTs and group members. The partipicant PTs and patients were informed face-to-face about the new group concept "moveSVMB" well in advance. Group members declining to be measured or not providing written informed consent were offered the usual exercise group intervention.

Eligibility criteria for patients with axSpA were  $\geq$  18 years old, a member of a SVMB exercise group, and provision of written informed consent. All patients were screened for cardiovascular risk (29). Patients identified as at increased risk required confirmation from a PT or medical doctor that a higher level of exercise was not contraindicated. Criteria for exclusion were medically relevant contradictions to exercise (8). Eligibility criteria for PTs were being a superviser PT of an existing SVMB group, willingness to follow the implementation protocol of the new exercise group concept (including a 2-day educational workshop), and provision of written informed consent.

At the beginning of the pilot study, some organisational requirements were defined. A cooperation with the BSc in physiotherapy programme at the ZHAW was established to recruit second year PT students to perform the fitness assessments in exercise groups under the supervision of a senior researcher. Additionally, third-party funding was secured prior to the start of negotiations with insurers concerning reimbursement. Throughout the pilot phase, the SVMB provided personnel (project manager, c.a. 17 h/week (40%), additional IT support), and was prepared to increase resources after evaluation of the pilot phase.

#### Statistics

Demographic, disease-specific, self-reported (diary or patient-reported outcome measures PROMS) and assessment data were presented as mean and standard deviations, or median and associated range for continuous data or as frequencies (percentages) for categorical variables. During the fitness tests, the maximal time of one leg stance was limited to 60 s, this being the length of time indicating that the individual has no balance problem. For each exercise dimension, a linear mixed model with random intercept was fitted to the data with the within-subject variables of time and measurement condition and the between-subject variables of age, disease duration, disease activity (Bath Ankylosing Spondylitis Mobility Index BASMI), and PA level (metabolic equivalent of tasks METs) as explanatory variables. PRISM data (in cm) were correlated with the Bath Ankylosing Spondylitis Disease Activity Index (BASDAI) and weekly METs using spearman rank correlation. Paired t-tests and Wilcoxon tests, with corresponding parametric and nonparametric effect sizes (ES), were computed to analyse within-group changes. All statistical analyses were performed using the R statistical software, R version 3.5.2 (R Foundation for Statistical Computing, Vienna, Austria).

		Assessment (time of measure)	Quality concept (T1)	Funding, status of negotiations (T1, T2)	Working hours/week (T1, T2)	Interview of CEO and staff (T1)						
	Organisation	Outcome	<i>Integration</i> within organisation	Finances	Personel resources	Satisfaction						
		Assessment (time of measure)	Diary (continuous reporting)	Interview (T1)		Interview (T1)						
	Physiotherapists	Outcome	Fidelity to new EG concept	Feasibility		Satisfaction						
0.0 V	SpA	Assessment (time of measure)	Exercise diary (continous reporting)	Survey (T1)	PRISM (every counselling-session)	Survey (T1)	Chester Step Test [61, 62] Swiss Olympic Core Test Battery [63] adapted for patients with axSpA [64] Single leg stance [65] Bath Ankylosing Spondylitis Mobility Index/BASMI [66]	International Physical Activity Questionnaire/IPAQ [67]	Bath Ankylosing Spondylitis Disease Activity Index/BASDAI [66] Bath Ankylosing Spondylitis Global Score/BAS-G [68]	Assessment of SpondyloArthritis international Society/ASAS-HI [69]	Euro-Quol Questionnaire [70]	TO /other to method
;	Patients with ax	ŋ		Feasibility	Burden of disease, importance of PA	Satisfaction (feeling supported)	Physical fitness in all exercise dimensions	General physical activity	Disease activity	Health status	Disease related quality of life	Accellant T4 /office dist
		Outcome	Primary: Adherence to recommended exercise behaviour	Secondary: feasibility & satisfaction			Tertiary: effectiveness of recommended PA behaviour					OL CONTROLOGIC CONTROL TO
		to noitsulava	irategy	ation s	tnəməl	dwj	entions	ivretni 1	ectiveness o	#3		Line of the

Table 2. Evaluation of the implementation strategy and the effectiveness of the recommended PA behaviour at the level of patients with axSpA.

. . .

The or eneroweness measures was to (pasenne), it (arter sx monus), it (arter it monus) The tertiary outcomes are the regular fitness assessments in the exercise groups, which were used to evaluate the effectiveness of interventions. Assessments are described in Supplement A. PRISM, Pictorial Respresentation of Illness and Self Measure; PA, physical activity; CEO, chief executive officer.

### Results

#### Demographic and clinical data of exercise group members

Of the 43 patient members of the four pilot groups, 30 (70%) signed informed consent and participated in the baseline assessments. Reasons for non- participation were reported by ten members of the groups: timing of assessments due to holidays (3), lack of time (2), age or disease activity (2), no interest (2), no axSpA (1). The characteristics of the four pilot groups are described in *Table 3*.

During the 6-month pilot phase, two patients dropped out due to specified reasons: "I do not need exercise counselling", and "it is too much pressure for me". In addition, two patients did not participate in the assessments after 6 months due to an influenza.

#### Results at the level of patients with axSpA

#### Adherence to recommended exercise behaviour

Thirteen patients (43%) submitted an exercise diary each week to assess adherence to PA recommendations. None of these thirteen people fulfilled the PA recommendations in all four exercise dimensions. Four patients fulfilled the recommended dose of weekly cardiovascular exercises, one the recommendation of one additional exercise dimension, and two patients the recommendations of two additional exercise dimensions.

#### Feasibility and satisfaction

The survey's response rate was 90% (27/30). Since two were returned empty, 25 were used for analysis. Of these, 15 (62%) patients were satisfied with the concept "moveSVMB" (mean satisfaction 6.7 ± 2.2) and 20 (80%) with the exercise counselling (mean satisfaction 7.7 ± 2.9). Patients rated and commented the use of PRISM contrary (11 "not useful" vs 12 "useful"); appreciating ("I understand the relation between axSpA and my life" (ID15)) or criticising ("this is coffee cup reading to me" (ID05)). Twenty-two (88%) patients defined a training goal, 17 (68%) developed an action plan and 15 (60%) achieved their training goal. Sixteen (64%) patients used the exercise diary, five used app-based reporting. Reasons for non-reporting were: the daily effort, technical problems with the app, and problems with allocating activities to exercise dimensions correctly (e.g. is the SVMB exercise group strength or neuromotor exercising?). Personal adaptations were made to individual exercise goals (e.g. in terms of frequency or intensity of training) and to individual exercise diaries (e.g. notes in the personal agenda instead of the proposed sheets). A minority (n = 7, 28%) reported using a technical device (such as a heart rate tracker) for cardiovascular exercise. Seventeen (68%) were satisfied with the fitness assessment (mean 7.5±1.8). Patients did not use the project name "moveSVMB" but suggested names such as "Bechterew-Fit" or "Bechterew-Gymnastics".

Table 3. Characteristics of axSpA patients and	d supervising PTs a	t baseline			
Characteristics	Total	Group 1	Group 2	Group 3	Group 4
Patients with axSpA					
Total number per group (n)	43	8	11	8	16
Number participating in "moveSVMB" (n)	30	c	9	7	14
Gender, women, n (%)	10 (33.3)	1 (33.3)	2 (33.3)	2 (28.5)	5 (35.7)
Years of age (median, range)	57.5 (37-75)	68 (37-72)	58 (51-71)	57 (45-68)	56.5 (49-75)
Disease duration, years (median, range)	30 (12-60)	49 (23-51)	33 (20-46)	30 (18-35)	30 (12-60)
ASAS-HI (median, range)	2.1 (0-11)	1 (0-3)	2.1 (0-6)	2 (0-8)	3 (0-11)
Physical activity level (IPAQ), total MET per week (median, range)	3919 (1520-14958)	3916 (2526-6933)	5820 (2298-10260)	2697 (1600-10638)	2853 (1520-14958)
Supervising PTs (n=4)					
PT group leader, gender (female/male)		f	f	f	E
PT group leader, age, years (median, range)	48.2 (34-54)	52	45	34	5
PT group leader since, years (median, range)	12 (2-23)	23	8	0	16
n, number; ASAS-HI, Assessment of SpondyloArthritis in	iternational Society Heal	th Index; PT, Physiotherapists;	IPAQ, International Physic	al Activity Questionnaire.	

I, FIIJSI L K ก 5 or ray to Arter the second 6 5 HO-III, AS þ

#### Changes according to PRISM

Neither the perceived burden of disease (distance to self at T0:  $13.7 \pm 7.2$ cm; at T1:  $12.1 \pm 6.2$ cm; t=0.386, p=0.703) nor the importance of PA (distance to self at, T0:  $7.3 \pm 5.3$ cm; at T1:  $7.0 \pm 4.5$ cm); t=0.246, p=0.808) changed over six months. The perceived importance of PA correlated with the International Physical Activity Questionnaire (IPAQ) measured METs at T1 (r=0.572, p=0.00), no correlation between perceived burden of disease and METs was found.

#### Effectiveness of individualised PA recommendations: Fitness status

Table 4 reports the results of the fitness assessments at baseline (T0) and after 6 months (T1). Patients' cardiorespiratory fitness (ES 1.22 (95%Cl 0.59,1.90), large ES and significant) and core strength ventral muscle chain (ES 0.62 (95%Cl 0.19, 1.06), medium ES and significant) improved over the six months. Core strength lateral and dorsal muscle chain, balance and flexibility did not change significantly over six months.

The linear mixed model analysis showed that disease activity was negatively (slope=-1.49, p=0.01) associated with cardiorespiratory fitness and the PA level (MET) positively (slope=3.28, p=0.01). The PA level was also positively associated with core muscle strength (slope= 8.62, p=0.05). Age was positively associated with flexibility (slope=0.10, p=0.04), but negatively associated with balance time (slope = -0.84 (p=0.00).

Fitness assessment n missing values mean (SD) median (IQR 25/75)	T0 (n=30)	T1 (n=30)	Change (n=30)	ES (95% confidence interval)
Cardiorespiratory fitness, VO2max n missing values mean (SD) median (IQR 25/75)	4 35.2 (6.5) 34.5 (32.6/39.7)	11 41.6 (6.5) 39.9 (38.2-46.0)	13 7.9 (6.5)	*1.22 (0.59, 1.90)
Core strength, ventral, sec n missing values mean (SD) median (IQR 25/75)	1 78.8 (51.9) 69.0 (40.0-100.0)	5 107.5 (65.4) 90.0 (64.0-149.0)	5 23.8 (38.6)	*0.62 (0.19, 1.06)
<b>Core strength, lateral, sec</b> n missing values mean (SD) median (IQR 25/75)	1 49.0 (31.5) 39.0 (29.0-66.0)	5 57.9 (31.1) 51.0 (38.0-82.0)	5 7.6 (30.2)	0.17 (0.01, 0.55)
<b>Core strength, dorsal, sec</b> n missing values mean (SD) median (IQR 25/75)	1 58.5 (50.1) 35.0 (24.0-80.0)	7 71.7 (56.6) 59.0 (32.5-85.5)	7 9.7 (35.5)	0.30 (0.02, 0.65)
One leg stance, open eyes, mean of 3 reps, sec n missing values mean (SD) median (IQR 25/75)	0 43.1(21.1) 60.0 (24.5-60.0)	4 46.6 (19.6) 60.0 (34.0-60.0)	4 3.7 (14.0)	0.10 (0.00, 0.48)

# Table 4. Results of fitness assessments of the whole patient sample at T0 (baseline) and T1 (after six months).
#### Table 4. Continued

<b>Fitness assessment</b> n missing values mean (SD) median (IQR 25/75)	T0 (n=30)	T1 (n=30)	Change (n=30)	ES (95% confidence interval)
One leg stance, closed eyes, mean of 3 reps, sec n missing values mean (SD) median (IQR 25/75)	0 7.7 (5.8) 5.5 (5.0-15.5)	4 9.5 (7.8) 7.9 (3.0-14.9)	4 1.7 (5.5)	0.30 (0.02, 0.63)
<b>BASMI, Score</b> n missing values mean (SD) median (IQR 25/75)	0 3.1 (2.3) 2.4 (1.2-4.5)	4 2.9 (2.0) 2.3 (1.2-4.7)	4 -0.2 (0.9)	*-0.28 (-0.68,0.12)
IPAQ, MET per week n missing values mean (SD) median (IQR 25/75)	1 5013 (3479) 3916 (1520-14958)	9 4101 (4262) 3150 (1470-4502)	10 -199.8 (2712.2)	0.15 (0.00, 0.58)
<b>BASDAI</b> n missing values mean (SD) median (IQR 25/75)	1 3.0 (1.7) 3.0 (1.8-4.1)	9 3.0 (2.2) 2.2 (1.3-4.4)	10 0.0 (1.1)	*0.06 (-0.39, 0.51)
<b>BAS-G</b> n missing values mean (SD) median (IQR 25/75)	1 2.9 (1.8) 2.4 (1.4-4.5)	9 2.6 (1.8) 2.2 (1.1-3.7)	10 -0.2 (1.6)	0.02 (0.00, 0.49)
<b>ASAS-HI</b> n missing values mean (SD) median (IQR 25/75)	1 3.0 (2.6) 2.1 (1.0-4.0)	9 2.5 (0.7) 3.0 (2.0-3.0)	10 -0.8 (3.3)	0.12 (0.00, 0.54)

Effect sizes (ES) were computed from Wilcoxon tests and paired t-tests, using r for non-parametric and cohen's d for parametric\* analysis, respectively. Interpretation of ES: r: 0.1-0.3 (small), 0.3-0.5 (moderate) and > 0.5 (large). Cohen's d: 0.2 (small), 0.5 (medium), 0.8 (large).

IQR, interquartilrange; sec, seconds; SD, standard deviation; BASMI, Bath Ankylosing Spondylitis Mobility Index; reps, repetitions; IPAQ, International Physical Activity Questionnaire; BASDAI, Bath Ankylosing Spondylitis Disease Activity Index; BAS-G, Bath Ankylosing Spondylitis patient global score; ASAS-HI, Assessment of SpondyloArthritis international Society Health Index.

#### Results at the PT level

#### Fidelity to the new exercise group concept

Three of the four PTs (75%) performed and reported on three group counselling sessions within the 6 months, with a mean duration of  $18 (\pm 13)$  minutes. Due to a misunderstanding, one PT did no group counselling sessions. The counselling sessions consisted of knowledge transfer (e.g. reading and discussing PA recommendations) and skills training (e.g. training with a heat rate monitor). The number of patients participating in the group counselling interventions varied between four and twelve per session and group.

All four PTs performed and reported three individual exercise counselling sessions with 27 of 30 patients (90%). Two patients dropped out after the first session and one patient participated in only two sessions before leaving the group for personal reasons. Of the 27 patients receiving the three counselling sessions, 11 (40%) received the three counselling sessions within 6 months, as required by the concept; 10 (37%) within 9 months, and six (22%) within 12 months. The delay was due to organisational factors in all cases.

The mean duration of an individual counselling session was 35 (±7) minutes. During the first session, the PTs used the PRISM to take the patient's history, explain the results of the fitness assessment and, through shared decision-making, to define one, or more, training goals. Subsequently, based on the patient's preferences, an action plan was developed. During the second and third sessions, PTs used PRISM to evaluate, together with the patient, whether the goal(s) had been reached, or whether adaptations to the goals or planned actions were needed due to changes in fitness status or PA-behaviour.

The mean duration of PRISM usage was 9 ( $\pm$ 3) minutes during the first session, 10 ( $\pm$ 4) minutes during the second, and 7 ( $\pm$ 3) minutes during the third.

Using PRISM, PTs measured and reported on the burden of disease and importance of PA for 25 participants.

#### Feasibility and satisfaction

PTs perceived the PRISM communication and exercise counselling as very useful, since they previously had little knowledge of the history and exercise goals of their patients. Exercise counselling supported the possibility of personalized exercise. They emphasised that the frequency of the sessions must be flexibly planned, based on the patient's individual training goal and motivation. The PRISM was perceived as a useful door opener, triggering emotions and information, if the patient's attitude was open to the new communication tool. PTs reported that they initially felt unconfident when applying PRISM; this feeling was diminished during the pilot phase through supervision and practice. The PTs perceived varying reactions from patients regarding the fitness assessments, some found that they increased motivation to exercise and others found them demotivating. During exercise counselling, PTs indicated that they found the fitness assessment data difficult to interpret because of the lack of background information (e.g. norm data) and that they rarely relied on them. The PTs felt, that the frequency of fitness assessments could be reduced from bi-annually to annually.

#### Results at the organisational level

Satisfaction and commitment of the SVMB management was found to be high after the pilot phase. Consequently, the SVMB management decided to integrate the new exercise group concept into the organisation and implement it as the new usual care in all its SVMB exercise groups.

#### Maintenance

After 1 year (T2), the same supervising PTs were responsible for their exercise groups and were actively involved in the nationwide implementation process. The four pilot exercise groups lost one member due to personal reasons, but gained two new members. Therefore, 29 patients actively participated in the programme, of whom 21 (72%) participated in the fitness assessments at T2.

#### Adaptations to the implementation strategy

Physiotheranists

Some implementation activities were found to be not fully feasible in our setting. Minor adaptations were made to improve adherence to the recommended exercise behaviour (patients), concept fidelity (PTs), and feasibility of nationwide implementation (SVMB). The barriers and adaptations are described in Table 5.

Strategy	Target of change	Barriers	Adaptation		
Education	Skills	Two-days workshop was too expensive and too time-consuming. Future supervising PTs do not need information on study procedures.	Reduction to one-day workshop + supervision		
Patients with axSpA					
Individualisation, Exercise counselling	Motivation and coping, self- efficacy	Due to organisational reasons, PTs had three exercise counselling sessions with only 40% of group members within 6 months.	Number of exercise counselling sessions reduced to 1-2 per year. No fixed, mandatory group		
		Group setting was not asked for, but 1:1 setting was appreciated.	discussions.		
Individualisation, Exercise counselling	Setting goals, action planning, evaluation goal attainment	PTs did not use the assessments to define exercise goals, as there were no norm data or previous data available.	New project on norm data will be launched. The more measurements are carried out, the more comparisons exist and enable a better interpretation. Paper version adapted (better explanations, examples).		
		The exercise diary was approved as a useful tool. Paper version needs adaptations. Electronic version ("Trainingstagebuch" by Johannes Tscholl) prone to error.			
			Development of electronic version is currently considered by SVMB IT.		
Organisation					
Integration of stakeholders	Acceptance of EG concept	62% satisfied with EG concept, 80% satisfied with exercise counselling, 68% satisfied with fitnessassessments. « MOVE SVMB» was not used by participants.	« MOVE SVMB » changed to «BeFit» and a new logo was released		

Table 5. Adaptations of Implementation Strategy to increase feasibility for nationwide implementation

PA, physical activity; SVMB, The Ankylosing Spondylitis Association of Switzerland; PT= physiotherapist, EG= exercise group, PRISM = Pictorial Respresentation of Illness and Self Measure

Exercise counselling and the fitness assessments were generally appreciated by the patients, but the frequencies of both was perceived as being too high. Accordingly, the fitness assessments were reduced from bi-annually to annually, and the individual exercise counselling sessions reduced from three to two sessions in the first 6 months. The first session was compulsory for all participants and the second dependent on individual needs and training goals. The group counselling sessions were perceived to be useful, but the definition of frequency and structure was removed. There is a great need for a simple exercise diary, covering all the exercise dimensions. An improvement was made to the paper diary through providing better explanations. This version will be used until a digital

exercise diary becomes available.

The 2-day educational workshop for PTs was thought to be too time-consuming for clinicians. Furthermore, the information strategy of the SVMB had already increased awareness and knowledge of the new concept amongst the PT community. The training will therefore be reduced to 1 day and the workshop will focus on exercise counselling (including behavioural change techniques and PRISM).

Based on suggestions from patients, the name "moveSVMB" was changed by the SVMB to "BeFit" ("Bechterew Fitness") and a new logo was developed. Following the pilot phase, information about BeFit was released by the SVMB, with no involvement of the study staff. In addition, the financial and personnel resources were established to ensure a continuation of the implementation.

# Discussion

To our knowledge, this is the first study to translate and implement PA recommendations within a specific patient organisation into an exercise group concept for patients with axSpA and their supervising PTs. Findings will inform the nationwide implementation in Switzerland and similar projects in the Netherlands.

The findings of the pilot phase found the implementation strategy was acceptable and satisfactory to patients, PTs, and the SVMB. This resulted in SVMB management adopting the exercise group concept as its new usual care and implementing it in all its exercise groups. Successful implementation can only happen if the SVMB takes responsibility and initiates crucial changes at organisational level. Findings generally confirm the value of evidence-based exercise groups provided by a patient organisation to support active PA behaviour.

The patients' adherence to recommended exercise behaviour was assessed as insufficient. Reasons for non-adherence of patients may be two-fold: (1) The assessment of adherence to recommended exercise behaviour in patients with axSpA was difficult due to technical problems with the app and possible reporting bias; (2) the adherence was low in terms of performing insufficient amount of exercise according to recommended frequency and dose, as well as reporting exercise sessions. According to the reporting, only four patients fulfilled the recommended amount of aerobic exercise, three of them fulfilled one or two additional exercise dimensions. Factors like limited personal motivation may moderate the level of adherence to change current exercise behaviour or difficulty to achieve the agreed goals. However, it would have been useful to have learned more about the attitude of the patients towards changing their PA behaviour through asking them, prior to the first counselling session, about their satisfaction with their current PA level and whether they were motivated to increase their exercise behaviour. To show improved fitness in exercise dimensions, specific and adequately dosed training is needed. The implementation fidelity in PTs was found to be good, with their adherence to performance and the reporting of exercise counselling beeing sufficient. The level of adherence was affected by factors such as patient' responsiveness, quality of counselling, and organisational aspects (e.g. time constraints). However, only small organisational adaptations needed to be made by the PTs, according to the individual local group setting, such as to the scheduling and the setting of counselling sessions. The reduced number and more flexible planning of counselling sessions should enhance feasibility in the future and result in improved performance (during pilot study only 40% of patients had the counselling sessions within six months).

The four pilot PTs could be viewed as a special group since they were presumably "innovators", an active group keen on new ideas and innovations (30, 31). The pilot PTs were immediately willing to adapt their usual care and confirmed the value of change. Grol and Wensing describe five phases in the process of change for health professionals:

orientation, insight, acceptance, change, maintenance (19). Other supervising PTs might be harder to persuade of the benefits of the new concepts, since old routines need to be first de-implemented to move from acceptance to change and maintenance. Therefore, clear leadership by the SVMB management and the pilot PTs, as role models, are essential for the further implementation among supervising PTs.

Patients reporting could be improved in future though technical applications, which might be more appropriate to the collection of objective data than a paper diary (32). These could also be used as a promotion strategy (e.g. feedback, reinforcement) (33, 34). However, the assessment of PA and reporting on all fitness dimensions simultaneously in the same tool is a big challenge, and a perfect tool is not yet available (35, 36). The adapted paper version of the diary must serve as a placeholder until an electronic solution is developed.

PT needs two critical competencies to work with an exercise group: detailed knowledge of the PA recommendations and communication skills. Even though PA promotion is a key concept of physical therapy (37), and both patients and PTs agree that advice and support of active PA behaviour is part of the therapist's role (38, 39), there seems to be a PT educational gap. Data from the UK and Canada indicate that the majority of PTs neither have detailed knowledge of PA recommendations, nor meet the recommended PA level themselves (40, 41).

A preliminary analysis of counselling sessions has revealed that during the first counselling session one of the techniques most used by PTs to change behaviour (42) was 'general intention building'. No specific goal setting technique was applied (unpublished data). PTs generally use only a few behavioural change techniques (43). This finding underpins the need for continuous training of PTs in communication and behaviour change techniques. Hilberdink and colleagues (44) have mapped a toolkit to optimise exercise behaviour of people with axSpA: (1) Patients should be offered behavioural change guidance, including education, motivational interviewing, goal setting, action planning, monitoring and feedback. Patients need to be encouraged to exercise in a group setting. (2) PTs need to be educated to tailor and instruct an exercise programme and apply behaviour change techniques. Some first effort is made to downscale the lack of evidence-based education and training for health professionals working with rheumatic conditions (45). Others (9, 44, 46) emphasise that interventions are needed for patients with axSpA including behaviour change techniques, that aim to deliver intrinsic motivation to overcome barriers to exercise (47-49) and to sustain an active lifestyle.

In this project, it took several years to determine the appropriate implementation process, following the cyclical approach of Grol and Wensing's implementation of change model (19). During this development period Koorts and colleagues (47) published the PRACTIS guide, which supports the planning and scaling-up of physical activity implementation projects. PRACTIS focuses more specifically on PA promotion research translation and gives

detailed ""how to" support. Both approaches have a common ground and are appropriate theoretical approaches to describe an implementation process and to explain what influences implementation outcomes and the implementation success. One strength of this pilot project is that an implementation scientist was involved from the start, following the "innovative collaborative model" (48). Further strengths are the in-depth contextual analysis that has been undertaken (20, 21) and the stakeholder involvement from the beginning. The group setting for exercise therapy is considered equally effective as individual therapy, but with lower healthcare cost (49). Furthermore, group therapy is attributed a potentially motivating factor for physical activity behaviour change (50). Despite this knowledge, however, the social rules or "group dynamics" (51) within the pilot groups and their impact on individual motivation or PA adherence was not investigated in this study. Social learning (52) within group activities (counselling, and assessment) appears to be an important subject and should be investigated further in future studies.

The sample size of the four pilot groups with 30 patients was small. However, since effectiveness aspects were not the main focus, no prior sample size calculation or biasprevention, such as blinding, were necessary. According to literature, men and women are equally affected by axSpA (53). However, in the pilot groups there were more men than women participated (2:1 ratio). A further limiting factor is the use of self-reported outcomes and not of validated assessments to measure the implementation success. However, this approach was pragmatic and feasible. Future research projects should try to overcome these limitations. The exercise group concept could be a valuable strategy in enhancing patients' daily exercise routines. However, fulfilling the PA recommendations in all exercise dimensions is a very high goal. Continuous monitoring of the exercise group concept is needed to follow-up on patients' exercise behaviour and further improve the strategies of how to fulfil the PA recommendations.

Many patients prefer individual therapy. For example, 32% of the Swiss axSpA patients prefer individual, mainly inactive therapy (20). Dutch patients participating in axSpA exercise groups, however, were positive about subsequently established additional home exercises, annual assessments and training using heart-rate monitors (54). To reach patients who prefer an individual setting, or the inactive therapy approach, a patient education and PA promotion concept supporting the understanding of exercise as medicine needs to be developed.

Further research is also needed to investigate the effects on patients with axSpA of strength or neuromotor exercises alone, not in combination with cardio-respiratory exercises, since no such studies have been conducted as yet (11). It has also previously been stated that flexibility exercises are beneficial to spinal flexibility (2). However, no studies on flexibility exercises alone were found although exercises combined with muscle strength or aerobic exercise did not affect flexibility (11). Recent studies have confirmed that intensive cardio-respiratory exercises is effective, safe and feasible in patients with axSpA (55). No detrimental

effects were shown, rather a beneficial impact on disease activity and risk of cardiovascular diseases was approved (9, 56, 57). Similar strong evidence is lacking for the other fitness dimensions. A recently published review confirmed that exercise intervention studies in people with inflammatory arthritis are not adequately reported, neither was the exercise dose underpinned with evidence (58). Furthermore, better knowledge of adherence patterns, e.g. coping mechanism related to PA, or patient profiles matching group versus individual exercise modes, is needed (59).

During a national implementation, continuous monitoring will be needed to avoid ineffective elements of the exercise group concept (60). The balance between fidelity (degree of adherence) to and adaptation of the implementation strategy and interventions (the degree to which users modify the strategy and interventions to suit local needs) should be monitored and evaluated. A threshold for patient adherence should be defined for better evaluation of an implementation success. Furthermore, economic analysis on the levels of patient, provider, organisation and system should be performed to evaluate the concept's efficacy from an economical point of view.

# Conclusions

This manuscript describes an implementation strategy to translate PA recommendations for patients with axSpA into a new exercise group concept for clinical practice and to embed then into the routine process (usual care) of a patient organisation. The implementation strategy was successfully evaluated in a pilot study including key stakeholders: patients, PTs, and patient organisation. The implementation project addressed a need at organisational level through delivering evidence-based therapy to patients and enhancing the competencies of the supervising PTs. Based on the results of the pilot study, technical problems, which hampered a reliable evaluation of the adherence rate of patients, and some other adaptations were necessary to increase the feasibility for a nationwide implementation. Further work is required in order to improve the quality of the assessments, the exercise counselling, and the individual exercising. The successful pilot strategy and necessary adaptations are explained in detail to make the knowledge available to future translational research projects in the field of RMDs. The upcoming nationwide implementation process needs to be monitored so that implementation problems or changes in the needs of patients with axSpA, PTs, or SVMB can be addressed early.

# Funding

The project was supported financially by Kurt-and-Senta-Herrmann-Foundation (Vaduz) and Rheumaliga Zurich (Zurich).

# Authors' contributions

AKRO drafted the manuscript, all authors reviewed it for important intellectual content, all authors approved the final version to be published. Study initialisation and execution: AKRO, KN. Study conception and design: AKRO, KN, TVV, LVB, SB, AC. Acquisition of data: AKRO, KN, BT, IN. Analysis and interpretation of data: AKRO, KN, IN, AM, TVV, LVB. All authors read and approved the final manuscript.

# Acknowledgements

We thank all individuals with axSpA and physiotherapists who participated in the project. Furthermore, we are grateful to René Braem,CEO of the Swiss Ankylosing Spondylitis Association, for supporting the project. Many thanks also go to Christian Horvath and Chiara Scherrer for their help during data collection.

# **Competing interests**

The authors declare no conflict of interest.

# Additional Files:

- Additional file 1. Standards for Reporting Implementation Studies: the StaRI checklist for completion.
- Additional file 2. Description of assessments (PRISM, fitness assessments, interview with PTs).

# References

- Mathieu S, Gossec L, Dougados M, Soubrier M. Cardiovascular profile in ankylosing spondylitis: a systematic review and meta-analysis. Arthritis Care Res (Hoboken). 2011;63(4):557-63.
- Dagfinrud H, Kvien TK, Hagen KB. Physiotherapy interventions for ankylosing spondylitis. Cochrane Database Syst Rev. 2008(1):CD002822.
- Dursun N, Sarkaya S, Ozdolap S, Dursun E, Zateri C, Altan L, et al. Risk of falls in patients with ankylosing spondylitis. J Clin Rheumatol. 2015;21(2):76-80.
- Sahin N, Ozcan E, Baskent A, Karan A, Kasikcioglu E. Muscular kinetics and fatigue evaluation of knee using by isokinetic dynamometer in patients with ankylosing spondylitis. Acta Reumatol Port. 2011;36(3):252-9.
- Peters MJ, Visman I, Nielen MM, Van Dillen N, Verheij RA, van der Horst-Bruinsma IE, et al. Ankylosing spondylitis: a risk factor for myocardial infarction? Ann Rheum Dis. 2010;69(3):579-81.
- van der Heijde D, Ramiro S, Landewe R, Baraliakos X, Van den Bosch F, Sepriano A, et al. 2016 update of the ASAS-EULAR management recommendations for axial spondyloarthritis. Annals of the rheumatic diseases. 2017;76(6):978-91.
- Noureldin B, Barkham N. The current standard of care and the unmet needs for axial spondyloarthritis. Rheumatology. 2018;57(suppl\_6):vi10-vi7.
- Rausch Osthoff AK, Niedermann K, Braun J, Adams J, Brodin N, Dagfinrud H, et al. 2018 EULAR recommendations for physical activity in people with inflammatory arthritis and osteoarthritis. Annals of the rheumatic diseases. 2018;77(9):1251-60.
- Sveaas SH, Bilberg A, Berg IJ, Provan SA, Rollefstad S, Semb AG, et al. High intensity exercise for 3 months reduces disease activity in axial spondyloarthritis (axSpA): a multicentre randomised trial of 100 patients. Br J Sports Med. 2019; 54:292-7.
- Peter WF, Swart NM, Meerhoff GA, Vliet Vlieland TPM. Clinical Practice Guideline for Physical Therapist Management of People With Rheumatoid Arthritis. Physical therapy. 2021; 101 (8), pzab127.
- Rausch Osthoff AK, Juhl CB, Knittle K, Dagfinrud H, Hurkmans E, Braun J, et al. Effects of exercise and physical activity promotion: meta-analysis informing the 2018 EULAR recommendations for physical activity in people with rheumatoid arthritis, spondyloarthritis and hip/knee osteoarthritis. RMD Open. 2018;4(2):e000713.
- Garber CE, Blissmer B, Deschenes MR, Franklin BA, Lamonte MJ, Lee IM, et al. American College of Sports Medicine position stand. Quantity and quality of exercise for developing and maintaining cardiorespiratory, musculoskeletal, and neuromotor fitness in apparently healthy adults: guidance for prescribing exercise. Medicine and science in sports and exercise. 2011;43(7):1334-59.
- American College of Sports M. ACSM's Guidelines for exercise testing and prescription. tenth edition ed. Philadelphia: Wolters Kluwer; 2018.
- Ferreira de Meneses S, Rannou F, Hunter DJ. Osteoarthritis guidelines: Barriers to implementation and solutions. Annals of physical and rehabilitation medicine. 2016;59(3):170-3.
- Swinkels RAHM, van Peppen RPS, Wittink H, Custers JWH, Beurskens AJHM. Current use and barriers and facilitators for implementation of standardised measures in physical therapy in the Netherlands. Bmc Musculoskeletal Disorders. 2011;12.
- McGuire WJ. Attitudes and attitude change. In: Lindzey G, Aronson E, editors. Handbook of social psychology:Vol 2 Special fields and applications (3rd ed, pp233-346). New York, NY: Random House1985.
- Nicolson PJA, Hinman RS, French SD, Lonsdale C, Bennell KL. Improving Adherence to Exercise: Do People With Knee Osteoarthritis and Physical Therapists Agree on the Behavioral Approaches Likely to Succeed? Arthritis Care Res (Hoboken). 2018;70(3):388-97.
- Roter DL, Hall JA, Merisca R, Nordstrom B, Cretin D, Svarstad B. Effectiveness of interventions to improve patient compliance: a meta-analysis. Med Care. 1998;36(8):1138-61.
- Grol R, Wensing M, Eccles M, Davis D. Improving patient care. The implementation of change in health care. 2nd ed. Oxford: Wiley Blackwell; 2013.
- Rausch Osthoff AK, van der Giesen F, Meichtry A, Walker B, Van Gaalen F, Goekoop-Ruitermans YPM, et al. The perspective of people with axial spondyloarthritis regarding physiotherapy: room for the implementation of a more active approach. Rheumatology Advances in Practice. 2019;0:1-10.
- Niedermann K, Nast I, Ciurea A, Vliet Vlieland T, van Bodegom-Vos L. Barriers and facilitators of vigorous cardiorespiratory training in axial Spondyloarthritis: Surveys among patients, physiotherapists, rheumatologists. Arthritis care & research. 2019; 71(6):839-851
- 22. Pinnock H, Barwick M, Carpenter CR, Eldridge S, Grandes G, Griffiths CJ, et al. Standards for Reporting Implementation Studies (StaRI) Statement. BMJ. 2017;356:i6795.
- 23. Proctor EK, Powell BJ, McMillen JC. Implementation strategies: recommendations for specifying and reporting. Implement Sci. 2013;8:139.
- 24. Curran GM, Bauer M, Mittman B, Pyne JM, Stetler C. Effectiveness-implementation hybrid designs: combining elements of clinical effectiveness and implementation research to enhance public health impact. Med Care. 2012;50(3):217-26.
- Buchi S, Buddeberg C, Klaghofer R, Russi EW, Brandli O, Schlosser C, et al. Preliminary validation of PRISM (Pictorial Representation of Illness and Self Measure) - a brief method to assess suffering. Psychother Psychosom. 2002;71(6):333-

41.

- 26. Miller WRR, S. Motivational Interviewing. Helping people change. Third Edition ed. New York: Guilford Publications; 2012.
- 27. Buchi S, Sensky T. PRISM: Pictorial Representation of Illness and Self Measure. A brief nonverbal measure of illness impact and therapeutic aid in psychosomatic medicine. Psychosomatics. 1999;40(4):314-20.
- Niedermann K, Buchi S, Ciurea A, Kubli R, Steurer-Stey C, Villiger PM, et al. Six and 12 months' effects of individual joint protection education in people with rheumatoid arthritis: a randomized controlled trial. Scand J Occup Ther. 2012;19(4):360-9.
- Scherer SA, Noteboom JT, Flynn TW. Cardiovascular assessment in the orthopaedic practice setting. The Journal of orthopaedic and sports physical therapy. 2005;35(11):730-7.
- 30. Rogers E. Diffusion of innovations. New York: Simon&Schuster Ltd; 2003.
- Prochaska JO, Velicer WF. The transtheoretical model of health behavior change. Am J Health Promot. 1997;12(1):38-48.
- Colley RC, Butler G, Garriguet D, Prince SA, Roberts KC. Comparison of self-reported and accelerometer-measured physical activity in Canadian adults. Health Rep. 2018;29(12):3-15.
- Mercer K, Li M, Giangregorio L, Burns C, Grindrod K. Behavior Change Techniques Present in Wearable Activity Trackers: A Critical Analysis. Jmir Mhealth Uhealth. 2016;4(2):615-23.
- Gell NM, Grover KW, Savard L, Dittus K. Outcomes of a text message, Fitbit, and coaching intervention on physical activity maintenance among cancer survivors: a randomized control pilot trial. J Cancer Surviv. 2020;14(1):80-8.
- Arvidsson D, Fridoltsson J, Borjesson M. Measurement of physical activity in clinical practice using accelerometers. J Intern Med. 2019;286(2):137-53.
- Dowd KP, Szeklicki R, Minetto MA, Murphy MH, Polito A, Ghigo E, et al. A systematic literature review of reviews on techniques for physical activity measurement in adults: a DEDIPAC study. Int J Behav Nutr Phy. 2018;15.
- 37. Hagstromer M, Franzen E. The importance of physical activity and health for physical therapy. Physical Therapy Reviews. 2017;22(3):1-8.
- Kunstler B, Fuller R, Pervan S, Merolli M. Australian adults expect physiotherapists to provide physical activity advice: a survey. Jounal of Physiotherapy. 2019;65:230-6.
- Shirley D, van der Ploeg HP, Bauman AE. Physical activity promotion in the physical therapy setting: perspectives from practitioners and students. Physical therapy. 2010;90(9):1311-22.
- Lowe A, Littlewood C, McLean S, Kilner K. Physiotherapy and physical activity: a cross-sectional survey exploring physical activity promotion, knowledge of physical activity guidelines and the physical activity habits of UK physiotherapists. BMJ Open Sport Exerc Med. 2017;3(1):e000290.
- Neil-Sztramko S, Ghayyur A, Edwards J, Campbell KL. Physical Activity Levels of Physiotherapists across Practice Settings: A Cross-Sectional Comparison Using Self-Report Questionnaire and Accelerometer Measures. Physiotherapy Canada. 2016;69(2):1-9.
- 42. de Bruin MK, G.; Schaalma, H; Hospers H. Coding manual for behavioral change techniques 2007.
- 43. Kunstler BE, Cook JL, Freene N, Finch CF, Kemp JL, O'Halloran PD, et al. Physiotherapists use a small number of behaviour change techniques when promoting physical activity: A systematic review comparing experimental and observational studies. Journal of Science and Medicine in Sport. 2018;21(6):609-15.
- 44. Hilberdink B, van der Giesen F, Vliet Vlieland T, Nijkamp M, van Weely S. How to optimize exercise behavior in axial spondyloarthritis? Results of an intervention mapping study. Patient education and counseling. 2020;103(5):952-9.
- 45. Metsios GS, Fenton SA, Moe HR, van der Esch M, van Zanten JV, Koutedakis Y, et al. Implementation of Physical Activity into routine Clinical pracTice in Rheumatic Musculoskeletal Disease: The IMPACT-RMD study protocol and rationale. Mediterr J Rheumatol. 2019;30(4):231-6.
- Sveaas SH, Dagfinrud H, Johansen MW, Pedersen E, Wold OM, Bilberg A. Longterm Effect on Leisure Time Physical Activity Level in Individuals with Axial Spondyloarthritis: Secondary Analysis of a Randomized Controlled Trial. J Rheumatol. 2020; 47(8):1189-1197.
- Koorts H, Eakin E, Estabrooks P, Timperio A, Salmon J, Bauman A. Implementation and scale up of population physical activity interventions for clinical and community settings: the PRACTIS guide. Int J Behav Nutr Phys Act. 2018;15(1):51.
- 48. Zullig LL. Integrating the implemenation science team over the lifecycle of an intervention (manuscript in preparation, presented IMPACT Basel February 2021). 2021.
- O'Keeffe M, Hayes A, McCreesh K, Purtill H, O'Sullivan K. Are group-based and individual physiotherapy exercise programmes equally effective for musculoskeletal conditions? A systematic review and meta-analysis. British journal of sports medicine. 2017;51(2):126-32.
- Knittle K, Nurmi J, Crutzen R, Hankonen N, Beattie M, Dombrowski SU. How can interventions increase motivation for physical activity? A systematic review and meta-analysis. Health Psychol Rev. 2018;12(3):211-30.
- 51. Blixt L, Solbraekke KN. The significance of social rules in group training sessions. Disability and Rehabilitation. 2017;39(24):2477-83.
- 52. Bandura A. Social Learning Theory. University of Michigan: Prentice Hall; 1977.
- Rusman T, van Vollenhoven RF, van der Horst-Bruinsma IE. Gender Differences in Axial Spondyloarthritis: Women Are Not So Lucky. Curr Rheumatol Rep. 2018;20(6):35.
- 54. Hilberdink B, van der Giesen F, Vliet Vlieland T, van Gaalen F, van Weely S. Supervised Group Exercise in Axial Spondyloarthritis: Patients' Satisfaction and Perspective on Evidence-Based Enhancements. Arthritis care & research.

2020;72(6):829-37.

- Sveaas SH, Smedslund G, Hagen KB, Dagfinrud H. Effect of cardiorespiratory and strength exercises on disease activity in patients with inflammatory rheumatic diseases: a systematic review and meta-analysis. British journal of sports medicine. 2017;51(14):1065-72.
- Niedermann K, Muggli C, Dagfinrud H, Hermann M, Tamborrini G, Ciurea A, Bischoff-Ferrari H. Effect of cardiovascular training on fitness and perceived disease activity in people with ankylosing spondylitis. Arthritis care & research. 2013;65(11):1844-52.
- 57. Saracoglul KG, Okur EO, Afsar E, Seyyar GK, Calik BB, Taspinar F. The effectiveness of specific exercise types on cardiopulmonary functions in patients with ankylosing spondylitis: a systematic review. Rheumatology international. 2017;37(3):409-21.
- 58. Boniface G, Gandhi V, Norris M, Williamson E, Kirtley S, O'Connell NE. A systematic review exploring the evidence reported to underpin exercise dose in clinical trials of rheumatoid arthritis. Rheumatology. 2020;59(11):3147-3157.
- Davergne T, Tekaya R, Deprouw C, Sellam J, Tournadre A, Mitrovic S, et al. To apply the recent EULAR recommendations, more knowledge on adherence patterns to medication and to physical activity is needed. Joint Bone Spine. 2021;88(2).
- 60. Carroll C, Patterson M, Wood S, Booth A, Rick J, Balain S. A conceptual framework for implementation fidelity. Implementation Science. 2007;2.
- 61. Bennett H, Parfitt G, Davison K, Eston R. Validity of Submaximal Step Tests to Estimate Maximal Oxygen Uptake in Healthy Adults. Sports Med. 2016;46(5):737-50.
- 62. Buckley JP, Sim J, Eston RG, Hession R, Fox R. Reliability and validity of measures taken during the Chester step test to predict aerobic power and to prescribe aerobic exercise. British journal of sports medicine. 2004;38(2):197-205.
- 63. BASPO BfS. Manual Leistungsdiagnostik. 2015.
- Rausch AK, Baltisberger P, Meichtry A, Topalidis B, Ciurea A, Vliet Vlieland TPM, et al. Reliability of an adapted core strength endurance test battery in individuals with axial spondylarthritis. Clinical Rheumatology. 2021; 40(4):1353-1360.
- Springer BA, Marin R, Cyhan T, Roberts H, Gill NW. Normative values for the unipedal stance test with eyes open and closed. J Geriatr Phys Ther. 2007;30(1):8-15.
- 66. Sieper J, Rudwaleit M, Baraliakos X, Brandt J, Braun J, Burgos-Vargas R, et al. The Assessment of SpondyloArthritis international Society (ASAS) handbook: a guide to assess spondyloarthritis. Ann Rheum Dis. 2009;68 Suppl 2:ii1-44.
- 67. Hallal PC, Victora CG. Reliability and validity of the International Physical Activity Questionnaire (IPAQ). Medicine and science in sports and exercise. 2004;36(3):556.
- Jones SD, Steiner A, Garrett SL, Calin A. The Bath Ankylosing Spondylitis Patient Global Score (BAS-G). Br J Rheumatol. 1996;35(1):66-71.
- Kiltz U, van der Heijde D, Boonen A, Braun J. The ASAS Health Index (ASAS HI) a new tool to assess the health status of patients with spondyloarthritis. Clin Exp Rheumatol. 2014;32(5 Suppl 85):S-105-8.
- 70. Graf J, Claes D, Greiner W. The German version of the EuroQol Questionnaire. Z Gesundh Wiss. 1998;6:3-20.

# 8

General discussion

Physical Activity (PA) is one of the most important lifestyle factors influencing human health. The general and disease-specific health benefits of PA are well known to the general population and to those people living with rheumatic musculoskeletal diseases (RMDs) (1-4). The importance of PA as an intervention for the management of RMDs has been endorsed by various stakeholders, e.g. the World Health Organisation/WHO (5), the European Commission (6), and the European Alliance of Associations for Rheumatology/ EULAR (7). The early translation of research evidence into clinical care benefits patients with RMDs greatly.

Axial spondyloarthritis (axSpA) is a chronic inflammatory RMD that predominately affects the axial skeleton, sacroiliac joints, and spine. It causes structural and functional impairments that have a significant impact on the individual's quality of life and ability to work (8, 9).

This thesis is focused on the effects and promotion of PA and exercise on people with RMDs, and particularly those with axSpA.

The overall aim of this thesis was to document the journey from the development of the EULAR PA recommendations for people with RMDs (including axSpA) to the practical implementation of a concept for group exercise therapy for people with axSpA that conforms to these recommendations. Chapter 1 outlines the background of this thesis. Chapter 2 describes the development and definition, based on expert opinion and current evidence, of the EULAR PA recommendations for people with RMDs. Chapter 3 addresses the effectiveness of PA interventions (based on the four exercise dimensions of cardiorespiratory, muscle strength, flexibility and neuromotor) and the promotion of PA to meet public health recommendations for people with RMDs. The following chapters, 4 to 7, outline the process of translating the EULAR PA recommendations into the "BeFit" group exercise therapy concept. This concept was subsequently pilot-tested in group exercise therapy for people living with axSpA in Switzerland.

# General conclusions

In Chapter 2, one of the EULAR recommendations on PA developed by an international, multistakeholder task force, states that the public health recommendations are also applicable to people with RMDs. Chapter 3 corroborates the evidence that exercising according to public health PA recommendations is effective, safe and feasible for people with RMDs, and that exercises should cover the four exercise dimensions of cardiorespiratory, muscle strength, flexibility and neuromotor exercise. The promotion of PA consistent with these recommendations should be an integral part of routine care throughout the course of RMDs (Chapter 2). It is noted, however, that people with RMDs may have disease-related barriers to PA and additional requirements might need to be considered. Therefore, the EULAR PA recommendations emphasise other important aspects relating to the planning and delivery

of PA interventions, such as shared decision-making, the consideration of behavioural change techniques, and the addressing of contraindications and related comorbidities (Chapter 2). The EULAR PA recommendations for people with RMDs strengthen the non-pharmacological treatment of their diseases.

Specifically in axSpA, exercise recommendations have been made for decades, however, new insights warranted an update of research on this topic. These insights concerned firstly the realisation that people with axSpA have an increased risk of cardiovascular diseases and, secondly, the questionable therapeutic validity of conventional exercises programmes compared with exercising according to general PA recommendations for people with axSpA. Through the publication of the EULAR PA recommendations (Chapter 2), appropriate exercise principles and standards of support were determined for people with axSpA. However, new scientific insights do not automatically get implemented into daily practice and tailored strategies were needed to embed the EULAR PA recommendations into routine clinical care. This thesis presents the process of translating the state-of-the-art knowledge on exercising according to the EULAR PA recommendations into the group exercise concept BeFit and, thus, to complement the care of people with axSpA living in Switzerland (Chapter 7). The BeFit concept aims to support people with axSpA in performing exercise training according to the EULAR PA recommendations by means of several implementation activities. The development of these tailored implementation strategies required several steps:

• The evaluation of the current usage, experiences, and preferences regarding the delivery of individual and group exercise physiotherapy in Switzerland and the Netherlands (Chapter 4). This study showed that: more than 80% of participants had received physiotherapy; the interventions in an individual setting were mainly of a passive nature; and that the majority of participants were unaware of the increased cardiovascular risk and the subsequent importance of aerobic exercises.

These findings led to:

- A qualitative study exploring patients' behavioural, normative and control beliefs regarding PA (Chapter 5). This study confirmed that some people with axSpA were unaware of the conceptual difference between general PA and specific aerobic exercises. However, the findings also revealed that people with axSpA had a positive attitude towards exercise and that they considered exercising to be an effective self-management strategy.
- A cross-sectional study (Chapter 6) focusing on finding a suitable assessment to measure core strength. This is essential when developing an exercise programme tailored to individual capabilities. Previously, an appropriate assessment for people with axSpA in a group setting had not been available. The intra-tester reliability was found to be moderate to substantial. Additionally, no correlations with diseasespecific factors, such as functional status, pain, or perceived strength, were identified. The adapted test battery chosen in the study to assess core strength was therefore used in the BeFit exercise concept.
- A final description of the implementation strategies for BeFit and pilot-testing

of the programme (Chapter 7). Overall, the acceptance and satisfaction with the new concept were found to be 'sufficient' at all levels, i.e., that of patients, physiotherapists, and the organisation. However, some adaptations were deemed necessary prior to its nationwide implementation, e.g., a reduction in the frequency of assessments and exercise counselling, the need of developing an electronic exercise diary (Chapter 7).

This thesis adds value by describing the entire development process in Switzerland, from the development of the EULAR PA recommendations for people with RMDs to their clinical implementation for people with axSpA. The relevant stakeholders were involved at all stages.

# Methodological considerations

According to the principles of good clinical research practice, scientific quality standards for design, execution and reporting, as well as ethical considerations, should be taken into consideration (10). The studies described in this thesis included a systematic review with meta-analyses, guideline development, qualitative and quantitative observations, and a pilot implementation study with a hybrid design. All such studies have inherent strengths and limitations.

# Chapters 2 and 3: Development of the EULAR PA recommendations

The methodology of the development of the EULAR recommendations is clearly defined and the related scientific publications were all multi-stage peer-reviewed and EULARapproved (11). The EULAR standardised operating procedures (SOPs) (11) align with the AGREE (Appraisal of Guidelines for Research and Evaluation) instrument for developing and reporting guidelines (12).

Some strengths and limitations of this approach should be discussed. Firstly, to increase their generalisability and applicability, the recommendations focused on three relatively prevalent RMDs, i.e., rheumatoid arthritis, spondyloarthritis, and hip/knee osteoarthritis. However, due to the large heterogeneity of these conditions, this may have limited the precision of the recommendations. On the other hand, it can also be debated whether this selection of RMDs may limit the generalizability to other RMDs. In that respect, the further development of additional, disease-specific recommendations could be of use. Moreover, there were some challenges regarding the informing SR and MA described in Chapter 3. The data extraction and interpretation of the included studies were hampered by the overall incomplete reporting of interventions. The investigated exercise intervention modalities or dosage were often not transparent (FITT-VP principles: frequency, intensity, type, time, volume, progression). Therefore, it was often difficult to determine whether a study fulfilled the inclusion criteria. Consistent reporting, according to the template for intervention description and replication checklist (TIDieR) (13) or FITT-VP principles

(14), would have eased understanding and interpretation of the studies. A vast array of outcomes and assessments exists in the field of PA, including in the exercise dimensions of cardiorespiratory, strength, flexibility and neuromotor exercise, that limits the conduct and generalisability of MAs. Indeed, the observed variation in reporting outcomes was large, elucidating the fact that a consensus on PA outcomes and their assessment could facilitate research using comparative outcomes and findings. For example, the use of verified objective measures and similar definitions of low, moderate, and high intensity PA, could positively influence the internal and external validity of findings. The task force members, however, were aware of the limitations regarding cross-study comparison and the MA formed a useful basis for their work. The need for more studies investigating long-term interventions with a larger number of participants and harmonised assessments nevertheless remains. A further limitation was that no, or minimal, information was found on the effects of the following interventions: flexibility exercise alone on flexibility/joint range of motion; of neuromotor exercise on neuromotor performance; or of interventions that targeted all four exercise dimensions. Lastly, the searches were performed over the period up to 7/2017 and do not, therefore, include more recent evidence. The research field of exercise interventions and the promotion of PA is a highly active one. In addition, the current WHO PA recommendations differ slightly from the recommendations that were authoritative during the development process of the EULAR PA recommendations, e.g., with respect to the minimum amount of activity to be counted as aerobic activity. From today's perspective, it can be assumed that important literature, e.g. on high intensity aerobic exercise, dose-response-relationship, or on the value of wearable-based PA promotion, could be added to fully reflect the fastdeveloping state of knowledge (4, 15-25). According to the update policy embedded in the EULAR SOPs (11), new insights should be monitored and considered by the convenor and members of the original task force, or experts in the field who can also apply for an update grant to EULAR. This process might be initiated in the near future.

### Chapters 4 to 7: Implementation of the EULAR PA recommendations

The average time between an innovation and the application of the evidence into health care is 17 years (26). Even more important is the prompt real-world implementation of evidencebased interventions to improve health care. Therefore, specific implementation studies are needed to evaluate the effectiveness of a PA intervention in the real-world setting and the reasons for the success or failure of PA interventions in a specific context (27).

# Study designs in implementation research

The benefit of a hybrid study design is that it takes a dual focus a priori on the assessments of clinical effectiveness and implementation (28). This enables a more rapid transfer of evidence into clinical care. There are three types of hybrid study designs that vary depending on the level of their focus on effectiveness and implementation: hybrid type 1 is primarily used to investigate the effectiveness of a clinical intervention, while simultaneously gathering information on its implementation in a real-world setting (28); hybrid type 2 has a dual focus on the effectiveness of the intervention and implementation related factors (29). It

was decided to use the hybrid type 3 design for the study on the implementation of a novel approach for group exercise therapy in axSpA (Chapter 7). A hybrid type 3 is focused largely on the implementation of the intervention once the effectiveness of the intervention has been established (28). This design is characterised by an explicitly described plausible implementation strategy in the real-world setting, together with clearly defined outcomes for implementation (primary) and intervention (secondary) components (29). This design is used for pilot-implementation studies evaluating feasibility and risks compromising the implementation or intervention. The study described in Chapter 7 indeed struggled with unclear cut-offs for sufficient implementation outcomes, such as patient adherence. However, lessons learned from this pilot study will feed into the national implementation study, which is likely also designed as a hybrid type 3 study.

#### Implementation approaches and frameworks

The task force responsible for the EULAR PA recommendations (Chapter 2) stated that specific strategies are needed to implement the recommendations in the different health systems and cultures across Europe. To ensure tailoring to the individual target groups, all stakeholders need to be involved from the start of the process. Accordingly, in Switzerland, the EULAR PA recommendations were translated into the BeFit concept for group exercise therapy, to be offered by the Ankylosing Spondylitis Association of Switzerland (SVMB) for people with axSpA (Chapter 7). Some of the concept development steps are described in Chapters 4-6, whereas other important steps have been published (or submitted) independently of this thesis. These steps included the: evaluation of the barriers and facilitators for vigorous exercise from the perspectives of the patients, physiotherapists, and rheumatologists (24); translation and validation of the German exercise self-efficacy scale (30); evaluation of the communication skills of physiotherapists (31); and the evaluation of the feasibility of assessments in a group exercise setting (32).

The implementation steps established in this thesis were the development of an implementation strategy, nested implementation activities, and evaluation procedures for each relevant target group, e.g., people with axSpA, physiotherapists and SVMB management.

Several systematic approaches are available to guide the process of translating research into practice (33). The implementation of change model by Grol&Wensing was applied in this thesis (34). It is comprised of seven iterative steps: (1) the development of a proposal for change; (2) analysis of the current situation; (3) understanding of the target groups and setting; (4) development of strategies and measures of change; (5) pilot testing; (6) adaptation; (7) continuous evaluation. This is one of the most established approaches from a vast number of models/frameworks/theories underlying implementation sciences (35) and has been applied across a wide range of health-care settings and clinical behaviours (36-38).

When the conceptualisation of this thesis was already well advanced, Koorts and colleagues

(39) published a similar, but more specific, guide on the planning and scaling-up of PA implementation projects, called PRACTIS (**Prac**tical planning for Implementation and **S**cale-up). PRACTIS follows four steps that guide, with detailed 'how to' support, through: 1) exploration of the implementation setting; 2) identification and engagement of the key stakeholder; 3) identification of barriers and facilitators; and 4) addressing the potential barriers and facilitators to implementation. An increasing number of studies implementing PA interventions are using this approach (40-42).

Both approaches have common ground and are theoretical approaches appropriate to describing an implementation process and explaining the influences on the implementation outcomes and success. However, Grol&Wensing's model covers the whole implementation process, emphasising the importance of continuous evaluation and adaptation of strategies and measures. PRACTIS focuses on the initiation of contextually relevant processes to translate research into practice. It implies a participatory approach, putting emphasis on the early identification and systematic collaboration of key stakeholders at multiple levels, whether part of the study staff or advisory board. Following the model of Grol&Wensing, the important stakeholders, such as patients, physiotherapists, rheumatologists, and SVMB management were indeed involved, but not in the systematic and collaborative way as proposed by PRACTIS. In this study, people with axSpA, the patient organisation SVMB, and experts representing several relevant professions were included at some stage during the development of the implementation strategy, through participation in focus groups, interviews and surveys. Engaging people with axSpA and physiotherapists more intensively from the very start, e.g., by means of regular consultation at a stakeholder advisory board, could have had a more positive impact on the acceptance of the implementation activities of the new group exercise concept. Although acceptance was found to be sufficient during the pilot phase, a great deal of effort was needed by the staff members, patients and physiotherapists. Both, patients and physiotherapists had to be persuaded to participate. Patients and physiotherapists had to be educated on the new insights regarding exercise, especially the importance of cardiorespiratory exercise. They had to be convinced of the value of the new group exercise concept. The patients, particularly, approved of the evidence-based adaptations to the concept. They wanted individual training support and to be sure that the health insurances would continue to pay for the group exercises. However, only 70% of the pilot group members agreed to participate in the new group exercise concept and give informed consent for data collection. The physiotherapists supervising the pilot groups were very motivated, enthusiastic about the application of the new ideas and did not expect to be fully compensated for their effort. Although the SVMB paid for the group exercise therapy and counselling sessions, the introduction of new ideas always takes the physiotherapist additional time, such as for organising appointments or explaining procedures. The physiotherapists were willing to adapt their usual care and facilitated a positive attitude towards BeFit within the groups. This positive view may consequently support the long-term maintenance of the therapy for both physiotherapists and the group members with axSpA. According to Grol&Wensing (34), health professionals go through

five phases of change behaviour: orientation, insight, acceptance, change, maintenance. Physiotherapists generally (other than those supervising the pilot groups) may initially not be so enthusiastic about the change to the group exercise concept and could resist its implementation. Clear leadership by SVMB management and the pilot physiotherapists, as role models, is essential for national implementation.

# Research into insights in current care delivery and patients' beliefs regarding PA and exercise

Chapter 4 presented a cross-sectional survey conducted on the use of physiotherapy in the western region of the Netherlands and the German-speaking part of Switzerland. This study was performed to learn more about the current situation in both countries and to provide information for upcoming implementation projects in Switzerland (project lead K. Niedermann and A. Rausch) and in the Netherlands (project lead T. Vliet Vlieland, S. van Weely, and B. Hilberdink). A number of limitations are associated with this survey. The recruitment and selection of respondents varied in the two countries, being drawn from the members of a patient organisation in Switzerland and from outpatient rheumatology clinics of three hospitals in the Netherlands. The survey was self-developed and could have been more specific about the therapy modalities (e.g., intensity, content of group therapy), especially in the group exercise setting. No pilot testing was performed to evaluate face or construct validity. Finally, the translation from the Dutch language into German and its cultural adaption was not guided by an appropriate approach, such as TRAPD (Translation, Review, Adjudication, Pretest, Documentation) (43) or back-translation (44). An additional survey conducted with the patients' physiotherapists could have delivered interesting information from their perspective. It is conceivable that physiotherapists would have described the content of the therapy differently to their patients, which would be relevant for the validity of findings. Although only an indication, research into other rheumatic conditions in which both the patient and physiotherapist perspectives were examined, showed their perceptions to be quite similar (45-48). Despite these limitations, the survey was a good starting point to explore the current use of physiotherapy by people with axSpA in the individual and group settings.

The focus group study on the beliefs related to PA (Chapter 5) should be interpreted with caution in terms of generalisability. As is common in focus groups, participants have a vested interest in the topic under study (49). Therefore, findings might not be representative of those individuals disinterested in PA and the study may thus suffer from selection bias (50). However, purposive and convenience sampling was also used to gain a broad insight into the topic (51). Generating a broad, but theory-driven, outline of views and ideas regarding PA was also the reason for the decision to use focus groups instead of individual interviews, despite the fact that individual interviews are in general considered to be more efficient in terms of time and the quality of in-depth answers (52). The focus group technique may have a further specific value since it depends less on the questions of the moderator. The moderator rather has the role of a facilitator with an investigative focus, asking questions, controlling the dynamics of group discussions, and sometimes engaging in dialogue with specific participants (53).

The findings of the study revealed the often neglected difference between the concepts of cardiorespiratory exercise and general PA, and the possible consequential impact on health. Furthermore, the study results support the importance of general PA and exercise as a self-management strategy for people with axSpA. This is reflected in the phenomenon that facilitators, such as internal motivation, may outweigh strong disease-related barriers, such as pain or fatigue. Physiotherapists need to learn how to use this knowledge wisely and integrate it into long-term interventions to promote successful active PA behaviour. With respect to cardiorespiratory training, the studies by Sveaas and colleagues (16, 18) show that even high intensity training is feasible and accepted by people with axSpA and that it has a positive effect on disease activity and quality of life.

#### Assessment of core strength in people with axSpA

According to the EULAR PA recommendations, regular fitness assessments in all the exercise dimensions are crucial in identifying those areas that have potential for improvement. However, a suitable assessment to measure core strength in people with axSpA was unavailable. This is a deficiency, since core strength exercising was part of previous exercise programmes for people with axSpA (54). As yet, no randomized controlled trials have been conducted to investigate the effect of core muscle strength exercises alone on muscle strength performance and disease-related outcomes in people with axSpA (Chapter 3). To provide an easy-to-perform assessment and to prepare for possible studies focusing on this under-researched topic, the adaption and testing of a core strength assessment battery (55) formed an important part of this thesis (Chapter 6). One limitation was that the reliability of the core strength assessment battery was tested by the physiotherapists normally leading the group exercise therapy, rather than an independent assessor in a laboratory setting. However, the pragmatic approach yielded clinically relevant information, showing that the physiotherapists were challenged in terms of timing and the accuracy of assessments. The physiotherapists performed the assessments during the usual exercise session and reported that it was difficult to measure all the participants within the planned schedule. This may have influenced their adherence to protocol. To improve reliability, the test conditions were adapted for the BeFit concept, with an external assessment team of physiotherapy students performing the regular assessments.

# The implementation of the EULAR PA recommendations for RMDs: current status and lessons learned

Since the EULAR PA recommendations (Chapter 2) were published, some initiatives have been launched to (linguistically) translate (56, 57) or to implement them into clinical care for people with axSpA (58-60). The working groups, led by Nikiphorou (60) and Metsios (58), followed an educational approach by developing EULAR recommendations on self-management intervention and learning-courses for health professionals in rheumatology respectively, both referring to the EULAR PA recommendations. Dutch researchers have also

carried-out research on the implementation of the EULAR PA recommendations specifically for people with axSpA. Firstly, Hilberdink et al. used the Intervention Mapping Protocol to develop components of an ideal intervention for people with axSpA living in the Netherlands and optimise exercise behaviour through: 1) behaviour change guidance; 2) training for physiotherapists on how to provide behaviour change guidance and how to tailor exercises; and 3) encouraging exercise in a group setting (61). Secondly, based on surveys, it was concluded that PA promotion intervention for the Dutch setting should focus on aerobic PA, vigorous-intensity PA, and education of the physiotherapists supervising the group exercises (62, 63). Thirdly, a hybrid effectiveness-implementation study was performed to evaluate the implementation success and effectiveness of an adaptation of the supervised exercise groups in four regions of the Netherlands (59). Patient satisfaction was found to be high. However, criticism arose concerning the lack of a patient-education intervention and tailored exercise instructions (additional to the group exercise sessions) (64). Furthermore, the moderate physiotherapists' fidelity to the implementation activities, partly caused by limited resources, e.g., uncompensated level of high effort required for assessments, was challenging. The patients did not increase their individual exercise behaviour outside the group exercise sessions, which are usually quite long. Participants could have suffered from exhaustion since fatigue is a major barrier. Hilberdink and colleagues concluded that the implementation activities need to be adapted to increase feasibility and that the implementation strategy should be extended to all stakeholders, including patients.

The implementation projects in Switzerland (65) and the Netherlands (59) have pursued the same goal, but with different strategies. Interestingly, some of the lessons learned are very similar: the guite low adherence to individual (not-supervised) exercise by patients (according to the EULAR PA recommendations) and the relatively moderate fidelity of physiotherapists to the implementation activities remain challenging (Chapter 7, (59)). Even if the major stakeholders (patients, physiotherapists, patient organisations, rheumatologists) are aware of the importance of PA and are, theoretically, willing to change their behaviour or procedures, the road to real short- and long-term change is very long and difficult. Motivation is the key! This is, of course, not a new insight. Behavioural psychology has shown this to be the case (34, 66). All stakeholders need to improve their conceptual understanding of PA versus exercise and the skills to perform or instruct them, according to the EULAR PA recommendations (Chapter 4, (24, 67, 68)). External factors, such as time management or limited resources, were found to be restricting for physiotherapists (59). In addition, SVMB management reflects that physiotherapists might believe that these changes are unwanted by the group participants. This stands in contrast to the patients' needs for tailored exercise programmes and continuous individual support (Chapter 5, (69)). Patients want physiotherapists to take on the role of counsellor, supporting self-management collaboratively and with a positive patient-therapist relationship (70). Clear incentives and successful framing of the most important message "Be physically active" are crucial. Some public health initiatives follow the approach of financial incentives, such as a reduction in health insurance premiums if proof of regular training can be provided, but non-financial incentives, such as gamification and social incentives, are also considered motivating (71, 72).

# Implications for future research

The EULAR PA recommendations have been cited in 215 publications (6<sup>th</sup> March 2022, Web of Science Cited Reference function), of which 62 were reviews and 6 guidelines. Only two of these had the aim of implementing the EULAR PA recommendations into routine clinical practice (Chapter 7, (59)). This relatively low number highlights the need for a compulsory and standardised implementation agenda following research projects, which would be a catalyst for the clinical implementation of evidence. It should be noted, however, that not all implementation activities are accompanied by (published) research.

Since the implementation of the BeFit group exercise concept is an iterative process that has been considerably delayed due to the Covid-19 pandemic, some accompanying projects are currently ongoing, e.g., the development and usability study of an electronic exercise diary. This diary will be useful to better evaluate the patients' adherence to exercise. The pandemic has had a significant impact on group exercise therapy and a large effort will be required to reflate and keep BeFit alive. An unpublished statistic by the SVMB indicates that, due to the special conditions of the Covid-19 pandemic, only some 12% of its members regularly participated in group exercise therapy in 2021.

There is data to underpin the social importance of group exercise (Chapter 4, (24, 61)). Some people, however, prefer to exercise in groups not specifically designed for people with axSpA (Chapter 4), or to perform individual exercise, or to not exercise at all. In times of Covid regulations, remote care options must be considered. Therefore, the SVMB will need to apply different PA promotional concepts that target the majority of its members who are not participating in group exercise therapy. These can probably be divided into several subgroups, characterised by diverse needs with respect to exercise behaviour. Further research is needed to develop specific strategies for self-management and PA promotion interventions to target these subgroups of SVMB members, to attract the young, and keep the older people on board. In this process, the promising approach of technology-based counselling must be considered, which enables regular remote contact when a weekly group setting is inappropriate (22, 73). Although there is currently no available gold standard for the objective assessment of PA (21, 74), studies have provided evidence on the efficacy of electronic or mobile health (e/mHealth) approaches in PA promotion (75-77). Passalent and colleagues (78) support the idea that targeted e-health interventions, with the aims of integrating the patient perspective, addressing barriers, using existing knowledge and filling knowledge gaps, may promote PA uptake in people living with axSpA. In line with this, and in the face of the Covid-19 pandemic, the SVMB has launched an online portal (www. rheumafit.ch). Currently, there are different video-based exercise instructions available.

Further development activities are needed to realise a concept supporting individual PA promotion interventions. Additionally, PTs' fidelity to BeFit need to be further supported by addressing individual (e.g. increasing knowledge and skills) and organisational (touching upon perceived lack of time or organisational support) factors (79).

During the further national implementation of BeFit, a continuous evaluation of its implementation success will be mandatory to support adoption and avoid ineffective interventions. Furthermore, an economic analysis of BeFit should be performed to evaluate its cost-effectiveness. Findings of positive economic benefits could support negotiations with the health insurance authorities. The fact that the SVMB group exercise therapy is covered by health insurance annually is unique. In the Swiss health system, it is more common that only a proportion of group exercise costs are recompensed, or for a limited period. If this special status were to change, the BeFit concept would have to be adapted.

# Conclusion

This thesis adds value to the wealth of research on PA by presenting the complete journey from the development of the EULAR PA recommendations for people with RMDs (including axSpA), which state that exercise according to public health recommendations is effective, safe, and feasible, to the practical implementation of a concept for group exercise therapy for people with axSpA that conforms to these recommendations. The studies in this thesis were supported by international experts from the fields of rheumatology, health behaviour, and implementation science, as well as national stakeholders, such as the Swiss patient organisation, SVMB. The EULAR PA recommendations provide a solid foundation for further, urgently needed research in the field of PA for people with RMDs. This thesis shows that the current state of the non-pharmacological treatment of axSpA has potential for improvement. In addition, it gives valuable insights into some of the challenges associated with the initialisation and maintenance of change in the provision of exercise therapy for people with axSpA.

# References

- 1. Committee PAGA. 2018 Physical Activity Guidelines Advisory Commitee Scientific Report. Washington DC: U.S. Department of Health and Human Services; 2018.
- Millner JRB, J. S.; Beinke, K. M.; Butterworth, R. H.; Chasle, B. E.; Dutton, L. J.; Lewington, M. A.; Lim, E. G.; Morley, T. B.; O'Reilly, J. E.; Pickering, K. A.; Winzenberg, T.; Zochling, J. Exercise for ankylosing spondylitis: An evidence-based consensus statement. Seminars in arthritis and rheumatism. 2016;45(4):411-27.
- Fransen MM, Harmer AR, Van der Esch M, Simic M, Bennell KL. Exercise for osteoarthritis of the knee: a Cochrane systematic review. British journal of sports medicine. 2015;49(24):1554-7.
- Hu HL, Xu AQ, Gao C, Wang ZQ, Wu X. The effect of physical exercise on rheumatoid arthritis: An overview of systematic reviews and meta-analysis. Journal of Advanced Nursing. 2021;77(2):506-22.
- 5. World Health Organisation. Factsheet Physical Activity 2020 [cited December 2021] Available from: https://www.who.int/ news-room/fact-sheets/detail/physical-activity.
- Comission E. Horizon Europe Work Programme 2021-2022. 4. Health Brussels2021 [cited December 2021] Available from: https://ec.europa.eu/info/funding-tenders/opportunities/docs/2021-2027/horizon/wp-call/2021-2022/wp-4health\_horizon-2021-2022\_en.pdf.
- Rausch Osthoff AK, Niedermann K, Braun J, Adams J, Brodin N, Dagfinrud H, et al. 2018 EULAR recommendations for physical activity in people with inflammatory arthritis and osteoarthritis. Annals of the rheumatic diseases. 2018;77(9):1251-60.
- Millner JRB, Beinke KM, Butterworth RH, Chasle BE, Dutton LJ, Lewington MA, Lim EG, Morley TB, O'Reilly JE, Pickering KA, Winzenberg T, Zochling J. Exercise for ankylosing spondylitis: An evidence-based consensus statement. Seminars in arthritis and rheumatism. 2016;45(4):411-27.
- Magrey MN, Danve AS, Ermann J, Walsh JA. Recognizing Axial Spondyloarthritis: A Guide for Primary Care. Mayo Clin Proc. 2020;95(11):2499-508.
- 10. World Health Organisation. Handbook for good clinical research practice (GCP): Guidance for implementation. Geneve 2002.
- van der Heijde D, Aletaha D, Carmona L, Edwards CJ, Kvien TK, Kouloumas M, et al. 2014 Update of the EULAR standardised operating procedures for EULAR-endorsed recommendations. Annals of the rheumatic diseases. 2015;74(1):8-13.
- 12. Brouwers MC, Kho ME, Browman GP, Burgers JS, Cluzeau F, Feder G, et al. AGREE II: advancing guideline development, reporting, and evaluation in health care. Prev Med. 2010;51(5):421-4.
- Hoffmann TC, Glasziou PP, Boutron I, Milne R, Perera R, Moher D, et al. Better reporting of interventions: template for intervention description and replication (TIDieR) checklist and guide. BMJ (Clinical research ed). 2014;348:g1687.
- 14. Liguori G. ACSM's Guidelines for Exercise Training and Prescription. Eleventh Edition ed. Philadelphia: Wolters Kluwer; 2021.
- Regnaux JP, Davergne T, Palazzo C, Roren A, Rannou F, Boutron I, et al. Exercise programmes for ankylosing spondylitis. The Cochrane database of systematic reviews. 2019;10:CD011321.
- Sveaas SH, Bilberg A, Berg IJ, Provan SA, Rollefstad S, Semb AG, et al. High intensity exercise for 3 months reduces disease activity in axial spondyloarthritis (axSpA): a multicentre randomised trial of 100 patients. British journal of sports medicine. 2020;54(5):292.
- 17. Bilberg A, Sveaas SH, Dagfinrud H, Mannerkorpi K. How Do Patients With Axial Spondyloarthritis Experience High-Intensity Exercise? Acr Open Rheumatol. 2020;2(4):207-13.
- Bilberg A, Dagfinrud H, Sveaas SH. Supervised intensive Exercise strengthen Exercise Health Beliefs in Patients with Axial Spondyloarthritis: A Multicentre Randomized Controlled Trial. Arthritis care & research. 2022; 74(7):1196-1204
- 19. Azeez M, Clancy C, O'Dwyer T, Lahiff C, Wilson F, Cunnane G. Benefits of exercise in patients with rheumatoid arthritis: a randomized controlled trial of a patient-specific exercise programme. Clinical rheumatology. 2020;39(6):1783-92.
- Siqueira USOV, L. G.; de Mello, M. T.; Szejnfeld, V. L.; Pinheiro, M. M. Effectiveness of Aquatic Exercises in Women With Rheumatoid Arthritis: A Randomized, Controlled, 16-Week Intervention-The HydRA Trial. American journal of physical medicine & rehabilitation. 2017;96(3):167-75.
- Sekhon M, White C, Godfrey E, Amirova A, Revenas A, King S, et al. Effectiveness of web-based and mobile health interventions designed to enhance adherence to physical activity for people with inflammatory arthritis: a systematic review. Rheumatol Adv Pract. 2021;5(1):rkab016.
- Li LDC, Feehan LM, Xie H, Lu N, Shaw CD, Gromala D, et al. Efficacy of a Physical Activity Counseling Program With Use of a Wearable Tracker in People With Inflammatory Arthritis: A Randomized Controlled Trial. Arthritis care & research. 2020;72(12):1755-65.
- Kraus VB, Sprow K, Powell KE, Buchner D, Bloodgood B, Piercy K, et al. Effects of Physical Activity in Knee and Hip Osteoarthritis: A Systematic Umbrella Review. Medicine and science in sports and exercise. 2019;51(6):1324-39.
- Niedermann K, Nast I, Ciurea A, Vliet Vlieland T, van Bodegom-Vos L. Barriers and Facilitators of Vigorous Cardiorespiratory Training in Axial Spondyloarthritis: Surveys Among Patients, Physiotherapists, and Rheumatologists. Arthritis care & research. 2019;71(6):839-51.

- Boniface G, Gandhi V, Norris M, Williamson E, Kirtley S, O'Connell NE. A systematic review exploring the evidence reported to underpin exercise dose in clinical trials of rheumatoid arthritis. Rheumatology (Oxford, England). 2020;59(11):3147-57.
- 26. Balas EA, Boren SA. Managing Clinical Knowledge for Health Care Improvement. Yearb Med Inform. 2000(1):65-70.
- 27. De Geest S, Zuniga F, Brunkert T, Deschodt M, Zullig LL, Wyss K, et al. Powering Swiss health care for the future: implementation science to bridge "the valley of death". Swiss Med Wkly. 2020;150:w20323.
- Curran GM, Bauer M, Mittman B, Pyne JM, Stetler C. Effectiveness-implementation hybrid designs: combining elements of clinical effectiveness and implementation research to enhance public health impact. Med Care. 2012;50(3):217-26.
- 29. Landes SJ, McBain SA, Curran GM. Reprint of: An introduction to effectiveness-implementation hybrid designs. Psychiatry Res. 2020;283:112630.
- 30. Saba R, Bruderer-Hofstetter M, Rausch Osthoff AK, Niedermann K. Exercise Self-Efficacy Scale for Physical Activity in people with Ankylosing Spondylitis: A German Translation and Validation. physioscience. accepted, March 2022.
- Vogt U, Rausch Osthoff AK, Niedermann K. Physiotherapist's communication skills used during a exercise coaching in people with axial Spondyloarthritis. physioscience. Under review, March 2021.
- 32. Rausch Osthoff AK, Horvath C, Niedermann K. Feasibility of fitness assessments in group exercise therapy for individuals with Ankylosing Spondylitis. physioscience. submitted, Dec 2021.
- 33. Nilsen P. Making sense of implementation theories, models and frameworks. Implement Sci. 2015;10:53.
- 34. Grol R, Wensing M. Improving patient care. The implementation of change in health care. 2nd ed. Oxford: Wiley Blackwell; 2013.
- Hall A, Richmond H, Mahoney K, Matthews J. Changing Health-Related Behaviors 3: Lessons from Implementation Science. Methods Mol Biol. 2021;2249:571-95.
- Bor P, van Delft L, Valkenet K, Veenhof C. Perceived Factors of Influence on the Implementation of a Multidimensional Project to Improve Patients' Movement Behavior during Hospitalization: A Qualitative Study. Physical therapy. 2021; 102 (2):pzab260
- Butow P, Shaw J, Shepherd HL, Price M, Masya L, Kelly B, et al. Comparison of implementation strategies to influence adherence to the clinical pathway for screening, assessment and management of anxiety and depression in adult cancer patients (ADAPT CP): study protocol of a cluster randomised controlled trial. BMC Cancer. 2018;18(1):1077.
- Reynolds SS, Murray LL, McLennon SM, Ebright PR, Bakas T. Implementation Strategies to Improve Knowledge and Adherence to Spinal Cord Injury Guidelines. Rehabilitation nursing : the official journal of the Association of Rehabilitation Nurses. 2018;43(1):52-61.
- Koorts H, Eakin E, Estabrooks P, Timperio A, Salmon J, Bauman A. Implementation and scale up of population physical activity interventions for clinical and community settings: the PRACTIS guide. Int J Behav Nutr Phys Act. 2018;15(1):51.
- 40. Kennedy SG, Smith JJ, Estabrooks PA, Nathan N, Noetel M, Morgan PJ, et al. Evaluating the reach, effectiveness, adoption, implementation and maintenance of the Resistance Training for Teens program. Int J Behav Nutr Phys Act. 2021;18(1):122.
- Koorts H, Salmon J, Timperio A, Ball K, Macfarlane S, Lai SK, et al. Translatability of a Wearable Technology Intervention to Increase Adolescent Physical Activity: Mixed Methods Implementation Evaluation. Journal of medical Internet research. 2020;22(8):e13573.
- 42. Teychenne M, Apostolopoulos M, Ball K, Olander EK, Opie RS, Rosenbaum S, et al. Key stakeholder perspectives on the development and real-world implementation of a home-based physical activity program for mothers at risk of postnatal depression: a qualitative study. BMC public health. 2021;21(1):361.
- Study EV. The TRAPD method for survey translation 2018 [cited December 2021] Available from: https:// europeanvaluesstudy.eu/methodology-data-documentation/survey-2017/methodology/the-trapd-method-for-surveytranslation/.
- Beaton DE, Bombardier C, Guillemin F, Ferraz MB. Guidelines for the process of cross-cultural adaptation of self-report measures. Spine (Phila Pa 1976). 2000;25(24):3186-91.
- 45. Liem S, van Leeuwen NM, Vliet Vlieland T, Boerrigter G, van den Ende C, de Pundert L, et al. Physical therapy in patients with systemic sclerosis: physical therapists' perspectives on current delivery and educational needs. Scandinavian journal of rheumatology. 2021:1-8.
- 46. Liem SIE, van Leeuwen NM, Vliet Vlieland TPM, Boerrigter GMW, van den Ende CHM, de Pundert LAJ, et al. Physical therapy in systemic sclerosis: the patient perspective. Arthritis care & research. 2021; doi: 10.1002/acr.24741.
- Peter WF, Nelissen RG, Vlieland TPV. Guideline Recommendations for Post Acute Postoperative Physiotherapy in Total Hip and Knee Joint Replacement Surgery: Are They Used in Daily Clinical Practice? Osteoarthritis and cartilage. 2014;22:S456-S.
- Peter W, Tilbury C, Verdegaal S, Onstenk R, Vehmeijer SB, Vermeulen EM, et al. The provision of preoperative and postoperative physical therapy in elderly people with hip and knee osteoarthritis undergoing primary joint replacement surgery. Current Orthopaedic Practice. 2016;27(2):173-83.
- 49. Acocella I. The focus groups in social research: advantages and disadvantages. Qual Quant. 2012;46:1125-36.
- 50. Collier D, Mahoney J. Insights and pitfalls Selection bias in qualitative research. World Polit. 1996;49(1):56-91
- 51. Moser A, Korstjens I. Series: Practical guidance to qualitative research. Part 3: Sampling, data collection and analysis. European Journal of General Practice. 2017;24(1).
- 52. Guest G, Namey E, Taylor J, Eley N, McKenna K. Comparing focus groups and individual interviews: findings from a randomized study. Int J Soc Res Method. 2017;20(6):693-708.

- 53. Parker A, Tritter J. Focus group method and methodology: current practice and recent debate. Int J Research & Method in Education. 2006;29(1):23-37.
- 54. Dagfinrud H, Vollestad NK, Niedermann K, Kvien TK, Hagen KB. Exercise programs in trials for patients with ankylosing spondylitis: do they really have the potential for effectiveness? Arthritis care & research. 2011;63(4):597-603.
- 55. Bourban P, Tschopp M. Manuelle Leistungsdiagnostik. Messverfahren Kraft. Magglingen: Bundesamt für Sport; 2015.
- Niedermann K, Rausch Osthoff AK, Braun J, Becker H, Boehm P, Braem R, et al. Die laienverständliche Version der 2018 EULAR Empfehlungen zu körperlicher Aktivität von Menschen mit entzündlichrheumatischen und degenerativen Erkrankungen. Zeitschrift fur Rheumatologie. 2021, doi: 10.1007/s00393-021-01079-z
- 57. Kiltz U, Kiefer D, Braun J, Rausch Osthoff AK, Herbold S, Klinger M, et al. Übersetzung der 2018 EULAR Empfehlungen zu körperlicher Aktivität von Menschen mit entzündlich-rheumatischen und degenerativen Erkrankungen ins Deutsche und sprachliche Validierung im deutschsprachigen Raum mit medizinischen Fachpersonen. Z Rheumatol. 2021. doi: 10.1007/s00393-021-01078-0
- Metsios GS, Fenton SA, Moe HR, van der Esch M, van Zanten JV, Koutedakis Y, et al. Implementation of Physical Activity into routine Clinical pracTice in Rheumatic Musculoskeletal Disease: The IMPACT-RMD study protocol and rationale. Mediterr J Rheumatol. 2019;30(4):231-6.
- Hilberdink B, van der Giesen F, Vliet Vlieland T, van Bodegom-Vos L, Van Weely S. Implementing enhancements in supervised group exercise for people with axial spondyloarthritis: a hybrid effectiveness-implementation study. Scandinavian journal of rheumatology. 2021:1-9.
- Nikiphorou E, Santos EJF, Marques A, Bohm P, Bijlsma JW, Daien CI, et al. 2021 EULAR recommendations for the implementation of self-management strategies in patients with inflammatory arthritis. Annals of the rheumatic diseases. 2021;80(10):1278-85.
- 61. Hilberdink B, van der Giesen F, Vliet Vlieland T, Nijkamp M, van Weely S. How to optimize exercise behavior in axial spondyloarthritis? Results of an intervention mapping study. Patient education and counseling. 2020;103(5):952-9.
- Hilberdink B, Vlieland TV, van der Giesen F, van Gaalen F, Goekoop R, Peeters A, et al. Adequately dosed aerobic physical activity in people with axial spondyloarthritis: associations with physical therapy. Rheumatology international. 2020;40(9):1519-28.
- 63. Hilberdink B, van der Giesen F, Vliet Vlieland T, van Weely S. Organisation and content of supervised group exercise for people with axial spondyloarthritis in The Netherlands. Rheumatology international. 2021;41(2):391-401.
- Hilberdink B, van der Giesen F, Vliet Vlieland T, van Gaalen F, van Weely S. Supervised Group Exercise in Axial Spondyloarthritis: Patients' Satisfaction and Perspective on Evidence-Based Enhancements. Arthritis care & research. 2020;72(6):829-37.
- Rausch Osthoff AK, Vliet Vlieland TPM, Meichtry A, van Bodegom-Vos L, Topalidis B, Buchi S, et al. Lessons learned from a pilot implementation of physical activity recommendations in axial spondyloarthritis exercise group therapy. BMC Rheumatology. 2022;6(12).
- 66. Biddle SJH, Mutrie N. Psychology of Physical Activity. London: Routledge; 2001.
- 67. Astley C, Pinto AJ, Bonfa E, da Silva CAA, Gualano B. Gaps on rheumatologists' knowledge of physical activity. Clinical rheumatology. 2021;40(7):2907-11.
- Yona T, Ben Ami N, Azmon M, Weisman A, Keshet N. Physiotherapists lack knowledge of the WHO physical activity guidelines. A local or a global problem? Musculoskelet Sci Pract. 2019;43:70-5.
- O'Dwyer T, McGowan E, O'Shea F, Wilson F. Physical Activity and Exercise: Perspectives of Adults With Ankylosing Spondylitis. Journal of physical activity & health. 2016;13(5):504-13.
- Killingback C, Thompson M, Chipperfield S, Clark C, Williams J. Physiotherapists' views on their role in self-management approaches: A qualitative systematic review. Physiotherapy theory and practice. 2021. 5;1-15.
- Ball K, Hunter RF, Maple JL, Moodie M, Salmon J, Ong KL, et al. Can an incentive-based intervention increase physical activity and reduce sitting among adults? the ACHIEVE (Active Choices IncEntiVE) feasibility study. International Journal of Behavioral Nutrition and Physical Activity. 2017;14.
- 72. Patel MS, Small DS, Harrison JD, Fortunato MP, Oon AL, Rareshide CAL, et al. Effectiveness of Behaviorally Designed Gamification Interventions With Social Incentives for Increasing Physical Activity Among Overweight and Obese Adults Across the United States The STEP UP Randomized Clinical Trial. Jama Intern Med. 2019;179(12):1624-32.
- 73. Passalent LK, R; Lawson, D; Hawke, C; Omar, A; Thavaneswaran, A; Haroon, N; Inman, Rd. Impact of e-learning on knowledge, self-efficacy and exercise behaviours of patients with axial spondyloarthritis: results from a longitudinal randomized control trial. Arthritis and rheumatology Conference: american college of rheumatology/association of rheumatology health professionals annual scientific meeting, ACR/ARHP 2017; 68:[1495-6 pp.]. Available from: http:// onlinelibrary.wiley.com/o/cochrane/clcentral/articles/246/CN-01293246/frame.html.
- Carlin T, Soulard J, Aubourg T, Knitza J, Vuillerme N. Objective Measurements of Physical Activity and Sedentary Behavior Using Wearable Devices in Patients With Axial Spondyloarthritis: Protocol for a Systematic Review. JMIR research protocols. 2021;10(11):e23359.
- Yerrakalva D, Yerrakalva D, Hajna S, Griffin S. Effects of Mobile Health App Interventions on Sedentary Time, Physical Activity, and Fitness in Older Adults: Systematic Review and Meta-Analysis. Journal of medical Internet research. 2019;21(11).
- Brickwood KJ, Watson G, O'Brien J, Williams AD. Consumer-Based Wearable Activity Trackers Increase Physical Activity Participation: Systematic Review and Meta-Analysis. JMIR Mhealth Uhealth. 2019;7(4):e11819.

- Xie Z, Jo A, Hong YR. Electronic wearable device and physical activity among US adults: An analysis of 2019 HINTS data. Int J Med Inform. 2020;144:104297.
- Passalent L, Cyr A, Juristica I, Mathur S, Inman RD, Haroon N. Motivators, barriers, and opportunity for e-health to encourage physical activity in axial spondyloarthritis: a qualitative descriptive study Arthritis care & research. 2022; 74(1):50-58
- 79. Rethorn ZD, Covington JK, Cook CE, Bezner JR. Physical Therapists' Knowledge, Skills, Beliefs, and Organizations Impact Physical Activity Promotion: A Systematic Review and Meta-Analysis. Physical therapy. 2022;102(3).

# 9

Summary

Exercise that is correctly dosed and wisely chosen is medicine for people with rheumatic and musculoskeletal diseases (RMDs), such as inflammatory arthritis and osteoarthritis. Even though the available guidelines and recommendations on disease management of RMDs in general stressed the importance of exercise, they were not sufficiently precise for clinical use and did not consider the public health recommendations for health-enhancing physical activity (PA). The current general PA recommendations for adults to gain health benefits and to reduce the risk of cardiovascular diseases include at least 150-300 minutes of moderate-intensity or 75-150 minutes of vigorous-intensity cardiorespiratory training per week. According to general PA recommendations, there are four domains of activities: cardiorespiratory/aerobic, strength, flexibility and neuromotor performance. Exercise is a subcategory of PA that is planned, structured, repetitive and has the objective to maintain or improve physical fitness.

PA behaviour is complex and even more so in people with RMDs. In addition to the challenges confronting the general population in living an active PA behaviour, people with RMDs, such as axial Spondyloarthritis (axSpA), face disease-specific barriers. In addition, they may have an increased risk of cardiovascular diseases.

This thesis reports on the development of PA recommendations for people with RMDs (including axSpA), as well as their implementation into a group exercise therapy concept specifically for people with axSpA in Switzerland.

**Chapter 1** provides a general introduction into the content and context of this thesis. Firstly, it presents an overview of the definition, epidemiology, and management of axSpA and the definition of and recommendations for PA and exercise in people with RMDs. Secondly, the literature on the importance and challenges of PA promotion in people with axSpA is summarised. Thirdly, the Swiss Health System and the Swiss patient organisation 'Ankylosing Spondylitis Association Switzerland' (Schweizerische Vereinigung Morbus Bechterew, SVMB), which offers group exercise therapy for people with axSpA, were introduced. Based on the conclusions from this overview, the two goals of this thesis were formulated. These were to:

- 1) Describe the development, definition, and evidence base of the 2018 EULAR recommendations for PA in people with inflammatory arthritis and osteoarthritis.
- 2) Detail the pilot implementation of exercise groups for people with axSpA in Switzerland, based on the EULAR PA recommendations. This includes an analysis of the situation, exploration of the determinants of PA behaviour, and the evaluation of a strength assessment.

# Main Findings of this thesis

In Chapter 2, the development of the EULAR (European Alliance of Associations for Rheumatology) recommendations for PA in people with RMDs (including axSpA) is described. Chapter 3 elaborates on the evidence base for these PA recommendations by means of a systematic review and meta-analysis.

Chapter 2 outlines the process of developing and defining PA recommendations for people with inflammatory arthritis and osteoarthritis, according to predefined EULAR standardised operating procedures. A task force comprised of members from 16 countries (including rheumatologists, other medical specialists and physicians, health professionals, patientrepresentatives, and methodologists) met twice. At the first meeting, 13 research questions were identified and defined. Following the meeting, a systematic literature review was performed to answer the 13 research questions. The evidence was presented and discussed at the second meeting. The evidence and expert opinion informed the definition of four overarching principles and ten recommendations. Given the evidence for its effectiveness, feasibility and safety, PA, guided by public health recommendations, was advocated as an integral part of standard care throughout the course of inflammatory arthritis and osteoarthritis. Nevertheless, it was found that general and disease-specific contraindications and barriers, as well as individual adaptations based on assessments of physical, social, and psychological factors need to be considered. Additional recommendations were defined, e.g., that individual exercise goals should be defined by shared decision and that the health care professionals providing PA interventions should apply behaviour change techniques (BCTs). The mean level of agreement on the overarching principles and recommendations (scale 0-10) ranged from 8.8 - 9.8 among the task force members.

Finally, the task force agreed on a related and comprehensive research and education agenda. It also formulated the ambition to implement the EULAR PA recommendations into clinical routine, using implementation strategies targeted at specific groups and taking into account the differences between health systems across Europe.

**Chapter 3** addresses the effectiveness of PA interventions meeting the requirements of the public health recommendations for PA in people with inflammatory arthritis (rheumatoid arthritis /RA and spondyloarthritis /SpA) and osteoarthritis (hip and knee /HOA and KOA). The systematic review included 63 randomized controlled trials (RCTs) in adults with RA, SpA, or H/KOA, investigating the effect of either: a) aerobic exercises on cardiorespiratory fitness; b) muscle strength exercise on lower limb muscle strength; c) flexibility exercise on flexibility; d) neuromotor exercise on neuromotor performance; or e) PA promotion (based on behaviour change techniques/BCTs) on the amount of daily PA. Studies were only included when the content and dosage of intervention was in accordance with the public health recommendations for PA, in particular the exercise principles advocated by the American College of Sports Medicine (ACSM). Data from 49 RCTs were pooled in a meta-

analysis, using a random-effect model, and presented as standardised mean difference (SMD). Moderate effects were found for aerobic exercises on cardiorespiratory fitness (SMD 0.56 (95% CI 0.38 to 0.75)) and for resistance training on muscle strength (SMD 0.54 (95% CI 0.35 to 0.72)). However, no effects of combined strength/aerobic/flexibility exercises on flexibility (SMD 0.12 (95% CI -0.16 to 0.41)) were seen. PA promotion interventions led to a small increase in PA behaviour (SMD 0.21 (95% CI 0.03 to 0.38)). The conclusions were that a) exercising according to ACSM principles has a moderate effectiveness on cardiorespiratory fitness and muscle strength, and b) PA promotion based on behaviour change techniques had a small effect on PA behaviour in people with RA, SpA, or H/KOA. No evidence was found for the effects of flexibility exercises and very little literature was available that evaluated neuromotor exercises.

Many studies had to be excluded due to poor reporting of the interventions. It was suggested that future studies should apply appropriate reporting guidelines for intervention studies, which will enable clinicians and researchers to better understand, compare and replicate findings.

The EULAR PA recommendations provide guidance for the development, conduct and evaluation of interventions on PA, exercise, and promotion of an active lifestyle in people with inflammatory arthritis and osteoarthritis. Due to the differences in health systems across Europe, culturally specific implementation strategies for specific target groups involving all stakeholders are needed. Chapters 4 to 7 describe the preparation, development, and evaluation of an implementation strategy for exercising according to EULAR PA recommendations in the group exercise setting for people with axSpA in Switzerland.

In Chapter 4 the findings of a cross-sectional survey, conducted among people with axSpA living in the western region of the Netherlands and the German-speaking part of Switzerland, are presented. The aim of the survey was to evaluate and compare the usage, experiences, and preferences regarding the delivery of individual and group exercise physiotherapy. In both countries, supervised exercise therapy was offered on an individual or group basis. The latter was usually water-based and/or land-based, offered once a week, supervised by a physiotherapist, and included an important social factor. In total, 713 (206 Dutch, 507 Swiss) individuals with axSpA participated (56% male, median age 55 years). Of these participants, 83% were using or had used physiotherapy in group and/or individual setting. Of these, individual therapy alone was used or had been used by 36%, a combination of individual plus land-based or water-based group therapy by 29%, and group therapy alone by only 5%. Less than half the participants attending individual therapy reported active therapy (such as aerobic, muscle strength and flexibility exercises). Although the majority (76%) was not aware of the increased cardiovascular risk, participants showed an interest in aerobic exercises, either in an individual or supervised setting. If supervised, the majority would prefer supervision by a specialised physiotherapist.
The findings of this survey suggested that there is a need to implement active exercises at an appropriate dose according to current exercise recommendations, particularly with a higher focus on aerobic exercises in an individual or group setting. Enabling people with axSpA to perform exercises independently (unsupervised) would most likely meet their needs and could enhance their daily PA.

Chapter 5 summarises a qualitative study in people with axSpA aiming at exploring behavioural, normative and control beliefs concerning general PA and cardiorespiratory training. Five semi-structured focus groups with 24 individuals living with axSpA were performed and data were analysed using structured thematic qualitative content analysis. Data were categorized into behavioural, normative and control beliefs. Behavioural beliefs revealed a positive attitude towards general PA, with participants mentioning numerous physical, psychological, and social benefits, but only few risks. However, the conceptual difference between general PA and cardiorespiratory training, and the relevance of cardiorespiratory training to reducing the increased risk of cardiovascular disease, was unclear to some participants. Normative beliefs were expressed as the beliefs of significant others that influenced their motivation to comply with such beliefs. Significant others can be spouses, other people living with axSpA, or rheumatologists. Regarding control beliefs, general PA and cardiorespiratory training were both mentioned as effective self-management strategies to control the disease. From experience, a high level of self-discipline, as well as the use of technology, were shown to be useful. Physiotherapists play a key role in PA promotion and could influence normative beliefs. This potential could be used to a greater extent in the future to promote active lifestyle competencies in people living with axSpA.

Chapter 6 focuses on the assessment that is needed to develop an exercise programme tailored to an individual axSpA patient's capabilities. Core muscles are particularly important for this group of patients, since inflammation and reduced mobility affect the dynamic stabilisation of the spine. A certain level of core stability is needed, in terms of strength and muscle fatigue resistance, for the activities of daily living and sport performance. However, the assessment of core strength is challenging and no appropriate assessment for people with axSpA in a group exercise therapy setting existed. The aim of this study was to adapt the isometric Core Strength Endurance test battery (aCSE), previously used for athletes, for use in people with axSpA and to evaluate the intra-tester reliability and its associations with the disease-specific factors of functional status, self-reported pain, and perceived strength performance. A cross-sectional study was conducted in exercise therapy groups, including 62 people with axSpA and 13 physiotherapist group leaders. The aCSE was repeated after 7-14 days to measure intra-tester reliability of the same rater (PT group leader). Reliability was calculated by means of computing the intraclass correlation coefficient (ICC), using a nested design. A moderate to substantial intra-rater reliability (ICCs (95%CI)) was found for the ventral (0.54 (0.35, 0.74)), lateral (0.52 (0.33, 0.70)) and dorsal (0.71 (0.58, 0.86)) core muscle chains. The associations between ventral, lateral and dorsal strength endurance and the disease-specific factors were calculated using Pearson correlation coefficients.

It was concluded that the aCSE was a feasible and reliable assessment instrument to measure core strength in people with axSpA. It can be combined with other appropriate assessments, covering the cardiorespiratory fitness, (spinal) mobility and neuromotor performance domains, to establish an inexpensive, practical, yet comprehensive, assessment battery for the assessment of individuals with axSpA in the group exercise setting. However, none of the aCSE measures correlated significantly with the disease-specific factors.

The aim of the SVMB was to revise their group exercise therapy concept according to stateof-the-art practices. Therefore, the previously described 2018 EULAR recommendations for PA in people with inflammatory arthritis and osteoarthritis were translated into a new group exercise therapy concept, called "BeFit", by executing five key implementation activities: a) training for the supervising physiotherapists; b) correctly dosed exercises in all fitness dimensions; c) exercise counselling; d) bi-annual fitness assessments; and e) individual exercise training (in addition to group exercise). By applying this concept to group exercise therapy, the programme not only meets the requirements of disease-specific exercises, but also accords to public health PA recommendations.

**Chapter 7** details the development of a tailored implementation strategy for BeFit and its pilot testing in four exercise groups, consisting of 30 people with axSpA and four physiotherapists. After six months of testing, the implementation success was evaluated at the levels of patient, physiotherapist, and the organisation. At the patient level, the primary outcome was adherence to the recommended exercise behaviour, assessed by the number of reported training sessions per exercise dimension in the electronic exercise diary over 6 months. The secondary outcomes were feasibility and satisfaction with the new exercise group concept, rated on a 0-10 numeric rating scale. The tertiary outcome, to evaluate the effectiveness of the individualised PA recommendations, was the measured change in the fitness levels of patients, assessed by fitness parameters. At the level of physiotherapist, the primary outcome was fidelity to the programme design concept, assessed using a diary. The secondary outcomes were treatment quality, feasibility, and satisfaction, evaluated with semi-structures interviews. Acceptance and the establishment of BeFit within the organisational structure were evaluated at the organisational level. The patients' self-reported adherence to the PA recommendations was found to be insufficient (43%), possibly due to technical problems with the electronic diary. The physiotherapists' fidelity to the new exercise group concept was reported as satisfactory. On all levels, the new concept was generally perceived as feasible and useful for supporting individual exercise. The suggested frequency of exercise counselling (quarterly) and the fitness assessments (bi-annually) were found to be too high and too rigid by both patients and physiotherapists. Overall, the pilot implementation of PA recommendations showed sufficient acceptance and satisfaction. Nevertheless, it appeared that some adaptations are needed to enhance its feasibility for nationwide implementation. These adaptations were described in detail to make the knowledge available for future implementation research projects. During the national implementation of BeFit, a continuous monitoring will be needed to avoid ineffective elements of the concept.

#### **General Discussion**

The 2018 EULAR recommendations for PA in people with inflammatory arthritis and osteoarthritis and the underlying literature review and meta-analysis, strengthen the evidence base for non-pharmacological care in RMDs. Based on strong evidence (Chapter 3), exercising according to the general PA recommendations covering the four exercise dimensions of aerobic, muscle strength, flexibility and neuromotor exercising, were acknowledged to be effective, safe and feasible for people with RMDs (Chapter 2). The EULAR PA recommendations highlight important considerations that should be followed in the context of PA interventions for people with RMDs. People with RMDs may face several disease-specific challenges, requiring adaptation according to individual needs, capabilities, and resources. Thus, it remains to be established whether further refinements are needed for their use in the clinical routine, in the form of additional disease-specific recommendations. Furthermore, there are still some relevant knowledge gaps, such as: the most effective intervention strategies for long term PA promotion; the feasibility of exercise interventions covering all four exercise dimensions; the most appropriate, valid and preferable objective PA assessment for clinical use. Fortunately, PA interventions and PA behaviour are highly researched fields and, by joining forces and good alignment, further steps can be made in the future.

The implementation of the EULAR PA recommendations into clinical routine demands a structured and theory-driven approach. This thesis described the translation of the EULAR PA recommendations into the group exercise therapy concept BeFit, together with its pilot implementation, based on Grol&Wensings' model of change and using a hybrid design type 3. Three studies (Chapters 4-6) were described that were designed to analyse the current situation and support the development process of the implementation activities. Strengths of this study were the inclusion of an implementation expert in the study staff and the in-depth contextual analysis. The relevant stakeholders, such as people with axSpA, physiotherapists, and the SVMB organisation were involved in this process. If we had used a different implementation approach to Grol&Wensing's model of change [1], e.g., the Practical planning for Implementation and Scale-up (PRACTIS) approach [2] that was designed specifically for the planning and scaling-up of PA interventions, the collaboration of stakeholders might have been even greater in the early phase. This could have had an additional positive impact on the acceptance of BeFit. Even though the physiotherapists and most participants in the pilot groups were interested in the new concept and willing to adapt their usual care, other exercise therapy groups may not be such "innovators" and not see the value of change. One important change for the physiotherapists was their new role as PA counsellor. Even though PA promotion is a key concept of physiotherapy [3], some physiotherapists might not yet have the necessary communication skills, behaviour change techniques, or detailed knowledge of PA recommendations to embed it successfully into their practice.

The focus group study (Chapter 5) on behavioural, normative and control beliefs concerning general PA and cardiorespiratory training in people with axSpA showed the relevance of PA as a self-management strategy. However, the conceptual difference between general PA and cardiorespiratory training was unclear to some participants. It was concluded that, because of the increased risk of cardiovascular diseases in people with axSpA of which most of the survey participants (Chapter 4) were unaware, patients and physiotherapists need to understand the difference between the concepts and also how to perform cardiorespiratory training. The survey (Chapter 4) showed that most people with axSpA use physiotherapy in an individual and/or a group setting. Therefore, physiotherapists are in a strong position to strengthen the individual's skills in using exercise as a self-management strategy. Exercise should be planned according to individual goals that should be re-evaluated regularly using (preferably objective) assessments. Chapter 6 showed a reliable, easy-to-perform method of how to assess core strength in people with axSpA. As such, it contributes to the desired, regularly performed comprehensive assessment of every individual with axSpA as part of an exercise programme. Only by means of an appropriate assessment can exercise and PA be tailored to the individual's health status and personal goals.

#### Conclusion

This thesis adds value to the wealth of research on PA by presenting the complete journey from the development of the EULAR PA recommendations for people with RMDs (including axSpA), which state that exercise according to public health recommendations is effective, safe, and feasible, to the practical implementation of a concept for group exercise therapy for people with axSpA that conforms to these recommendations. The EULAR PA recommendations provide a solid foundation for further research. Additionally, this thesis gives insights into some of the challenges associated with the initialisation and maintenance of change in the provision of exercise therapy for people with axSpA.

# References

- 1. Grol, R. and M. Wensing, Improving patient care. The implementation of change in health care. 2nd ed. 2013, Oxford: Wiley Blackwell.
- Koorts, H., et al., Implementation and scale up of population physical activity interventions for clinical and community settings: the PRACTIS guide. Int J Behav Nutr Phys Act, 2018. 15(1): p. 51.
- World\_Physiotherapy, Physical therapists as exercise and physical activity experts across the life span. Policy statement, in https://world.physio/sites/default/files/2020-04/PS-2019-Exercise-experts.pdf. 2019, World Confederation for Physical Therapy.

# 10

Nederlandse samenvatting

Lichaamsbeweging, mits goed uitgevoerd en juist gedoseerd, werkt als een medicijn bij mensen met reumatische en musculoskeletale aandoeningen (Rheumatic and Musculoskeletal Diseases, RMDs). Hoewel de tot voor kort beschikbare richtlijnen en aanbevelingen voor de behandeling van mensen met RMDs wel het belang van bewegen in het algemeen benadrukten, waren ze niet erg specifiek met betrekking tot de praktische toepassing en namen ze de beweegrichtlijnen die voor de algemene bevolking gelden niet mee. Volgens de beweegrichtlijnen voor de algemene bevolking moeten volwassenen minimaal 150-300 minuten met matig intensieve, of 75-150 minuten met hoog intensieve intensiteit per week bewegen om hun gezondheid te verbeteren en de kans op cardiovasculaire aandoeningen te verkleinen. In deze beweegrichtlijnen worden naast bovengenoemde cardiorespiratoire/ aerobe beweegvormen drie andere beweegvormen onderscheiden, te weten activiteiten om de spierkracht, flexibiliteit en neuromotorische ofwel functionele capaciteiten en/of balans te verbeteren.

Bij het spreken over lichaamsbeweging moet een onderscheid gemaakt tussen oefenen en bewegen. Bewegen is elke vorm van lichamelijke activiteit waarbij energie wordt verbrand. Oefenen of trainen is een subcategorie van bewegen, waarbij het gaat om geplande, gestructureerde, en herhaalde bewegingen die tot doel hebben de fysieke fitheid in stand te houden of te verbeteren.

Beweeggedrag is complex en dit is des te meer het geval bij mensen met RMDs. Voor gezonde mensen is het vaak al lastig om een actief leven te leiden, maar voor mensen met RMDs, zoals axiale spondyloartritis (axSpA), is het ten gevolge van klachten die gepaard gaan met hun aandoening zoals pijn, stijfheid of vermoeidheid, vaak extra moeilijk. Toch is het voor hen uitermate belangrijk, niet alleen omdat ze door middel van lichamelijke activiteit hun klachten kunnen verminderen en algehele gezondheid kunnen verbeteren, maar ook omdat bij mensen met RMDs het risico op cardiovasculaire aandoeningen verhoogd is. Dit proefschrift beschrijft de ontwikkeling van een beweegrichtlijn specifiek voor mensen met RMDs (waaronder ook axSpA) en de integratie en implementatie van deze richtlijn in een concept voor groepsoefentherapie specifiek voor mensen met axSpA in Zwitserland.

**Hoofdstuk 1** geeft een algemene inleiding op het onderwerp en schetst de context waarbinnen dit proefschrift tot stand is gekomen. Het hoofdstuk begint met de definitie van axSpA en een beschrijving van de epidemiologie van dit ziektebeeld, alsmede van de behandeling. Daarnaast worden de termen lichamelijke activiteit ofwel lichaamsbeweging en oefenen gedefinieerd en worden bestaande beweeg- en oefenrichtlijnen uiteengezet. Het hoofdstuk geeft vervolgens een overzicht van de literatuur over het belang van en de uitdagingen op het gebied van bevordering van lichamelijke activiteit van mensen met axSpA. Het hoofdstuk sluit af met de introductie van het Zwitserse gezondheidszorgstelsel en de Zwitserse patiëntenorganisatie voor axSpA, de Schweizerische Vereinigung Morbus Bechterew (SVMB), die een belangrijke rol speelt in verschillende studies beschreven in dit proefschrift. De SVMB verzorgt oefenprogramma's in groepsverband voor mensen met axSpA.

Hoofdstuk 1 wordt afgesloten met de volgende twee algemene doelen van dit proefschrift:

- Het beschrijven van de empirische basis van de ontwikkeling en de formulering van een beweegrichtlijn specifiek voor mensen met inflammatoire artritis en artrose op initiatief van de "European Alliance of Associations for Rheumatology" (EULAR) in 2018.
- 2) Het toepassen en evalueren van de principes van deze beweegrichtlijn binnen de groepsoefentherapie voor mensen met axSpA in Zwitserland. Deze doelstelling omvatte naast een pilotstudie ook een analyse van de huidige zorgverlening, een exploratie van bevorderende en belemmerende factoren voor adequaat oefen- en beweeggedrag en een evaluatie van een methode om rompspierkracht te meten.

# De belangrijkste bevindingen van het onderzoek in dit proefschrift

In Hoofdstuk 2 wordt de ontwikkeling van de EULAR aanbevelingen voor lichamelijke activiteit van mensen met RMDs (inclusief axSpA) beschreven. Hoofdstuk 3 gaat dieper in op de empirische basis van deze richtlijn aan de hand van een systematische review en een meta-analyse van de relevante literatuur.

In **Hoofdstuk 2** wordt beschreven hoe de aanbevelingen voor lichamelijke activiteit van mensen met inflammatoire artritis en artrose werden ontwikkeld en geformuleerd volgens de bestaande "standard operating procedures" van de EULAR. Een taskforce met leden uit 16 verschillende landen (onder wie reumatologen, andere medisch specialisten en artsen, health professionals, patiëntvertegenwoordigers en methodologen) kwam in het kader van de richtlijnontwikkeling twee keer bijeen. Tijdens de eerste bijeenkomst zijn 13 onderzoeksvragen geformuleerd. Vervolgens zijn deze vragen beantwoord door middel van systematisch literatuuronderzoek. Tijdens de tweede bijeenkomst werden de uitkomsten van het literatuuronderzoek gepresenteerd en besproken en deze vormden, aangevuld met de expertise van de taskforce, de basis voor de uiteindelijke formulering van vier overkoepelende principes en tien aanbevelingen.

Omdat er voldoende bewijs is dat lichamelijke activiteit volgens de beweegrichtlijnen voor de algemene bevolking voor mensen met RMDs effectief, haalbaar en veilig is, werd aanbevolen om lichamelijke activiteit integraal deel uit te laten maken van de reguliere zorg voor mensen met inflammatoire artritis en artrose. Belangrijk is wel dat daarbij rekening moet worden gehouden met ziektespecifieke en algemene contra-indicaties en belemmerende factoren. Daartoe moeten, op basis van een uitgebreide assessment van lichamelijke, sociale, en psychologische factoren zo nodig individuele aanpassingen aan het individuele beweegplan gemaakt worden.

Andere aanbevelingen hadden onder andere betrekking op de noodzaak van het vaststellen

van individuele doelen op het gebied van oefenen en bewegen, en de toepassing van gedragsveranderingstechnieken door zorgprofessionals die beweeginterventies aanbieden. De gemiddelde score van de taskforceleden over de mate waarin zij het eens waren met de overkoepelende principes en aanbevelingen varieerde tussen de 8,8 en 9,8 op een schaal van 0 tot 10.

Tot slot stelde de taskforce agenda's op voor toekomstig onderzoek en onderwijs. De taskforce heeft ook de ambitie uitgesproken om de richtlijn te implementeren in de klinische praktijk met behulp van strategieën gericht op specifieke doelgroepen, rekening houdend met de verschillen tussen de gezondheidszorgstelsels binnen Europa.

In **Hoofdstuk 3** wordt een literatuuronderzoek beschreven naar de effectiviteit van beweeginterventies die voldoen aan de beweegrichtlijnen voor de algemene bevolking, bij mensen met inflammatoire artritis (reumatoïde artritis /RA en spondyloartritis /SpA) en artrose (heup en knie artrose /HKOA). In het literatuuronderzoek werden 63 gerandomiseerde, gecontroleerde onderzoeken (RCT's) bij volwassenen met RA, SpA, of HKOA betrokken, waarbij werd gekeken naar het effect van: a) cardiorespiratoire/aerobe oefeningen; b) spierkrachtoefeningen; c) flexibiliteitsoefeningen; d) neuromotore oefeningen ofwel functionele- of balansoefeningen; en e) bevordering van algehele lichamelijke activiteit door middel van gedragsveranderingstechnieken. Onderzoeken werden alleen meegenomen in de review als de inhoud en dosering van de interventie in overeenstemming waren met de beweegrichtlijnen voor de algemene bevolking, met name met de trainingsprincipes van de American College of Sports Medicine (ACSM).

De gegevens van 49 RCT's werden meegenomen in een meta-analyse en geanalyseerd met een random-effects model, waarbij de uitkomsten werden uitgedrukt als gestandaardiseerd gemiddeld verschil (Standardized Mean Difference, SMD). De uitkomsten van de meta-analyse lieten een matig groot effect zien van aerobe oefeningen op de cardiorespiratoire fitness (SMD 0,56 (95% Betrouwbaarheidsinterval (BI) 0,38-0,75)) en van (spier)krachttraining op spierkracht (SMD 0,54 (95% BI 0,35-0,72)), maar geen effect van gecombineerde kracht-/ aerobe-/flexibiliteitsoefeningen op flexibiliteit (SMD 0,12 (95% BI -0,16-0,41)). Interventies ter bevordering van algehele lichamelijke activiteit leidden tot een kleine toename van lichamelijke activiteit (SMD 0,21 (95% BI 0,03-0,38)).

Opbasisvandemeta-analysewerdendevolgendeconclusiesgetrokken: a) oefeningenvolgens ACSM-principes hebben een matig groot effect op cardiorespiratoire fitness en spierkracht; en b) het bevorderen van bewegen met behulp van gedragsveranderingstechnieken geeft een kleine verbetering van beweeggedrag bij mensen met RA, SpA, en K/HOA. Er werd geen bewijs gevonden voor de effectiviteit van flexibiliteitsoefeningen en er was te weinig literatuur beschikbaar over de effectiviteit van neuromotore oefeningen om daarover een conclusie te kunnen trekken.

Veel onderzoeken konden niet worden meegenomen in de meta-analyse omdat de interventies niet duidelijk genoeg waren beschreven. Om hier verbetering in te brengen werd geadviseerd om bij toekomstig onderzoek de richtlijnen voor de beschrijving van interventieonderzoeken vaker en beter toe te passen, zodat de bevindingen van onderzoeken beter kunnen worden geïnterpreteerd, vergeleken en gereproduceerd.

De EULAR aanbevelingen voor lichamelijke activiteit van mensen met RMDs zijn van toepassing op de ontwikkeling, het aanbod en de evaluatie van interventies gericht op bewegen, oefenen, en bevordering van een actieve leefstijl bij mensen met inflammatoire artritis en artrose. Vanwege de verschillen tussen de Europese gezondheidszorgstelsels moeten implementatiestrategieën op nationaal niveau worden ontwikkeld. Daarbij is het belangrijk dat steeds alle stakeholders worden betrokken. In de hoofdstukken 4 t/m 7 wordt de ontwikkeling en evaluatie van een planmatige implementatiestrategie voor de concepten van de EULAR aanbevelingen voor lichamelijke activiteit binnen de groepsoefentherapie voor mensen met axSpA in Zwitserland beschreven.

In **Hoofdstuk 4** worden de bevindingen van een cross-sectioneel vragenlijstonderzoek gepresenteerd. Het onderzoek werd uitgevoerd bij mensen met axSpA in het westen van Nederland en in het Duitssprekende deel van Zwitserland. Het doel van dit onderzoek was het evalueren en vergelijken van het gebruik, de ervaringen en de voorkeuren ten aanzien van fysio/oefentherapie. In beide landen wordt zowel gesuperviseerde individuele fysio/ oefentherapie als groepsoefentherapie aangeboden. Groepsoefentherapie vindt in beide landen meestal één keer per week plaats, in het water en/of op het droge, onder toezicht van een fysio- of oefentherapeut, en bevat ook een belangrijke sociale component.

In totaal namen 713 individuen met axSpA (206 Nederlanders en 507 Zwitsers; 56% man; mediane leeftijd 55 jaar) deel aan dit onderzoek. Van deze groep maakte 83% op dat moment of eerder gebruik van fysiotherapie in groeps- en/of individuele setting; Hiervan 36% kreeg individuele therapie of had dit eerder gekregen; 29% kreeg een combinatie van individuele en groepsoefentherapie op het land of op het droge; slechts 5% volgde uitsluitend groepsoefentherapie. Minder dan de helft van de deelnemers aan individuele therapie meldde dat de therapie bestond uit het doen van actieve oefeningen (zoals aerobe oefeningen, spierkracht- en flexibiliteitsoefeningen).

Hoewel de meerderheid (76%) zich niet bewust was van het verhoogde risico op cardiovasculaire aandoeningen bij axSpA, toonden de deelnemers belangstelling voor adequaat gedoseerde cardiorespiratoire training, zelfstandig of gesuperviseerd. Bij gesuperviseerde oefentherapie werd door de meerderheid van de respondenten de voorkeur gegeven aan een fysiotherapeut met specifieke deskundigheid op het gebied van reumatische aandoeningen.

De bevindingen van dit onderzoek suggereren dat het nodig is om goed gedoseerd actieve training, met name bestaande uit cardiorespiratoire oefeningen te integreren in het behandelbeleid, zowel bij individuele als groepsoefentherapie. Een deel van de mensen met axSpA zal met de juiste instructie in staat zijn om zelfstandig (zonder supervisie) te oefenen en bewegen volgens beweegrichtlijnen.

**Hoofdstuk 5** geeft een overzicht van een in het kader van dit proefschrift uitgevoerd kwalitatief onderzoek onder mensen met axSpA, dat ten doel had hun 'beliefs' ten aanzien van bewegen en cardiorespiratoire beweegactiviteiten te exploreren. Daartoe werden vijf semigestructureerde focusgroepen georganiseerd, waaraan in totaal 24 mensen met axSpA deelnamen. De uitkomsten werden geanalyseerd met een gestructureerde, thematische kwalitatieve inhoudsanalyse.

De analyse van de zogenaamde 'behavioural beliefs' liet zien dat de deelnemers een positieve houding hadden ten aanzien van bewegen: deelnemers noemden veel lichamelijke, psychologische en sociale voordelen, en maar weinig risico's van bewegen. Het conceptuele verschil tussen bewegen in het algemeen en cardiorespiratoire training, en hoe cardiorespiratoire training de verhoogde kans op cardiovasculaire aandoeningen kan verlagen, was sommige deelnemers echter niet duidelijk. De genoemde 'normative beliefs' omvatten de overtuigingen van relevante anderen (partner, andere mensen met axSpA en reumatologen) die van invloed kunnen zijn op het gedrag. Ten aanzien van 'control beliefs' werden bewegen in het algemeen en cardiorespiratoire training beide genoemd als effectieve zelfmanagementstrategieën om met de ziekte om te gaan. De ervaring leert dat het hebben van veel zelfdiscipline en het ter beschikking hebben van technische hulpmiddelen zoals wearables nuttig kunnen zijn bij het bevorderen en behouden van gezond beweeggedrag. Daarnaast spelen fysiotherapeuten hierin een belangrijke rol, waarbij zij onder andere invloed kunnen uitoefenen op de eerdergenoemde 'normative beliefs'. In de toekomst zouden fysiotherapeuten nog vaker kunnen worden ingezet om mensen met axSpA te stimuleren hun mogelijkheden op het gebied van een actieve leefstijl te benutten.

**Hoofdstuk 6** beschrijft een onderzoek naar een meetinstrument dat gebruik kan worden bij het uitgebreide assessment dat nodig is om een beweegprogramma dat toegesneden is op de mogelijkheden van een individuele patiënt met axSpA te kunnen opstellen. De rompmusculatuur is van bijzonder belang bij deze patiëntengroep, omdat ontstekingen en verminderde mobiliteit invloed hebben op de dynamische stabilisatie van de wervelkolom. Een zekere mate van rompstabiliteit, in termen van spierkracht en spieruithoudingsvermogen, is nodig bij activiteiten in het dagelijks leven en sportbeoefening. Het beoordelen van de spierkracht van de rompmusculatuur is echter lastig en er bestond geen geschikte beoordelingsmethode voor mensen met axSpA die deelnemen aan groepsoefentherapie.

Het doel van het onderzoek was om de Core Strength Endurance test battery (CSE), die voorheen bij sporters werd gebruikt, aan te passen voor gebruik bij mensen

met axSpA, leidend tot de zogenaamde adapted CSE (aCSE). Daarbij werd de intrabeoordelaarsbetrouwbaarheid bepaald en de associatie met drie ziektespecifieke factoren, namelijk functionele status, zelfgerapporteerde pijn en perceptie van spierkracht. Hiertoe werd een cross-sectioneel onderzoek uitgevoerd onder 62 mensen met axSpA die deelnamen aan groepsoefentherapie en 13 fysiotherapeuten die als supervisor van deze groepen fungeerden. De aCSE werd afgenomen en na 7-14 dagen herhaald om de intrabeoordelaarsbetrouwbaarheid van de supervisoren te bepalen. De betrouwbaarheid werd berekend met behulp van de intraclass-correlatiecoëfficiënt (ICC).

De intrabeoordelaarbetrouwbaarheid (ICCs (95% BI)) voor de ventrale (0,54 (0,35-0,74)), laterale (0,52 (0,33-0,70)) en dorsale (0,71 (0,58-0,86)) spiergroepen van de romp was matig tot aanzienlijk. Om de associatie tussen het krachtuithoudingsvermogen van de ventrale, laterale en dorsale rompspieren en de ziektespecifieke factoren vast te stellen, werden Pearson correlatiecoëfficiënten berekend. Op basis van deze bevindingen werd geconcludeerd dat de aCSE een haalbaar en betrouwbaar beoordelingsinstrument is om rompkracht te meten bij mensen met axSpA. Deze methode kan worden gecombineerd met andere relevante methoden om cardiorespiratoire conditie, beweeglijkheid (van de wervelkolom) en neuromotore activiteiten te meten. De combinatie omvat daarmee een betaalbaar, praktisch, en toch uitgebreid instrumentarium voor het assessment van deelnemers aan groepsoefentherapie voor mensen met axSpA. Daaruit bleek dat geen van de aCSE-parameters significant was gecorreleerd met de ziektespecifieke factoren.

Het doel van de SVMB was om het tot dan toe gehanteerde concept voor groepsoefentherapie te herzien volgens state-of-the-art maatstaven. Om deze reden werd de EULAR aanbevelingen voor lichamelijke activiteit van mensen met inflammatoire artritis en artrose uit 2018 vertaald naar een nieuw concept voor groepsoefentherapie, "BeFit" geheten. Vervolgens werden vijf belangrijke implementatieactiviteiten uitgevoerd: a) trainen van de begeleidende fysiotherapeuten; b) integreren van correct gedoseerde oefeningen in alle oefendomeinen (dat wil zeggen: cardiorespiratoir, spierkracht, lenigheid, neuromotore of functionele/balansoefeningen); c) geven van educatie en beweegadvies; d) uitvoeren van halfjaarlijkse conditietests; en e) aanbieden van individuele oefeningen (naast groepstrainingen). Door dit concept toe te passen binnen de groepsoefentherapie, zou het programma niet alleen in overeenstemming zijn met de principes van oefentherapie voor mensen met axSpA maar ook met de beweegrichtlijnen voor de algemene bevolking.

**Hoofdstuk 7** beschrijft de ontwikkeling van een implementatiestrategie-op-maat voor BeFit en een pilotonderzoek om deze strategie te evalueren bij vier oefengroepen. Na zes maanden testen van BeFit, werd de implementatie geëvalueerd vanuit het perspectief van de patiënt, de fysiotherapeut en de organisatie. In totaal namen 30 mensen met axSpA en vier fysiotherapeuten hieraan deel.

Vanuit het patiëntperspectief was de primaire uitkomst de mate waarin het aanbevolen

oefen- en beweeggedrag werd nageleefd. Dit werd beoordeeld op basis van het aantal in het elektronische beweeglogboek vermelde trainingssessies, opgesplitst naar oefendomein, over een periode van 6 maanden. De secundaire uitkomsten waren de haalbaarheid van en de tevredenheid over het nieuwe concept voor oefentherapie, gemeten op een schaal van 0 tot 10. Om de effectiviteit van de geïndividualiseerde beweegadviezen te evalueren was de tertiaire uitkomst de conditieverbetering van de patiënten, bepaald aan de hand van gemeten fitnessparameters. Vanuit het perspectief van de fysiotherapeut was de primaire uitkomst de mate waarin de concepten waarop het programma was gebaseerd werden gevolgd. Om dit vast te kunnen stellen werd een logboek bijgehouden. De secundaire uitkomsten waren de kwaliteit en haalbaarheid van en de tevredenheid over de behandeling. Deze aspecten werden geëvalueerd met behulp van semigestructureerde interviews. Op organisatieniveau werd gekeken of BeFit geaccepteerd werd en of het een vaste plek kreeg binnen de structuur van de organisatie.

De door de patiënten gerapporteerde naleving van de beweegadviezen werd onvoldoende bevonden (43%). Mogelijk kwam dit door technische problemen met het logboek. De mate waarin de fysiotherapeuten het nieuwe beweegconcept volgden, werd als voldoende beoordeeld. Vanuit alle drie de perspectieven werd het nieuwe concept in het algemeen gezien als haalbaar en als een zinvolle aanvulling op het beweegaanbod inclusief het zelfstandig oefenen. De aanbevolen frequentie van het aanbieden een individueel oefen- en beweegadvies (eens per kwartaal) en de afname van de conditietests (eens per halfjaar) werd door zowel de patiënten als de fysiotherapeuten als te hoog en te weinig flexibel ervaren. In het algemeen was de acceptatie van BeFit en de tevredenheid over dit concept in de implementatiepilot voldoende. Desondanks lijkt het op grond van de bevindingen noodzakelijk om een aantal aanpassingen zijn in hoofdstuk 7 in detail beschreven, zodat zij niet alleen aan de SVMB maar ook aan andere, soortgelijke, toekomstige implementatieonderzoeksprojecten ten goede kunnen komen.

#### Algemene Discussie

De EULAR aanbevelingen voor lichamelijke activiteit van mensen met inflammatoire artritis en artrose uit 2018, en de literatuurstudie en meta-analyse die daaraan ten grondslag liggen, dragen bij aan het empirische bewijs voor de effectiviteit van niet-farmacologische zorg voor mensen met RMDs. Er is sterk bewijs (Hoofdstuk 3) dat lichamelijke activiteit volgens beweegrichtlijnen voor de algemene bevolking binnen de vier oefendomeinen (cardiorespiratoire conditie, spierkracht, flexibiliteit en neuromotoriek/functioneel/balans) effectief, veilig en haalbaar is voor mensen met RMDs (Hoofdstuk 2). De EULAR aanbevelingen benadrukken belangrijke aspecten waarmee bij beweeginterventies voor mensen met RMDs rekening moet worden gehouden. Mensen met RMDs kunnen ziektespecifieke of andere beperkingen hebben die vragen om aanpassingen van beweegadviezen aan hun individuele behoeften, voorkeuren, capaciteiten en de beschikbare voorzieningen en middelen. In de toekomst zal moeten worden vastgesteld worden of verdere verfijning van de richtlijn, in de vorm van extra, ziektespecifieke aanbevelingen, nodig is in om de toepassing ervan in de dagelijkse praktijk te vergemakkelijken.

Daarnaast zijn er nog andere relevante kennishiaten, zoals meer inzicht in welke interventiestrategieën het meest effectief zijn om lichamelijke activiteit ook op de lange termijn te bevorderen, wat de haalbaarheid is van beweeginterventies die alle vier de domeinen van beweging omvatten, en welke methoden voor het objectief beoordelen van de effecten van oefenen en bewegen het meest geschikt, valide en bruikbaar zijn in de klinische praktijk. Gelukkig is er al veel onderzoek gedaan op het gebied van beweeginterventies en beweeggedrag, en zal vooral door een goede afstemming en samenwerking tussen onderzoekers en onderzoeksgroepen, vooruitgang worden geboekt.

limplementatie van de EULAR aanbevelingen in de klinische zorg vergt een systematische aanpak, op basis van implementatie-theorie. Dit proefschrift beschrijft de vertaling van de EULAR aanbevelingen in BeFit, een concept voor groepsoefentherapie voor mensen met axSpA. Tevens wordt een pilot implementatiestudie van BeFit beschreven (Hoofdstuk 7). De pilot was gebaseerd op het implementatiemodel van Grol & Wensing (1), waarbij gebruik gemaakt werd van een zogenaamd "hybride designtype 3".

Aan de basis van de implementatiestudie lagen drie onderzoeken (Hoofdstukken 4-6) die er op gericht waren om de huidige situatie te analyseren en het ontwikkelingsproces van de implementatieactiviteiten te ondersteunen. Deze diepgaande analyse van de context is een sterk punt van de implementatiestudie. Bovendien waren de belangrijke stakeholders, in het bijzonder mensen met axSpA, fysiotherapeuten en de SVMB, betrokken bij het onderzoek. Als een andere implementatiemethode was gebruikt dan die van Grol & Wensing, bijvoorbeeld de PRACTIS-methode (Practical planning for Implementation and Scale-up) (2), die speciaal is ontworpen voor het plannen en opschalen van beweeginterventies, was er mogelijk in de beginfase van het onderzoek nog meer en beter samengewerkt tussen alle belanghebbenden en was de implementatie van BeFit succesvoller verlopen. Hoewel de fysiotherapeuten en de meeste deelnemers aan de pilot-oefengroepen geïnteresseerd waren in het nieuwe concept en bereid waren hun gebruikelijke zorg aan te passen, kan het zijn dat andere oefengroepen niet zo innovatief ingesteld zijn en het belang van verandering niet inzien. Een belangrijke verandering voor de fysiotherapeuten was hun nieuwe rol als adviseur op het gebied van lichamelijke activiteit. Bevordering van bewegen is weliswaar een kernelement van fysiotherapie (3), maar sommige fysiotherapeuten beschikken mogelijk nog niet over de benodigde communicatievaardigheden, gedragsveranderingstechnieken en/of gedetailleerde kennis van beweegrichtlijnen om BeFit succesvol in de praktijk te kunnen integreren.

Het focusgroeponderzoek naar 'beliefs' van mensen met axSpA ten aanzien van algemene lichamelijke activiteit en cardiorespiratoire training liet zien dat bewegen belangrijk is als

strategie voor patiënten om zo goed mogelijk met hun ziekte om te kunnen gaan. Het conceptuele verschil tussen lichamelijke activiteit en cardiorespiratoire training was voor sommige deelnemers echter niet duidelijk. Aangezien axSpA geassocieerd is met een verhoogd risico op cardiovasculaire aandoeningen, is de conclusie dat patiënten en fysiotherapeuten het verschil tussen deze twee concepten beter moeten begrijpen en moeten weten hoe ze het beste de cardiorespiratoire training kunnen uitvoeren. Daarnaast is een aandachtspunt dat de meeste deelnemers waren niet op de hoogte waren van het verhoogde risico op cardiovasculaire aandoeningen bij axSpA patiënten (Hoofdstuk 4). Uit het vragenlijstonderzoek (Hoofdstuk 4) werd duidelijk dat de meeste mensen met axSpA gebruik maken van fysiotherapie, individueel en/of in groepsverband. Fysiotherapeuten zijn dus bij uitstek in de positie om de gezondheidsvaardigheden van patiënten te verbeteren zodat zij lichamelijke activiteit kunnen inzetten als zelf-managementstrategie bij het omgaan met de ziekte. Beweegactiviteiten moeten zorgvuldig worden gepland aan de hand van individuele doelen die regelmatig opnieuw moeten worden geëvalueerd door middel van - bij voorkeur objectieve - beoordelingscriteria. In Hoofdstuk 6 wordt een betrouwbare en eenvoudig uit te voeren methode beschreven om de kracht van de rompmusculatuur te meten bij mensen met axSpA. Deze test zou kunnen worden ingezet bij een periodiek assessment dat bij mensen met axSpA zou moeten worden uitgevoerd om hun beweegdoelen en beweegprogramma vast te stellen en waar nodig aan te passen.

#### Conclusie

De bijdrage van dit proefschrift aan de al bestaande kennis op het gebied van oefenen en bewegen van mensen met RMDs is dat het een overzicht geeft van het gehele ontwikkelingstraject van de EULAR aanbevelingen voor lichamelijke activiteit van mensen met RMDs (inclusief axSpA), tot en met de implementatie van de concepten van de richtlijn in de groepsoefentherapie voor mensen met axSpA. Volgens de EULAR aanbevelingen is lichamelijke activiteit volgens beweegrichtlijnen voor de algemene bevolking voor mensen met RMDs effectief, veilig en haalbaar. Daarnaast leveren de studies in dit proefschrift meer inzicht in de mogelijkheden en uitdagingen die samenhangen met het implementeren van de nieuwe inzichten op het gebied van lichamelijke activiteit binnen de groepsoefentherapie voor mensen met axSpA.

# Referenties

- 1. Grol, R. and M. Wensing, Improving patient care. The implementation of change in health care. 2de ed. 2013, Oxford: Wiley Blackwell.
- Koorts, H., et al., Implementation and scale up of population physical activity interventions for clinical and community settings: the PRACTIS guide. Int J Behav Nutr Phys Act, 2018. 15(1): p. 51.
- World\_Physiotherapy, Physical therapists as exercise and physical activity experts across the life span. Policy statement, in https://world.physio/sites/default/files/2020-04/PS-2019-Exercise-experts.pdf. 2019, World Confederation for Physical Therapy.

# 8

List of Publications Curriculum Vitae Acknowledgements

### List of Publications

#### Publications in peer-reviewed journals

Vogt U, **Rausch Osthoff AK**, Niedermann K. Welche Kommunikationstechniken sind bei Physiotherapeut\*innen während einer Bewegungsberatung mit Personen mit axialer Spondyloarthritis identifizierbar? physioscience. *Accepted July 2022* 

**Rausch Osthoff AK**, Nast I, Niedermann K. Understanding beliefs related to physical activity in people living with axial spondyloarthritis - a theory-informed qualitative study. BMC Rheumatology 2022; 6 (1):40

Ettlin L, Bruderer-Hofstetter M, **Rausch Osthoff AK**, Nast I, Gaugler O, Niedermann K. Evaluation of the strategy for implementing the GLA:D programme in Switzerland: protocol for an implementation-effectiveness hybrid type 3 design study with a mixed-method approach. BMJ Open. 2022;12(6):e057993

**Rausch Osthoff AK**, Horvath C, Niedermann K. Assessments zur Erhebung des Fitnesszustandes in Therapiegruppen für Personen mit axialer Spondyloarthritis – eine Machbarkeitsstudie. physioscience 2022; 18: 1–8

Saba R, Bruderer-Hofstetter M, **Rausch Osthoff AK**, Niedermann K. Übersetzung, Test-Retest-Reliabilität und Konstruktvalidität der deutschen Version der Exercise Self-Efficacy Scale für körperliche Aktivität bei Personen mit axialer Spondyloarthritis. physioscience 2022; 18:1-10

**Rausch Osthoff AK**, Vliet Vlieland TPM, Meichtry A, van Bodegom-Vos L, Topalidis B Büchi, S, et al. Lessons learned from a pilot implementation of physical activity recommendations in axial spondyloarthritis exercise group therapy. BMC Rheumatol. 2022; 6(1):12

**Rausch Osthoff AK**, Baltisberger P, Meichtry A, Topalidis B, Ciurea A, Vliet Vlieland TPM, et al. Reliability of an adapted core strength endurance test battery in individuals with axial spondylarthritis. Clin Rheumatol. 2021;40(4):1353-1360

Ettlin L, **Rausch Osthoff** AK, Nast I, Niedermann K. Applicability of exercise and education programmes for knee osteoarthritis management to Switzerland. Front Health Serv 2021; 1:760814

Niedermann K, **Rausch Osthoff AK**, Braun J, Becker H, Böhm P, Bräm R, et al. Die laienverständliche Version der 2018 EULAR Empfehlungen zu körperlicher Aktivität von Menschen mit entzündlich-rheumatischen und degenerativen Erkrankungen : Übersetzung ins Deutsche und sprachliche Validierung im deutschsprachigen Raum mit Betroffenen. Z Rheumatol. 2021 doi: 10.1007/s00393-021-01079-z

Kiltz U, Kiefer D, Braun J, **Rausch Osthoff AK**, Herbold S, Klinger M, et al. Übersetzung der 2018 EULAR Empfehlungen zu körperlicher Aktivität von Menschen mit entzündlichrheumatischen und degenerativen Erkrankungen ins Deutsche und sprachliche Validierung im deutschsprachigen Raum mit medizinischen Fachpersonen. Z Rheumatol. 2021. doi: 10.1007/s00393-021-01078-0

**Rausch Osthoff AK**, Baur H, Reicherzer L, Wirz M, Keller F, Opsommer et al. Physiotherapists' use and perceptions of digital remote physiotherapy during COVID-19 lockdown in Switzerland: an online cross-sectional survey. Arch Physiother. 2021; 11(1):18

**Rausch Osthoff AK**, Beyer S, Gisi D, Rezek S, Schwank A, Meichtry A, et al. Effect of counselling during pulmonary rehabilitation on self-determined motivation to be physically active for people with chronic obstructive pulmonary disease: a pragmatic RCT. BMC Pulm Med. 2021; 21(1):317

Scheermesser M, Reicherzer L, Beyer S, Gisi D, Rezek S, Hess T, Wirz M, **Rausch Osthoff AK.** The Influence of Pulmonary Rehabilitation and Counselling on Perceptions of Physical Activity in Individuals with COPD - A Qualitative Study. Int J Chron Obstruct Pulmon Dis. 2021; 16:2337-2350

Kahrmann N, Scheermesser M, **Rausch Osthoff AK**. The effect of Motivational Interviewing on physical activity in people with chronic obstructive pulmonary disease: a systematic review. physioscience 2019; 15(2): 81-91

**Rausch Osthoff AK**, van der Giesen F, Meichtry A, Walker B, Van Gaalen F, Goekoop-Ruitermans YPM, et al. The perspective of people with axial spondyloarthritis regarding physiotherapy: room for the implementation of a more active approach. Rheumatology Advances in Practice. 2019; 3 (2)

**Rausch Osthoff AK**\*, Niedermann K\*, Braun J, Adams J, Brodin N, Dagfinrud H, et al. 2018 EULAR recommendations for physical activity in people with inflammatory arthritis and osteoarthritis. Annals of the rheumatic diseases. 2018;77(9):1251-60 \*contributed equally to the manuscript

**Rausch Osthoff AK**, Juhl CB, Knittle K, Dagfinrud H, Hurkmans E, Braun J, et al. Effects of exercise and physical activity promotion: meta-analysis informing the 2018 EULAR recommendations for physical activity in people with rheumatoid arthritis, spondyloarthritis and hip/knee osteoarthritis. RMD Open. 2018;4(2):e000713

Bruderer-Hofstetter M, **Rausch Osthoff AK**, Meichtry A, Münzer T, Niedermann K. Effective multicomponent interventions in comparison to active control and no interventions on physical capacity, cognitive function and instrumental activities of daily living in elderly people with and

without mild impaired cognition - A systematic review and network meta-analysis. Ageing Res Rev. 2018; 45:1-14

**Rausch Osthoff AK**, Greco N, Schwank A, Beyer S, Gisi D, Scheermesser M, et al. Effect of counselling during pulmonary rehabilitation on self-determined motivation towards physical activity in people with chronic obstructive pulmonary disease - protocol of a mixed methods study. BMC Pulm Med. 2017; 17(1):115

Kälin S, **Rausch Osthoff AK**, Bauer CM. What is the effect of sensory discrimination training on chronic low back pain? A systematic review. BMC Musculoskelet Disord. 2016; 2;17:143

Hügli AS, Ernst MJ, Kool J, Rast FM, **Rausch Osthoff AK,** Mannig A, et al. Adherence to home exercises in non-specific low back pain. A randomised controlled pilot trial. J Bodyw Mov Ther. 2015;19(1):177-85

Ernst MJ, Crawford RJ, Schelldorfer S, **Rausch Osthoff AK**, Barbero M, et al. Extension and flexion in the upper cervical spine in neck pain patients. Man Ther. 2015;20(4):547-52

**Rausch Osthoff AK**, Ernst MJ, Rast FM, Mauz D, Graf ES, Kool J, et al. Measuring lumbar reposition accuracy in patients with unspecific low back pain: systematic review and meta-analysis. Spine 2015; 5;40(2): E97-E111

van Gestel AJ, Baty F, **Rausch Osthoff AK**, Brutsche MH. Cardiopulmonary and gas-exchange responses during the six-minute walk test in patients with chronic obstructive pulmonary disease. Respiration. 2014;88(4):307-14.

**Rausch Osthoff AK**, Kohler M, Sievi NA, Clarenbach CF, van Gestel AJ. Association between peripheral muscle strength, exercise performance, and physical activity in daily life in patients with Chronic Obstructive Pulmonary Disease. Multidiscip Respir Med. 2014:9(1):37

**Rausch Osthoff AK**, Taeymans J, Kool J, Marcar V, van Gestel AJ. Association between peripheral muscle strength and daily physical activity in patients with COPD: a systematic literature review and meta-analysis. J Cardiopulm Rehabil Prev. 2013;33(6):351-9

#### Other Publications

**Rausch Osthoff AK** Barrieren für die evidenzbasierte Praxis in der Physiotherapie. Editorial. Physioscience. *accepted August 2022* 

**Rausch Osthoff AK** Evaluation von körperlicher Aktivität. Chapter in: Physiotherapie bei chronischen Atemwegs- und Lungenerkrankungen – Evidenzbasierte Praxis. 3. Auflage. Springer Verlag Berlin. *Accepted May 2022*  Rausch Osthoff AK, Steier J. Lebensqualität. Chapter in: Physiotherapie bei chronischen Atemwegs- und Lungenerkrankungen – Evidenzbasierte Praxis. 3. Auflage. Springer Verlag Berlin. Accepted May 2022

Karstens S, Braun T, Bruderer-Hofstetter M, Diermayr G, Gafner S, Kokow C, Pott C, **Rausch Osthoff AK**, et al. Relevanz und nicht mehr Signifikanz- Einbidung und Interpretation von Konfidenzintervallen zur Beurteilung von Behandlungseffekten. Editorial. Physioscience 2022; 18 (02): 49-51

**Rausch Osthoff AK**, Reicherzer L, Stahn M, Wirz M, Zukunftsmodell Telephysioherapie – ein Überblick für die Schweiz. Physioactive 2022; 2

Stahn M, **Rausch Osthoff AK**. 2022. Teletherapie. Von der "Notlösung" zum Gamechanger? Z. f. Physiotherapeuten 2022; 74 (8): 18-24

Nast I, Niedermann K, Carlander M, Mattli R, **Rausch Osthoff AK**, Sommer B, Wieber F, Wirz M. (2021) Bewegung als Therapie. Schlussbericht im Auftrag des Bundesamts für Gesundheit (BAG). www.bag.admin.ch/home/das-bag/publikationen/forschungsberichte/forschungsberichte-rinaehrung-bewegung.htm

**Rausch Osthoff AK,** Bruderer-Hofstetter M. Exercise is medicine – eine globale Gesundheitsinitiative. Editorial. Physioscience 2019; 15:49-50

**Rausch Osthoff AK,** Bauer C, Wirz M. Keep record! Zukunftsvision: die standardisierte, elektronische Dokumentation von physiotherapeutischen Leistungen. Editorial. Physioscience 2020; 16: 145-146

Rausch Osthoff AK Ethik in der Physiotherapie. Editorial. Physioscience 2016; 12. 1-2

#### **Curriculum Vitae**

Anne-Kathrin Rausch was born on 12th November 1984 in Offenbach am Main, Germany. After graduating from high school (private Gymnasium Canisianum, Lüdinghausen), she studied physiotherapy at the Hochschule Fresenius Idstein (DE), which was one of the first Universities of Applied Sciences in Germany to offer the opportunity to obtain a Bachelor of Science in European Studies, in cooperation with the Hogeschool Utrecht (NL). During her bachelor studies, she earned the propaedeutic, attaining a teaching licence in the federal state of Hessen. Additionally, she worked at the physiotherapy clinic of the Olympic Centre Frankfurt-Rhein-Main. In 2008, Anne-Kathrin wrote her bachelor thesis on a small study investigating the electromyographic activity of the pelvic floor during horse riding. This experience awoke her interest in research. Consequently, she decided to continue her studies with a Masters' degree. Two years later, after gaining additional clinical experience working as a physiotherapist at an outpatient clinic in Aarau (Switzerland), she registered for an MSc in Physiotherapy at the Zurich University of Applied Sciences (ZHAW). This was the first course of its kind in Switzerland. The part-time programme ran for three years and included two short-term internships at the Maastricht University (NL), Department for Epidemiology and the George Institute for Global Health (AUS), Musculoskeletal Division. At the George Institute, she worked on the PEDro project, led by Anne Mosely, initiating the inclusion of German randomized controlled studies into the Physiotherapy Evidence Database. In 2013, Anne-Kathrin wrote her Master thesis at the University Hospital Zurich (USZ), Department of Pulmonology, supervised by Prof. Dr. Arnoldus van Gestel, on the association between peripheral muscle strength and daily physical activity in patients with chronic obstructive pulmonary disease.

In 2012, Anne-Kathrin started working as a research fellow in the Research & Development department, Institute of Physiotherapy, at the ZHAW. Since then, she has been involved in many research projects related to various fields of physiotherapy, applying qualitative and quantitative methods. Anne-Kathrin's main research focus is on the promotion of physical activity in people with chronic conditions. As a consequence, in 2015, she became a part-time PhD student at the Leiden University Medical Center (LUMC), supervised by Prof. Dr. Thea P.M. Vlieland and Prof. Dr. Karin Niedermann (ZHAW). Her PhD studies have focussed on physical activity promotion in the field of rheumatology. She was introduced into the network of EULAR Health Professionals in Rheumatology (HPR) by both supervisors. Currently, Anne-Kathrin is a member of the EULAR HPR representation committee and HPR liaison in the EULAR patient representation committee (PARE). She has presented her work at many national and international conferences. Anne-Kathrin is member of the editorial board of 'Physioscience' (Thieme), a journal promoting physiotherapy-related research in German. Since Spring 2022, she is responsible for the focus area "physical activity" of the Master of Science in Physiotherapy at the ZHAW.

Anne-Kathrin lives in Zurich, Switzerland, with her husband Stefan and their three children Johanna (2014), Samuel (2015) and Lorenz (2019).

### Acknowledgements

This thesis could have only been realised with the support and valuable contribution of many people. My thanks go to everyone who has helped me on this journey, but there are a few persons who deserve a special mention.

I give special thanks to my supervisors Prof. Dr. Thea P.M. Vliet Vlieland and Prof. Dr. Karin Niedermann for accompanying me with trust, humour, enthusiasm, and patience during the seven years of this PhD. It takes a unique skill to positively stimulate a PhD student, while at the same time remaining critical of the results. You have both mastered this delicate balancing act to perfection – Thea with a "fresh mind" from a distance and Karin during daily business. You have given me a great deal of freedom to pursue my ideas. But you were always there to advise me when I needed support. Both of you introduced me to the work of EULAR, which is a fantastic network. Many thanks for that. I appreciate very much the constructive and successful collaboration - despite the distance and rare face-to-face meetings.

I would like to thank the members of the promotiecommissie for giving their valuable time to read and critically evaluate my thesis.

Conducting patient-related research is dependent on finding study participants. Therefore, I am grateful to all patients and physiotherapists for their support and commitment. My thanks also go to the staff of the Schweizerische Vereinigung Morbus Bechterew, led by René Bräm, for their trust in me and their willingness to innovate.

My team at the ZHAW have given me their continuous support and always shown interest in my work. I am especially grateful to the head of research, Prof. Dr. Markus Wirz for his "open door at any time" policy. To Christa Wachter for her down-to-earth view on research and her humour, to André Meichtry for his critical thinking and essential statistical support, to Markus Ernst for his R-tricks, to Prof. Dr. Irina Nast for her support on qualitative methods and her critical review of this thesis, to Leah Reicherzer for the coding-support, to Dr. Lea Ettlin and to Dr. Marina Bruderer-Hofstetter for their critical feedback on the thesis.

From the LUMC, I would like to thank Dr. Leti van Bodegom-Vos for introducing me to the world of implementation science, Dr. Florus van der Giesen for his staying power with the BEVER-manuscript, Anika Hoogenstraaten for her constant support, and Bas Hilberdink, for being my sparring partner during the defence preparation.

I thank Karen Linwood for her mindful English editing of the manuscripts and thesis. Thanks also to Jonas Wüllner for the cover design of this thesis.

I am very grateful to my friends for providing me with distractions, your interest in my work and of course your friendship. It was sometimes difficult to stay in touch with everyone alongside the research. Thank you.

Without the support of my family, this thesis would not exist. Thank you for your continuous encouragement, interest and support. Having three small kids challenged my PhD trajectory. My parents, parents-in-law, Bernadette, and Stefan: you accompanied me to conferences and meetings at the ZHAW, Lisbon, Enschede and other places, cuddling the babies in side rooms or running the show at home while I was working. Thank you for your fantastic efforts and infinite patience.

I owe my deepest thanks to my husband Stefan and our children Johanna, Samuel, and Lorenz. Your love, humour and understanding gave me the means to overcome the challenges of this thesis. You have always reminded me that real life cannot be celebrated in front of the computer. Thank you.